This is a true and correct copy of the Regional Plan: Air for Otago, incorporating Plan Changes 1 and 2, that was approved by the resolution of the Otago Regional Council on Wednesday, 10 December 2008.

The Common Seal of the Otago Regional Council was hereunto affixed pursuant to the resolution of the Council passed on 10 December 2008, in the presence of:

aini



Stephen Cairns Chairperson

for

Graeme Martin Chief Executive

Key Event	Date Notified	Date Decision Released	Date Operative
Regional Plan: Air	28 February 1998	30 June 2001	1 January 2003
Plan Change 1 (Fire Service Training)	17 December 2005	29 March 2006	10 April 2006
Plan Change 2 (National Environmental Standards)	14 April 2007	12 December 2007	1 January 2009

Regional Plan: Air for Otago



Otago Regional Council Updated to 1 January 2009

ISBN 1-877265-76-4

Chairperson's Foreword

Clean, clear air is a resource that is unseen and taken for granted until it is lost. As soon as the air is not fresh and clear we all get concerned. Concerns about loss of air quality range across visual, property, odour and health effects, and even extend to climate.

Here in Otago the air quality is usually better than in many other parts of the world, but we do have areas where, at times, the quality of air is degraded. This Regional Plan: Air sets out the objectives, policies and rules the Otago Regional Council will use for keeping our air clean, and improving the quality of air where it has been, or is at risk of being, degraded.



This Regional Plan: Air ushers in a new era for air quality management in Otago and sets out clear directions for activities that might affect air at a local or regional level. The Plan provides an integrated approach to air management issues, the aim of which is to reduce the adverse effects associated with activities that affect air quality. For that reason backyard burning has effectively been banned in urban areas of Dunedin City.

This revision incorporates changes made in response to the Government's 2004 National Environmental Standards for Air Quality (NESAQ), which set standards governing ambient (outdoor) air quality and new wood burners installed in urban areas.

While prepared by the Otago Regional Council, the Plan belongs to the region. It was developed following extensive consultation, carried out over a number of years with the community, Otago's city and district councils and industry.

Maintaining air quality is in everyone's interest and we must work together to retain and improve our air quality.

I thank all those groups and individuals who participated in developing this Plan. Your comments have been invaluable and the Council looks forward to working with the Otago community in implementing the Regional Plan: Air.

Stephen Cairns 1 January 2009

How to use the Regional Plan: Air for Otago

This Regional Plan: Air for Otago considers the use and management of Otago's air resource. The Plan provides objectives, policies, rules and other methods of implementation to address the use and management of the air resource. The rules of the Plan determine the status of any particular activity and determine whether a resource consent will be required before that activity can be carried out.

A resource consent is required for any activity which this Regional Plan: Air specifies as being a discretionary activity.

In some cases, the Plan specifies certain activities as being prohibited activities. These are activities which may not occur and for which no resource consent will be granted.

Table of Contents

Part I Introdu	ction		1
1.1	Purpose of this Pla	າກ	2
1.2	The need for this	Plan	2
1.3	Area covered by t	nis Plan	2
1.4	•	rocess	
1.5		lan	
2		licy framework	
2.1		nagement Act 1991	
2.2		ner resource management documents	
2.2	-		
3	Ų	: Air perspective	
J		Acknowledgements	
		-	
. 0		es	
4			
4.1	U)	
4.2		go	
4.3	Discharges from i	ndustrial or trade premises	12
4.4	Domestic heating	and burning of waste	13
4.5	Dust from area so	Irces	15
4.6	Odour		16
4.7	Agrichemical spra	y drift	17
4.8	•	g on production land	
4.9		ns	
4.10	*		
	14 3.4		- 1
		t: Objectives and Policies	
5.1		ework	
6.1			
7.1		nu ki Otago	
8.1		nt air quality	
8.2		or managing discharges of contaminants into the air	
9.1		ng discharges of PM ₁₀	
10.1	-	m area sources	
11.1	2		
12.1	Policy for agriche	mical spray drift	39
13.1	Policy for the burn	ning of vegetative matter on production land	41
14.1	Policy for motor w	ehicle emissions	41
15.1	Policy for global i	ssues	42
Part IV Rules			
16.1			
10.2			
16.3			
10.5		es from domestic heating appliances	
		burning	
		or incineration of specified materials	
		of combustion from fuel burning equipment	
		ses from industrial or trade processes	
		blasting	
		anagement	
		e farming	
		nical application	
		pour, heat and energy	
	16.3.11 Ventilat	on or vapour displacement	76

16.3.1	2 Nuclear power generation or nuclear weapon manufacturing	76
16.3.1	3 General permitted activities	77
16.3.1	4 Other discretionary activities	
16.3.1	5 Discharges of PM ₁₀	
16.4 Inform	nation requirements	81
Dout V Mothoda Oth	er Than Rules	95
	uction	
	n with city and district councils	
	n with other organisations	
	onmental education and promotion	
	cacy and information transfer	
	Environmental Results	
	uction	
18.2 Antici	pated environmental results	
Part VII Cross-Boun	dary Issues	
	uction	
	ods	
Part VIII Monitoring	g and Review	
20.1 Introd	uction	
20.2 Eleme	ents to be monitored	
20.3 Monit	oring techniques	
20.4 Plan re	eview	99
Schedules		101
Schedule 1	Regional Ambient Air Quality Guidelines	
Schedule 2	Maps Showing Air Zone boundaries	
Schedule 3	Hazardous Air Contaminants	139
Schedule 4	Good Management Practices for Agrichemical Application	
Schedule 5	Good Management Practices to Prevent or Minimise	
	the Discharge of Smoke from Burning Vegetation	
Schedule 6	Setting Chimney Heights for Industrial or Trade Processes	147
Schedule 7	Standard Dispersion Modelling Procedure	
Glossary		151

Part I Introduction



1.1 Purpose of this Plan

This Plan is to be known as the Regional Plan: Air for Otago. It has been prepared in order to promote the sustainable management of the air resource in Otago and thereby assist the Otago Regional Council to carry out its functions under the Resource Management Act.

1.2 The need for this Plan

The functions of the Otago Regional Council under the Resource Management Act 1991 include air quality management.

Section 65 of the Resource Management Act enables the Otago Regional Council to prepare a regional plan in respect of any function for which it is responsible. Section 65(3) states that regional councils shall consider the desirability of preparing a regional plan whenever the following circumstances arise or are likely to arise:

- (h) any use of land or water that has actual or potential adverse effects on soil conservation or air quality or water quality;
- *(i)* any other significant issue relating to any function of the regional council under this *Act.*

Consultation during the preparation of this Plan has highlighted the potential for some land uses to adversely affect air quality, and in particular to create problems of dust, odour and agrichemical spray drift. Degradation of ambient air quality has also been identified in the Regional Policy Statement for Otago as an issue of significance in the Otago region.

After consultation, and having regard to the purpose of the Resource Management Act, other methods available, and after having carried out an evaluation of the likely benefits and costs of the principal alternative means, the Otago Regional Council has decided that this Plan is the best method for:

- Establishing a long term and integrated strategy for efficiently managing Otago's air resource; and
- Maintaining and enhancing Otago's existing air quality as directed by the Regional Policy Statement for Otago.

1.3 Area covered by this Plan

This Plan applies to the whole of the Otago region, including Otago's coastal marine area which extends 12 nautical miles (22.2 kilometres) from mean high water springs. The Otago Region was constituted under the "Local Government (Otago Region) Reorganisation Order 1989" and is shown in Figure 1. The region comprises the districts of the:

- Dunedin City Council;
- Central Otago District Council;
- Clutha District Council;
- Queenstown-Lakes District Council; and
- Waitaki District Council (part only).

PART I INTRODUCTION



Figure 1: The Otago Region

1.4 Plan preparation process

This Plan has been prepared following a process in accordance with the requirements of the First Schedule to the Resource Management Act.

A discussion paper entitled "Air Quality Discussion Paper" was released in May 1996 and a total of 224 submissions were received.

This Plan was publicly notified on 28 February 1998. There were 261 submissions and 47 further submissions received. The decisions were released on 30 June 2001 and, after some appeals had been resolved, the Regional Plan: Air was made operative on 1 January 2003.

This Plan was changed by Plan Change 1, proposed on 17 December 2005 and made operative on 10 April 2006 following the receipt of no submissions. That introduced Rule 16.3.13.2 in place of Rule 16.3.13.1(7) and made several consequential changes.

This Plan was amended by Plan Change 2, proposed on 14 April 2007 and made operative on 1 January 2009 following the receipt of 840 submissions and 44 further

PART I INTRODUCTION

submissions. The amendments related to ambient air quality, and introduced new rules regarding discharges from domestic heating appliances.

1.5 Structure of this Plan

The structure of this Plan is based on the requirements for a regional plan as set out in Section 67(1) of the Resource Management Act. The Plan is divided into eight parts.

Part I (this part) outlines the purpose of the Plan, the area it covers and the plan preparation process. Part I also details the legislative framework for air quality management, how integrated management is to be achieved and outlines the perspective of Kai Tahu, Otago's manawhenua, to air.

Part II outlines the main air quality issues in Otago.

Part III outlines the approach adopted in this Plan for the management of air quality within the Otago region. Firstly, overall objectives for air quality management are defined. Secondly, policies for addressing the issues identified in Part II are stated. These policies set out the general direction that the Council will take, and establish criteria to guide the Council's decision making on resource consent applications.

Part IV sets out the rules which apply for the discharge of contaminants into air in Otago. It also specifies the type of information that will be required with any resource consent application. This part of the Plan is particularly important as it sets out situations where consents are required for the discharge of contaminants to air.

Part V outlines the methods other than rules which will be used to achieve the Plan's objectives.

Part VI states the environmental results anticipated from the implementation of the policies and methods (including rules) in this Plan.

Part VII sets out the methods the Otago Regional Council may use when effects of discharges into air cross district or regional boundaries.

Part VIII sets out the processes to be used for monitoring and reviewing this Plan.

Although this Plan is structured into discrete parts, these parts should not be viewed in isolation as the Plan needs to be read as a whole. This Plan uses a system of cross-referencing to provide linkage between parts, and issues, objectives, policies and methods. Within this system:

- Each of the issues is referenced through to the relevant objectives and policies within Part III of the Plan;
- Each of the objectives is referenced through to the relevant policies; and
- Each of the policies is referenced through to the relevant rules in Part IV, and other methods in Part V.

This cross-reference system is for information purposes, to aid understanding of the integrated nature of the Plan and to assist users to read the Plan as a whole by highlighting potential links between related provisions.

2 Legislative and policy framework

2.1 The Resource Management Act 1991

The Resource Management Act provides the statutory framework for managing air quality, for controlling the actual or potential effects of land use activities on the environment (including air) and for controlling discharges into air.

Section 30 of the Resource Management Act allocates regional councils the principal role for the management of air quality. Regional councils are the consent authorities for activities that would otherwise breach Section 15 of the Act. Section 15 restricts the discharge of contaminants into air, as follows:

"15. Discharge of contaminants into environment-

- (1) No person may discharge any -
 - (a) Contaminant or water into water; or
 - (b) Contaminant onto or into land in circumstances which may result in that contaminant (or any other contaminant emanating as a result of natural processes from that contaminant) entering water; or
 - (c) Contaminant from any industrial or trade premises into air; or
 - (d) Contaminant from any industrial or trade premises onto or into land unless the discharge is expressly allowed by a rule in a regional plan and in any relevant proposed regional plan, a resource consent, or regulations.
- (2) No person may discharge any contaminant into the air, or into or onto land, from -
 - (a) Any place; or
 - (b) Any other source, whether moveable or not, in a manner that contravenes a rule in a regional plan or proposed regional plan unless the discharge is expressly allowed by a resource consent, or regulations, or allowed by Section 20 (certain existing lawful activities allowed)."

Section 15(1) means that any discharge of a contaminant into air from any industrial or trade premises in Otago is allowed only if it is expressly authorised by a permitted activity rule in this Plan, a resource consent or by regulations. Under Section 15(2) the opposite presumption applies to discharges into air from any other source. That is, unless there is a relevant rule, discharges from sources other than industrial or trade premises can take place without a resource consent from the regional council.

Without this Plan, discharges of contaminants into air from industrial or trade premises, no matter how minor, require resource consents, while significant discharges from other sources do not.

A key function of this Plan is to allow minor discharges into air from industrial and trade premises which are unlikely to have any significant adverse effects, and to regulate other discharges which may have significant adverse effects.

Air quality can also be affected by the use of land and by discharges into air or to land from that use. Section 31 of the Act allocates a role to city and district councils in terms of controlling any actual or potential effects of land use on the environment (including air). District plans may therefore contain provisions regulating land use to manage air quality. Such provisions must not be inconsistent with this Plan.

Under Section 43 of the Resource Management Act, the Government can issue national environmental standards. The Resource Management (National Environmental Standards Relating to Certain Air Pollutants, Dioxins, and Other Toxics) Regulations 2004 (NESAQ) applied nationally from 8 October 2004. The NESAQ have:

- Seven standards banning activities that discharge significant quantities of dioxins and other toxics into the air;
- Five standards for ambient (outdoor) air quality;
- A design standard for newly installed woodburners installed in buildings on properties less than 2 hectares in size; and
- A requirement for landfills over 1 million tonnes of refuse to collect greenhouse gas emissions.

Every regional, city and district council is required to enforce these standards.

Rules in this Plan and resource consents granted under this Plan cannot be more lenient than the requirements of the NESAQ, however in some circumstances they can impose more stringent standards, and may provide for matters not specified by the NESAQ. Existing resource consents to discharge contaminants to air prevail over the requirements of the NESAQ until a review of the consent (under Section 128(1)(ba) of the Resource Management Act) has been made to ensure consistency with the NESAQ.

2.2 Relationship to other resource management documents

This Plan only addresses the effects of discharges of contaminants into air. This means that any person discharging contaminants into air may also need to comply with or be consistent with:

- The Regional Policy Statement for Otago;
- Other regional plans prepared by the Otago Regional Council; and
- District plans prepared by the relevant city or district council. In Otago the relevant district council will be one of the following depending on the location of the activity: Dunedin City Council; Waitaki District Council; Central Otago District Council; Queenstown-Lakes District Council; or Clutha District Council.

In terms of the Otago Regional Council's own Regional Policy Statement and regional plans, the activity or operation will need to comply or be consistent with the provisions of the following documents:

- **Regional Policy Statement for Otago** (Operative 1998) The Regional Policy Statement provides an overview of the resource management issues of Otago. It also provides policies and methods for achieving integrated management of the region's natural and physical resources. In the Regional Policy Statement air is recognised as a resource of regional significance. The objectives, policies and methods of the Regional Policy Statement have been taken into account in the development of this Plan.
- **Regional Plan: Waste for Otago** (Operative 1997) This plan deals with solid waste management. It contains rules which require consents to be obtained for activities involving discharges of contaminants into air at contaminated sites, landfills (including farm landfills, clean-fill landfills, greenwaste landfills and offal pits) and composting and silage production.

- **Regional Plan: Coast for Otago** (Operative 2001) This plan covers the coastal marine area of Otago which extends from the line of mean high water springs out to the 12 nautical mile limit. The plan contains rules which deal with the following: public access and occupation of space; structures; alteration of the foreshore and seabed; discharges; taking, use, damming or diverting water; noise; exotic plants and natural hazards. It does not control discharges of contaminants into air.
- **Regional Plan: Water for Otago** (Operative 2004) This plan establishes a framework for the use, development and protection of Otago's rivers, lakes, aquifers and wetlands. It contains rules which require resource consents to be obtained for the: taking of water; damming or diversion of water; discharge of water into water; discharge of contaminants into water, or onto or into land in circumstances which may result in that contaminant entering water; use, erection, alteration, extension, removal or demolition of structures in the bed of any lake or river and the introduction of vegetation or deposition of any substance or drainage or reclamation in the bed of any lake or river.

2.3 Other legislation

Other legislation has implications for air quality, or activities which discharge contaminants into air. This legislation includes the following:

- Health Act 1956;
- Building Act 1991;
- Health and Safety in Employment Act 1992;
- Traffic Regulations 1976; and
- Hazardous Substances and New Organisms Act 1996.

It is the role of Otago's city and district councils to enforce the provisions of the Health Act and Building Act. The Department of Labour, the Land Transport Safety Authority and Ministry of Health have responsibilities under the Health and Safety in Employment Act, the Traffic Regulations and Health Act, respectively. The Environmental Risk Management Authority is responsible for implementing the Hazardous Substances and New Organisms Act.

3 Kai Tahu ki Otago: Air perspective

Section 8 of the Resource Management Act 1991 requires all people exercising functions and powers under it, in relation to managing the use, development, and protection of natural and physical resources, to take into account the principles of the Treaty of Waitangi. The Treaty guarantee of rangatiratanga to Kai Tahu should be reflected in the provision of adequate and appropriate opportunities for participation in air quality planning, management and monitoring.

Attitudes to resource use need to be reconciled with the relationships between iwi and te taiao, mauri, wairua, tapu, mana and rangatiratanga. Central to an understanding of these concepts is a knowledge of the Maori holistic values and the cosmogenic origins that make Kai Tahu part of the physical environment - they belong in it, and it does not belong to them. As kaitiaki they have an inherited responsibility to protect and preserve mauri and to conserve taoka.

Kai Tahu recognise the Regional Plan: Air as an important complementary part of the hierarchy of planning for the Otago Regional Council. The Regional Plan: Air is consistent with the cultural inter-relationships between the domains of Ranginui and

Papatuanuku, and the interaction between the key demi-gods Tane Mahuta, Tangaroa and Tawhirimatea.

The Kai Tahu perspective is guided by their belief in the tradition of Papatuanuku, Ranginui, Tawhirimatea, Tane Mahuta, and Tangaroa. The inseparable nature of these values means they cannot be addressed in isolation. What impacts on one impacts on all, a holistic perspective that is beginning to be recognised in the resource management process.

Following the separation of Ranginui and Papatuanuku, Tawhirimatea was the only child who objected to Ranginui's treatment and fled with him to his new home in the sky. From there he controls the elements and occasionally vents his spite on the world below. The activities of Tawhirimatea can be benign or spiteful, and it is in the nature of Maori ethos to ensure that nothing is done to enrage Tawhirimatea.

Kai Tahu believe that good health is the positive state of physical, mental, spiritual and social well being. Kai Tahu is concerned with the effect air quality has upon people's health. The Council must recognise that adverse effects on the air resource have the potential to harm people's health and well being.

Kai Tahu consider that when managing the air resource it is important to take into account the principle of kaitiakitanga – the duty to ensure that the air resource in Otago is managed with care and respect, for the benefit of current and future generations, and the health of the environment. The duty of the Crown to protect the iwi interests is an important factor.

The Ngai Tahu Claims Settlement Act 1998 (NTCSA) contains statements, called statutory acknowledgements, by Te Runanga o Ngai Tahu, of the particular cultural, spiritual, historic and traditional association of Ngai Tahu with specific areas in the Otago region.

The statutory acknowledgements and the areas to which they relate are set out in the Schedules of the NTCSA.

The effect of the statutory acknowledgements is set out in Part 12 of the NTCSA.

Under Section 208 of the NTCSA, local authorities must have regard to the statutory acknowledgements in deciding whether Te Runanga o Ngai Tahu is an affected person whose written approval must be obtained for a non-notified resource consent application, or must be served with a notified application for a resource consent.

Section 211 of the NTCSA enables Ngai Tahu to cite these acknowledgements in submissions or in proceedings before consent authorities or the Environment Court. A statutory acknowledgement is not binding on the consent authority or Court, but may be taken into account.

This note is for the purpose of public information only and does not form part of the Plan. It is required by Section 220 of the Ngai Tahu Claims Settlement Act 1998.

Part II Otago's Air Quality Issues



4 Introduction

This part of the Plan outlines the air quality issues in the Otago region. These issues have been identified through:

- Consultation during the development of the Regional Policy Statement for Otago;
- Evaluating issues raised in public comments on the Air Quality Discussion Paper; and
- Assessing the relevance of national and international air quality issues.

4.1 Kai Tahu ki Otago

4.1.1 Discharges into air may adversely affect:

- (a) Waahi tapu;
- (b) Waahi taoka;
- (c) Mahika kai; and
- (d) Marae.

Explanation

Kai Tahu ki Otago have a spiritual relationship with the air. Its creation is recognised by Kai Tahu tradition as the dawn of light. The air is an integral part of the environment which must be valued, used with respect, and passed on intact to the next generation.

Kaitiakitanga requires Kai Tahu to be actively engaged in the planning, consenting and monitoring of the air resource to ensure their holistic values are recognised. Because of the importance of the air resource to Kai Tahu, and in order to provide for kaitiakitanga, it is necessary to involve Kai Tahu in the management of the air resource.

When contaminants are discharged into air they may have an adverse effect on customary values and practices including waahi tapu, waahi taoka, mahika kai and marae. For example:

- Discharges into the air may adversely affect traditional sources of mahika kai thereby further reducing the ability to gather food;
- Discharges such as those from a crematorium can, if located in close proximity to waahi tapu or other taoka, be spiritually offensive; and
- Discharges into air can adversely affect health on a marae and can be culturally offensive.

Issues of particular concern, which stem from the Kai Tahu perspectives are:

- Air quality in Otago. To ensure the air resource is passed on intact to the next generation, Otago's air resource should be maintained and enhanced where it is degraded.
- **Discharges from industrial or trade premises** adversely affect local and ambient air quality and can affect papakaika and mahika kai.
- **Domestic discharges** are a concern because of their effect on mahika kai and other cultural uses.
- **Dust** can have an adverse effect on traditional Maori rock art and people's health.
- **Odour** may affect marae and waahi taoka.

- Agrichemical spray drift has the potential to cause adverse effects on people's health and non-target neighbouring areas. Of particular concern are site specific areas such as marae, waahi tapu, mahika kai, water and indigenous flora.
- Vegetation burning can have an adverse effect on waahi tapu, waahi taoka, mahika kai and marae.
- **Transport emissions** can have an adverse effect on public health, mahika kai, and on native flora and fauna used for natural medicines.

Other issues 4.2.1 to 4.10.1 Objectives 6.1.1, 6.1.2, 6.1.3 Policies 7.1.1, 8.2.3

4.2 Air quality in Otago

4.2.1 The discharge of contaminants into air can degrade ambient air quality.

Explanation

Ambient air quality is the general quality of the air that surrounds us and is a reflection of the cumulative impacts of discharges from human activities and natural processes.

There are many benefits of good ambient air quality, including:

- Avoiding costs in the health, recreation, built environment, commerce and natural environment areas;
- Enabling our quality of life to be maintained, and in particular those values relating to amenity and visibility; and
- Maintaining the perception of this country as a clean environment and thus enhancing its attractiveness to visitors.

In 1994, the Ministry for the Environment produced a document entitled "Ambient Air Quality Guidelines" (AAQG). The AAQG provides the minimum requirements that ambient air quality should meet in order to protect human health and the environment. These 1994 guidelines were updated in 2002. The updated AAQG contains new ambient air quality guideline values for both existing and newly listed contaminants, revised guidance on how they should be used to manage air quality under the Resource Management Act, and new guidance on assessing the potential impacts of air pollution on ecosystems.

In 2004 the NESAQ was gazetted, as discussed in section 2.1. The NESAQ provides baseline ambient air quality protection for all New Zealanders, the standards of which are based upon those given in the AAQG. Where pollutants are not covered by the NESAQ, those guidelines given in the AAQG still apply. The requirements of the NESAQ also override any less stringent requirements in regional plans.

The Otago Regional Council's ambient air quality monitoring programme, which commenced in 1997, has focused on measuring the levels of PM_{10} (particles smaller than ten microns), oxides of nitrogen, carbon monoxide and sulphur dioxide. A report produced by the Otago Regional Council in 2005 entitled "Ambient Air Quality in Otago 1997 – 2004, Nitrogen Dioxide, Sulphur Dioxide and Carbon Monoxide", summarises the air quality monitoring results for these

pollutants, and concludes that they are not causing a significant problem in Otago's urban areas. However, the monitoring programme is on-going to enable the Council to identify trends and further increase its understanding of air quality, and the effects of discharges of contaminants to air.

The NESAQ requires the ambient concentration of PM_{10} in all parts of New Zealand to meet 50 µg/m³ (24-hour mean) by 1 September 2013, with only one excedence of this standard allowed in a 12-month period. With regard to PM_{10} , a report produced by the Otago Regional Council in 2005 entitled "Ambient Air Quality in Otago 1997 – 2004, Particulate Matter" summarises air quality monitoring undertaken. Monitoring data has shown breaches of the NESAQ requirements for PM_{10} in a number of urban areas throughout the region, including Alexandra, Arrowtown, Cromwell, Dunedin, Milton, Mosgiel, Oamaru, and Palmerston.

The relative contributions of domestic, industrial, commercial and transport sources on air quality were investigated in 1999 and 2006 through an inventory of emissions for Otago's main urban areas. The results of the inventory show that during winter when measured PM_{10} concentrations have been high, emissions from domestic sources are the largest source of PM_{10} .

Outside of urban areas, ambient air quality in Otago is generally considered to be good.

Other issues 4.3.1 to 4.10.1 Objective 6.1.1 Policies 8.1.1, 8.1.2, 8.2.3, 8.2.4, 8.2.6, 8.2.8, 9.1.1, 9.1.2, 9.1.3, 9.1.4, 13.1.1, 14.1.1

4.3 Discharges from industrial or trade premises

4.3.1 Discharges into air from industrial or trade premises can adversely affect air quality.

Explanation

Discharges from industrial or trade premises may be point source (e.g., a chimney) or from an area source such as a storage yard. These discharges can adversely affect human health, amenity and heritage values and the physical and biological environments surrounding the discharges. The main factors giving rise to these effects include the presence of:

- Dust (area source discharges are addressed in Issue 4.5.1);
- Hazardous or toxic contaminants;
- Particulates; and
- Odour, which is a significant effect resulting from discharges from some industrial or trade premises and is addressed in Issue 4.6.1.

In terms of this Plan, discharges on industrial or trade premises do not include those from the engines of motor vehicles, trains or aircraft on these premises.

While most, if not all discharges of contaminants into air from industrial or trade premises will have some effect on air quality, the effects will vary, and therefore the level of control necessary to avoid, remedy or mitigate adverse effects will vary. It is therefore important to establish a level of control that is appropriate to the actual or potential effects of the discharge, and one which is at a similar level to the controls established for non industrial or trade discharges which have similar effects on the environment.

Other issues 4.5.1, 4.6.1, 4.9.1 Objectives 6.1.1, 6.1.2, 6.1.3 Policies 8.1.1 to 8.2.6, 8.2.8, 9.1.1, 9.1.2, 10.1.1, 11.1.1

4.4 Domestic heating and burning of waste

4.4.1 Domestic heating using wood and fossil fuel produces emissions that have the potential to adversely affect human health and amenity values in Otago's urban areas.

Explanation

The burning of wood and fossil fuels produces smoke, water vapour, carbon dioxide, carbon monoxide, nitrogen oxides, hydrocarbons and other volatile organic compounds. Sulphur oxides (primarily sulphur dioxide and sulphur trioxide) are also discharged from the burning of fuels containing sulphur, such as coal. Monitoring of levels of nitrogen, carbon monoxide and sulphur dioxide in ambient air in urban areas of Otago has been undertaken, and is documented in the 2005 report "Ambient Air Quality in Otago 1997 – 2004, Nitrogen Dioxide, Sulphur Dioxide and Carbon Monoxide". The report concludes that they are not causing a significant problem in Otago's urban areas.

In terms of the burning of wood and fossil fuels for domestic heating, the most significant effect of discharges in Otago is the accumulation of fine particles (smaller than 10 microns (PM_{10})) in the air that we breathe (ambient air). This is an issue because these particles can adversely affect human health as they enter the respiratory system and can cause loss of lung function, onset or aggravation of respiratory illness, and a loss of capacity to resist infection.

Monitoring of PM_{10} in ambient air in urban areas has been undertaken, and is documented in the 2005 report "Ambient Air Quality in Otago 1997 – 2004, Particulate Matter". This monitoring has shown that PM_{10} concentrations have reached or exceeded levels where adverse health effects may occur in many of Otago's urban areas. Emissions inventories undertaken in 1999 and 2006 for Otago's urban areas indicate that the majority of this PM_{10} comes from domestic heating discharges. In 1999, many urban areas were surveyed and such discharges were found to contribute an average of 75% of winter PM_{10} emissions, with contributions for individual urban areas ranging from 52% to 92%. In 2006, an emissions inventory was undertaken for Dunedin, Mosgiel and Alexandra. The main source of PM_{10} emissions in all areas during the winter was domestic home heating, which accounted for 90% of total PM_{10} emissions in Dunedin and Mosgiel, and 99% of total PM_{10} emissions in Alexandra.

Discharges from domestic heating can also result in adverse effects on amenity values in terms of odour, nuisance and visibility.

The adverse effects of discharges from domestic heating are caused, or exacerbated by, inefficient heating appliances, incorrect use of appliances and poor quality fuels. For example, open fires discharge more PM_{10} than multifuel or woodburners, which in turn discharge more PM_{10} than pellet fires, gas heaters or electricity. Poor quality fuels, such as coal with a high sulphur content or wood with a high moisture content, will produce more adverse effects than low sulphur coal or dry wood, which in turn produce more adverse effects than wood pellets or gas.

Objectives 6.1.1, 6.1.2, 6.1.3 Policies 8.1.1, 8.1.2, 8.2.7, 8.2.8, 9.1.1, 9.1.3, 9.1.4

4.4.2 Burning waste outdoors can produce emissions which have adverse localised effects beyond the boundary of the property where the burning occurs.

Explanation

The Otago Regional Council receives a significant number of complaints about outdoor burning. These are typically associated with burning in drums or on the open ground in backyard situations. In the three years to 30 June 1999 these accounted for 23% of air pollution complaints with the majority being made within the Dunedin City area.

The main effects from the burning of waste on residential properties are adverse localised effects which occur mostly in areas which have a higher density of settlement. These effects include:

- The generation of smoke which can cause a nuisance to neighbours by soiling property and reducing visibility;
- The emission of odours which can reduce the pleasantness and general amenity value of an area;
- Exposure to high concentrations of various combustion products which may cause adverse health effects such as respiratory irritation, an issue which can be significant for people with asthma or other respiratory illness; and
- The potential for hazardous air contaminants to be discharged. These contaminants commonly result from the burning of materials such as plastics, chemically treated wood and fabrics.

Discharges from the outdoor burning of waste can also accumulate in the ambient air and increase contaminant concentrations (along with contaminants from other sources) to levels where adverse human health effects occur. An emissions inventory undertaken for Otago's urban areas in 1999 has shown that only a small proportion (less than 1%) of contaminants originate from the burning of waste on residential properties. Because of this, their cumulative effects are considered to be relatively insignificant. Instead, it is the adverse localised effects that occur from such burning, rather than its effects on ambient air quality, that are the significant concern.

Such effects can result from both open burning and incineration. Outside the main urban areas, a more dispersed population and good management practices can assist in mitigating any adverse effects. Note that Otago's city and district councils may also implement bylaw controls on the burning of waste for the purpose of reducing fire risk.

Objectives 6.1.1, 6.1.2, 6.1.3 *Policies* 8.2.6, 8.2.7, 8.2.8

4.5 **Dust from area sources**

4.5.1 Dust emissions from area sources can be a nuisance.

Explanation

Area sources of dust are those where the dust is not emitted from a particular point such as a chimney. The main area sources of dust in the Otago region are:

- Land and soil disturbance associated with subdivision development, construction activities, land clearance and cultivation;
- Unsealed roads, driveways or yards;
- Gravel extraction and mining; and
- Natural sources such as dry river and lake beds.

The adverse environmental effects of dust from these sources are primarily associated with nuisance and can include unwanted deposition of dust on property, reduced visibility and diminished amenity values. Health impairment may also occur, particularly when small particles are present.

Once emitted, dust from area sources is difficult to control because the emissions are generally intermittent and difficult to contain. Managing the adverse effects of dust discharges is therefore best achieved through avoiding dust emissions by using appropriate land management practices. Thus, while this Plan advocates the use of appropriate land management practices to avoid dust emissions, and contains rules to control activities on industrial or trade premises that have the potential to result in significant dust emissions, it is envisaged that the two primary means by which such controls will be implemented will be via the district plans of Otago's city and district councils and by conditions on land use consents granted by city and district councils. Dust controls may also form part of industry codes of practice or other mechanisms which promote good management practices.

Dust emissions from point sources are commonly related to specific activities undertaken on domestic, industrial or trade premises. As such, the effects of point source discharges are addressed via other issues in this Plan.

Objectives 6.1.1, 6.1.2, 6.1.3 Policy 10.1.1

4.6 Odour

4.6.1 The discharge of odorous compounds can have significant adverse effects on amenity values and human health.

Explanation

Odour is a high profile air quality issue in the Otago region, accounting for 33% of air pollution complaints received by the Otago Regional Council in the year ending 31 December 1997. The main sources giving rise to these complaints are:

- Abattoirs, fellmongers and rendering plants;
- Solid waste disposal, e.g., landfills and transfer stations;
- Fish processing plants;
- Sewage and waste water treatment systems;
- Commercial food processing operations; and
- Farming activities such as dairy shed effluent treatment and disposal, silage making and intensive farming.

Over 80% of odour complaints relate to activities in the Dunedin and Mosgiel areas. This appears to be because of the concentration of industry in these areas and the intensive nature of development which often produces situations where incompatible land uses are located in close proximity to one another. Incompatible land uses can exacerbate the effects of odour discharges by permitting the establishment of activities that produce odorous discharges and odour-sensitive activities in the same area.

The main effects of odour are on amenity values and effects on human health. At its extreme, on-going exposure to an offensive odour can create tensions between the producers of the odour and people exposed to the odour.

Odour, possibly more than any other contaminant discharge, directly affects the amenity values of an area. These amenity values are often described in subjective terms by communities and this can make it difficult to identify the effects that odour may have on them.

Odour itself is also subjectively defined. People respond to smells differently. While one person may detect an odour associated with a contaminant at a low concentration, others may not be able to detect it until it is much stronger. Once detected, people make judgements on the pleasantness or otherwise of an odour based on their individual experience.

While there are a small number of odours that almost all people would agree are offensive, there are a wide range of responses within the population. For example, silage may be considered by some to have a 'sweet' or 'nice' odour, while others may consider the odour to be quite offensive.

It is this subjectivity which makes it both difficult to measure odour objectively and to assess its effects. This means that the management of odour is often contentious and difficult. Odour management should initially concentrate on avoiding the occurrence of odour problems through effective planning, good process design and control, and general site management. Where it is not possible to avoid odour problems, a variety of management options exist and the most appropriate option, given the characteristics of the site and the odour, can be implemented to minimise the adverse effects. The Ministry for the Environment has produced a report entitled "Odour Management Under the Resource Management Act" (1995), which sets out considerations for regional, city and district councils for the management of odour.

Objectives 6.1.1, 6.1.2, 6.1.3 Policies 8.2.1, 8.2.3, 8.2.4, 8.2.5, 11.1.1

4.7 Agrichemical spray drift

4.7.1 The application of agrichemicals has the potential to cause adverse effects on non target areas or species as a result of spray drift.

Explanation

The use of agrichemicals to control plant pests, insect pests and fungal diseases is common throughout Otago. Such chemicals are used in horticulture, agriculture, arable farming, forestry and fruit production. Agrichemicals are also used in the management of public amenity areas and in residential gardens.

While problems associated with the application of agrichemicals are not widespread in Otago, there is the potential for adverse effects to occur as a result of sprays drifting onto non-target areas and onto neighbouring properties, thus creating conflict amongst neighbours. Changing land use patterns in some areas of Otago, due to the development of rural residential living and viticulture, may also increase the potential for such conflict to occur. Spray drift may have adverse effects on amenity values, natural resources, human health, economic well being and sensitive areas or places.

Areas or places can be sensitive for two main reasons. Firstly, the area or place may contain something that is particularly susceptible to damage from the agrichemicals being used. Secondly, the area or place might be sensitive because people consider agrichemical use to be inappropriate in that location.

The main factors influencing the occurrence of agrichemical spray drift are weather conditions, whether buffer zones exist between the intended spray area and non-target areas, the application method, and the frequency and duration over which agrichemicals are applied.

Many users of agrichemicals have recognised the need for better management of agrichemical application and control over the use of agrichemicals. In particular, the New Zealand Agrichemical Education Trust runs "Growsafe" courses to certify users of agrichemicals, and has produced the "*Code of Practice for the Management of Agrichemicals*" as a New Zealand Standard (NZS 8409:1999).

Objectives 6.1.2, 6.1.3 Policies 8.2.1, 8.2.2, 8.2.3, 8.2.4, 8.2.5, 8.2.8, 12.1.1

4.8 Vegetation burning on production land

4.8.1 Smoke generated by the burning of vegetative matter on production land can reduce visibility and cause adverse effects including nuisance.

Explanation

Land clearance and vegetation control by burning is a common practice on production land where activities such as horticulture, agriculture, arable farming, forestry and fruit production are undertaken. Where good burning practices are followed, any adverse effects are usually short lived, however when they are not followed, smoke may be discharged over a prolonged period of time. This can reduce visibility and cause adverse effects. It is of particular concern when the burning occurs in the vicinity of urban areas, lifestyle properties or areas where people participate in recreational activities such as on ski fields, and within national parks and reserves.

Discharges from the burning of vegetative matter in backyards can also have adverse effects but, as these effects are different, they are dealt with in Issue 4.4.2 in the Plan.

The burning of vegetative matter may also have effects on land and water resources. These effects will be addressed in other regional plans. The fire risk associated with land clearance is controlled by the Forest and Rural Fires Act 1977. This act is primarily administered by Otago's city and district councils, however the Commissioner for Crown Lands and the Department of Conservation administer the act in relation to Crown land.

Objectives 6.1.1, 6.1.2, 6.1.3 Policy 8.2.8, 13.1.1

4.9 Transport emissions

4.9.1 Emissions from transport sources have the potential to adversely affect human health and amenity values.

Explanation

Within Otago, emissions from motor vehicles such as cars, buses and trucks are the most dominant form of air pollution from transport sources. Emissions from ships and aircraft, however, can have intermittent and localised effects in and around Port Chalmers, the Port of Dunedin and Dunedin and Queenstown airports. It is impractical to manage the emissions from ships and aircraft on a regional basis because they only visit the Otago region for short periods and cause very localised effects. Integrated management of these emissions on a national and international basis is more effective, and as such they are not controlled by this Plan.

In 1996 there were 68,748 motor vehicles recorded as being available for private use in Otago, with 62 % (or 43,014) of these in Dunedin City (Statistics New Zealand, 1996 Census).

Motor vehicles discharge large quantities of contaminants into the air as exhaust fumes. These emissions include nitrogen oxides, carbon monoxide, carbon dioxide, unburned hydrocarbons, and volatile organic compounds such as benzene and formaldehyde. The nature and rate of emissions depends on: the type of fuel used; the efficiency of the engine or state-of-tune; whether or not emission treatment technologies are employed; driving style; the types of vehicles on the road; traffic levels and measures adopted to manage traffic flows.

Discharges from motor vehicles can:

- Have cumulative adverse effects on the global air resource since motor vehicle emissions are one of the main sources of "greenhouse" gases;
- Contribute to the creation of a brown haze or smog in urban areas which has the effect of reducing visibility and amenity values; and
- Contribute to high concentrations of contaminants in urban areas, particularly within "traffic corridors". In some circumstances the concentrations can reach levels where human health can be adversely affected.

Dunedin's central shopping area is the area most susceptible to adverse effects arising from motor vehicle emissions. This is because it has the highest levels of traffic congestion in Otago combined with a complex topography of poorly ventilated street canyons (tall buildings), a proximity to heavily trafficked roads (such as the state highways) and adverse meteorological conditions, such as the evening temperature inversion in the winter.

Objective 6.1.1, 6.1.2, 6.1.3 Policy 14.1.1

4.10 Global issues

4.10.1 The discharge of contaminants into air in Otago may have global effects.

Explanation

Air movement is not constrained by physical boundaries. The discharge of contaminants into air in Otago can therefore have effects on global air quality. Similarly global air quality can have regional effects.

Global issues of most significance for Otago are:

- The depletion of the ozone layer by substances such as chlorofluorocarbons (typically used in refrigeration), halons (used in fire fighting equipment) and methyl bromide (a fumigant); and
- Climate change through the emission of 'greenhouse gases' such as carbon dioxide, methane and water vapour from activities such as the burning of fossil fuels, motor vehicle emissions and methane gas from agricultural production.

Depletion of the ozone layer increases the levels of ultra violet radiation. This radiation is potentially harmful to human health and living tissue. It can lead to an increased incidence of skin cancer and eye damage, and may also affect plant growth and animal health.

PART II OTAGO'S AIR QUALITY ISSUES

Greenhouse gases insulate the earth and maintain global temperatures at constant levels. These gases have increased in concentration over the last 130 years. While there is still uncertainty about the consequences that increased concentrations may have, it is expected that sea level will rise and weather patterns will be modified.

The New Zealand Government is a signatory to international agreements including the Kyoto Protocol, the Vienna Convention for the Protection of the Ozone Layer and the Montreal Protocol on Substances that Deplete the Ozone Layer. The government has primary responsibility for developing and implementing national responses to global air quality issues.

In the Regional Policy Statement for Otago, the Council adopted policies to support initiatives aimed at avoiding, remedying and mitigating the production and discharge of greenhouse gases and ozone depleting substances.

While global effects are the primary responsibility of central government, the policies, methods and rules in this Plan will assist in achieving national objectives, through controlling the discharge of contaminants into air.

Objectives 6.1.1, 6.1.3 *Policy* 15.1.1

Part III Air Quality Management: Objectives and Policies



PART III AIR QUALITY MANAGEMENT

5.1 Management framework

This part of the Plan outlines the framework for managing air quality and discharges of contaminants into air. This framework is based on the following:

- Establishing objectives for air quality in the region; and
- Determining the most effective policies for addressing each of the issues identified in Part II.

6.1 **Objectives**

6.1.1 To maintain ambient air quality in parts of Otago that have high air quality and enhance ambient air quality in places where it has been degraded.

Explanation

This objective seeks to maintain ambient air quality where it is high rather than let it degrade, and to enhance air quality where it has been degraded to the point that the air resource is not being sustainably managed.

Principal reasons for adopting

This objective will ensure management of the region's air resource is consistent with Objective 7.4.1 of the Regional Policy Statement for Otago. The objective also reflects the community's desire for high ambient air quality throughout the region at all times.

This objective is implemented by all of the policies in this Plan.

6.1.2 To avoid adverse localised effects of contaminant discharges into air on:

- (a) Human health;
- (b) Cultural, heritage and amenity values;
- (c) Ecosystems and the plants and animals within them; and
- (d) The life-supporting capacity of air.

Explanation

This objective recognises that many air management issues relate to the localised effects of discharges, such as the effects of odour and smoke as opposed to effects on the region's ambient air quality. It seeks to ensure that the localised effects of the contaminants discharged into air do not have adverse effects on human health, cultural, heritage and amenity values, and the life supporting capacity of air.

Principal reasons for adopting

This objective is adopted to ensure that discharges do not have localised adverse effects on the environment.

This objective is implemented by all the policies in this Plan

6.1.3 To allow for the sustainable use of Otago's air resource.

Explanation

The primary function of the Plan is to provide for the sustainable use and protection of the region's air resource. This objective recognises that people have traditionally made extensive use of the air resource and will continue to do so in the future.

Principal reasons for adopting

This objective is adopted to ensure continued access to Otago's air resource for a range of existing and new uses, recognising the need to provide for the social, economic and cultural well being of Otago's people and communities.

This objective is implemented by all of the policies in this Plan

7.1 Policy for Kai Tahu ki Otago

7.1.1 To recognise and provide for the relationship Kai Tahu have with the air resource through procedures that enable Kai Tahu to participate in management of the air resource.

Explanation

Within Otago, Kai Tahu have cultural and spiritual values that can be adversely affected by the discharge of contaminants into air. In order to recognise Kai Tahu's relationship with the air resource and the potential for cultural and spiritual values to be adversely affected, it is essential to ensure that Kai Tahu participate in the management of the resource.

Principal reasons for adopting

Part II of the Resource Management Act recognises the special importance of iwi values. In order for the Council to achieve its responsibilities under the Act, it is necessary to ensure that iwi are involved in the management of Otago's air resource. This relationship is also recognised and provided for by Policy 7.5.1 of the Regional Policy Statement for Otago.

Rules 16.3.1.7, 16.3.2.6, 16.3.3.2, 16.3.4.3, 16.3.5.9, 16.3.6.3, 16.3.7.3, 16.3.8.2, 16.3.9.4, 16.3.10.3, 16.3.11.2, 16.3.14.1, 16.3.15.2 to 16.3.15.5

8.1 Policies for ambient air quality

8.1.1 To have regard to the Otago Goal Levels identified in Schedule 1 and comply with the Resource Management (National Environmental Standards Relating to Certain Air Pollutants, Dioxins and Other Toxics) Regulations 2004 in managing the region's ambient air resource.

Explanation

Otago Goal Levels are the levels of contaminants in air that are aimed for in parts of Otago where air quality may be degraded, particularly urban areas. The aim is to ensure as much of Otago as possible has ambient contaminant concentrations below these levels. The levels should not, however, be regarded as a limit up to which it is acceptable to allow pollution to reach in those areas with good air

PART III AIR QUALITY MANAGEMENT

quality where contaminant concentrations are currently well below these levels. Concentrations exceeding these levels are considered to be in the "alert" category as defined by the Ministry for the Environment in *Environmental Performance Indicators: Proposals for Air, Freshwater and Land* (1997). These are warning levels set at between 66 and 100% of the guideline values that can lead to the AAQG or NESAQ being exceeded if trends are not curbed.

Should monitoring results indicate that contaminant concentrations are approaching or exceed Otago Goal Levels, the Council will undertake research to identify the sources of contaminants and determine whether emissions and concentrations are likely to increase over time. The results of such research will then be used to develop and implement an appropriate management response in consultation with the community.

The Otago Goal Levels are concerned with the cumulative impacts of discharges into air from human activities and natural processes. Because of this it is generally inappropriate to apply them directly to set emission limits or determine consent monitoring requirements for individual discharges. They can, however, be used as a factor in determining the duration of a consent and assessing cumulative effects.

The NESAQ is mandatory and provides for a specific range of prohibitions and restrictions on discharges from certain activities, and where practicable, these have been incorporated within the policy and rule framework of this Plan.

These regulations include prohibitions (with some exceptions) on the granting of resource consents for certain activities; lighting of fires and burning of waste at landfill; burning of tyres, bitumen, coated wire and oil; the operation of incinerators at school or healthcare institutions unless resource consent has been granted for the discharge; and the operation of high-temperature hazardous waste incinerators.

The NESAQ also sets ambient air quality standards for five contaminants; regulates aspects of granting or declining resource consents to discharge PM_{10} , carbon monoxide, oxides of nitrogen, volatile organic compounds and sulphur dioxide; and sets specific standards for woodburners and for the control of greenhouse gas emissions at landfills.

Principal reasons for adopting

The Resource Management Act requires an effects-based approach to managing the air resource. Such an approach requires guidelines against which to measure the cumulative effects of activities. This enables problem areas to be targeted and the effectiveness of management strategies to be monitored.

Because Otago's community has identified the maintenance and enhancement of air quality as the key management objective, Otago Goal Levels have been adopted which set higher standards than those contained in the AAQG and the NESAQ.

The incorporation, where practicable, of the NESAQ into this Plan's objectives, policies and rules enables an integrated approach to management of air quality by identifying in one regulatory document, all relevant restrictions and prohibitions on discharges to air.

Other policies 8.1.2, 8.2.3, 8.2.4, 8.2.6, 8.2.8, 9.1.1, 9.1.2, 9.1.3, 9.1.4, 13.1.1, 14.1.1 Rules 16.3.1.7, 16.3.2.6, 16.3.3.2, 16.3.4.3, 16.3.5.9, 16.3.6.3, 16.3.7.3, 16.3.8.2, 16.3.9.4, 16.3.10.3, 16.3.11.2, 16.3.14.1, 16.3.15.2 to 16.3.15.5

8.1.2 To manage ambient air quality by airsheds and air zones.

Explanation

The Otago region comprises a number of areas specified as separate airsheds by the Minister for the Environment through notice in the New Zealand Gazette, as well as an airshed comprising the balance of the region.

Twenty-two Otago airsheds have been gazetted, representing Otago's main urban areas. The 23rd airshed comprises the balance of the Otago region. In accordance with the definition of airsheds in the NESAQ, the balance of the region is an airshed by default. The maps of the airsheds are given in Schedule 2. The Otago Regional Council monitors air quality in each of these airsheds where it is likely that the ambient air quality standards set by the NESAQ will be breached.

For simplicity of management, the twenty-two airsheds have been categorised into three Air Zones, as follows:

- Air Zone 1: including Alexandra, Arrowtown, Clyde and Cromwell airsheds.
- Air Zone 2: including Balclutha, Dunedin, Green Island, Hawea, Kingston, Milton, Mosgiel, Naseby, Oamaru, Palmerston, Port Chalmers, Queenstown, Ranfurly, Roxburgh, Waikouaiti and Wanaka airsheds.
- Air Zone 3: the whole of Otago, excluding the areas that are zoned 1 or 2.

The Clyde airshed is identified as a sub-zone within Air Zone 1 until 1 April 2009, pending further monitoring and an analysis of how discharges to air within the Clyde airshed affect the Alexandra airshed.

Principal reasons for adopting

This policy is adopted to enable appropriate local responses to comply with regulations 15, and 17 to 21 of the NESAQ.

Other policies 8.1.1, 8.2.1, 8.2.3, 8.2.4, 8.2.6, 9.1.1 to 9.1.4 Rules 16.3.1.1 to 16.3.1.7, 16.3.2.1 to 16.3.2.6, 16.3.4.1 to 16.3.4.3, 16.3.5.1 to 16.3.5.3, 16.3.5.9, 16.3.14.1, 16.3.15.1 to 16.3.15.5

8.2 General policies for managing discharges of contaminants into the air

8.2.1 To allow discharges of contaminants into air from industrial or trade premises that have minor or no adverse effects without the need for a resource consent.

Explanation

Section 15(1) of the Resource Management Act 1991 allows discharges of contaminants into air from industrial or trade premises only when the discharge is expressly authorised by a permitted activity rule, a resource consent or by regulations. This policy establishes a framework which allows discharges into air from industrial or trade premises, without consent when the Council is satisfied that there will be no adverse effects or that any adverse effects will be minor in nature.

Principal reasons for adopting

This policy avoids unnecessary regulation of discharges from industrial or trade premises that are unlikely to result in any significant adverse effects on the environment.

Rules 16.3.1.2 to 16.3.1.6, 16.3.2.2, 16.3.2.3, 16.3.2.5, 16.3.4.1, 16.3.4.2, 16.3.5.1 to 16.3.5.8, 16.3.6.1, 16.3.6.2, 16.3.7.1, 16.3.9.2, 16.3.10.1, 16.3.10.2, 16.3.11.1, 16.3.13.1, 16.3.13.2 Methods 17.5.2.1, 17.5.2.2

8.2.2 Priority will be given to avoiding adverse effects from hazardous air contaminants identified in Schedule 3 when considering the effects of any discharge of contaminants into air.

Explanation

The contaminants listed in Schedule 3 have been identified by the Ministry for the Environment in the *Ambient Air Quality Guidelines* (1994) as either being known or suspected to cause acute human health effects or adverse effects on the environment. While the Ministry did not identify threshold concentrations at which these effects occur, it advocated a precautionary approach to minimise their emissions.

Accordingly when assessing applications to discharge hazardous air contaminants, the Otago Regional Council will expect applicants to demonstrate that measures have been taken to avoid the occurrence of adverse effects. Such measures could include but are not limited to:

- Employing processes that do not discharge hazardous contaminants;
- Modifying processes so as to reduce the mass of hazardous contaminants discharged;
- Installing emission control equipment; or
- Using site planning mechanisms and other land use management techniques to reduce the potential for off-site effects.

Principal reasons for adopting

This policy recognises that discharges of hazardous air contaminants can have adverse effects on human health and the environment. The policy states that the Otago Regional Council's priority is the avoidance of such adverse effects.

Other policies 8.2.3, 8.2.4, 8.2.5 Rules 16.3.1.7, 16.3.2.6, 16.3.3.2, 16.3.4.3, 16.3.5.9, 16.3.6.3, 16.3.7.3, 16.3.8.2, 16.3.9.4, 16.3.10.3, 16.3.11.2, 16.3.14.1 Methods 17.5.2.1, 17.5.2.2

- 8.2.3 In the consideration of any application to discharge contaminants into air, Council will have:
 - (a) Particular regard to avoiding adverse effects including cumulative effects on:
 - (i) Values of significance to Kai Tahu;
 - (ii) The health and functioning of ecosystems, plants and animals;
 - (iii) Cultural, heritage and amenity values;
 - (iv) Human health; and
 - (v) Ambient air quality of any airshed; and
 - (b) Regard to any existing discharge from the site, into air, and its effects.

Explanation

This policy sets out those matters which will be given consideration in the resource consent process. It is not intended to limit those matters that the Council must consider in terms of Section 104 of the Resource Management Act.

When considering applications for consents to discharge contaminants into air, particular regard must be had to avoiding adverse effects (including cumulative effects) on values identified in (a). Where adverse effects are considered to be unavoidable, a resource consent may be declined or, if granted, be subject to conditions requiring the adverse effects to be remedied or mitigated.

Cumulative effects may arise over time as a consequence of the grouping of discharges. Their consideration will include having regard to the Otago Goal Levels identified in Schedule 1.

Item (i) explicitly provides for consideration of the effects of discharges on values of significance to Kai Tahu and complements Policy 7.1.1.

Items (ii) to (iv) will ensure consideration of the localised and ambient effects of discharges on the environment.

In assessing whether the proposed discharge is likely to increase significantly the concentration of PM_{10} in the airshed, the Council will have regard to:

- The characteristics of the PM₁₀ discharge, including how much PM₁₀ is to be discharged, and the frequency, rate and manner of the discharge;
- The nature of the receiving environment, including local climatic and geographic conditions and any surrounding environmental conditions that may affect the frequency, duration, intensity and degree of environmental effects;

- To what extent the proposed PM_{10} discharge contributes to the airshed breaching, or being at risk of breaching, the applicable curved or straight line path or the ambient air quality standard; and
- Whether any offset of PM_{10} is proposed, and the extent to which the offset mitigates the proposed discharge.

Part (b) is included to ensure that the past and present standard of the discharge of contaminants at that site forms part of the consideration of the consent authority, where an application is made to continue the discharge.

Principal reasons for adopting

This policy provides guidance on the effects that should be given consideration when making decisions on applications for resource consent.

Other policies 8.2.2, 8.2.4, 8.2.5 Rules 16.3.1.7, 16.3.2.6, 16.3.3.2, 16.3.4.3, 16.3.5.9, 16.3.6.3, 16.3.7.3, 16.3.8.2, 16.3.9.4, 16.3.10.3, 16.3.11.2, 16.3.14.1, 16.3.15.2 to 16.3.15.5 Methods 17.5.2.1, 17.5.2.2

8.2.4 The duration of any permit issued to discharge contaminants into air will be determined having regard to:

- (a) The mass and nature of the discharge;
- (b) The nature and sensitivity of the receiving environment; and
- (c) Any existing discharge from the site, into air, and its effects.

Explanation

Permits to discharge contaminants may be granted for a duration of up to 35 years under the Resource Management Act, depending on the nature of the discharge and the receiving environment. Where the discharge has the potential for more than minor adverse effects, the duration of the permit may be set at less than the maximum term allowable.

While other policies in the Plan provide the primary means for assessing the effects of discharges, this policy identifies additional matters to be considered in relation to determining the duration of permits.

All assessments to establish the appropriate duration of a permit will be undertaken on a case by case basis having regard to the mass and nature of the discharge and the nature and sensitivity of the receiving environment. Council will take into consideration any existing discharge at the site and its effects. Council will set consent durations on an equitable basis to attain the environmental objectives of this Plan.

Principal reasons for adopting

This policy is adopted to give guidance in determining the appropriate duration of any consent to discharge contaminants into air.

The policy recognises that the maintenance and enhancement of existing ambient air quality within parts of Otago cannot be achieved if all permits are issued for the maximum duration of 35 years allowed under the Resource Management Act. It encourages new and existing dischargers to take measures to ensure that the
effects of their discharges are no more than minor in order to obtain permits for a longer duration. The guidance provided by the policy will also ensure that consent durations are applied on an equitable basis depending on the mass and nature of the discharge, the nature and sensitivity of the receiving environment, and the standard of discharges into air at that site.

Other policies 8.2.2, 8.2.3, 8.2.5 Rules 16.3.1.7, 16.3.2.6, 16.3.3.2, 16.3.4.3, 16.3.5.9, 16.3.6.3, 16.3.7.3, 16.3.8.2, 16.3.9.4, 16.3.10.3, 16.3.11.2, 16.3.14.1, 16.3.15.2 to 16.3.15.5

8.2.5 To require, as appropriate, that provision be made for review of the conditions of any resource consent to discharge contaminants into air.

Explanation

Resource consents to discharge contaminants into air are issued with the expectation that there will not be any significant adverse effects on air quality. There are occasions, however, when consented discharges may have unforeseen adverse effects on air quality that are considered noxious, dangerous, offensive or objectionable. This may be because the production process or emissions control equipment is not performing as expected. In these circumstances, a review of the resource consent conditions may become necessary before the end of the consent term. Reviews of this nature are provided for by Section 128(1)(a)(i) of the Resource Management Act. To enable this review, a condition will be included as appropriate, on any resource consent to discharge contaminants into air.

Principal reasons for adopting

This policy is adopted to inform the public that there is legislative provision to review the conditions of resource consents during their term to address any unforeseen adverse effects on air quality resulting from the exercise of the resource consent. Where such a resource consent has been issued for a long term, it may be unacceptable to wait for the end of its term to deal with any problems that have arisen.

Other policies 8.2.2, 8.2.3, 8.2.4 Rules 16.3.1.7, 16.3.2.6, 16.3.3.2, 16.3.4.3, 16.3.5.9, 16.3.6.3, 16.3.7.3, 16.3.8.2, 16.3.9.4, 16.3.10.3, 16.3.11.2, 16.3.14.1, 16.3.15.2 to 16.3.15.5

8.2.6 To control the effects from outdoor burning by:

- (a) Enabling burning to occur providing it does not have significant adverse effects beyond the boundary of the property where burning occurs;
- (b) Restricting the type of materials that may be burnt, and controlling the means of burning, to minimise the discharge of hazardous air contaminants identified in Schedule 3; and
- (c) Applying separation distances for outdoor burning of waste in Air Zones 1 and 2, taking into account the potential for burning to occur in close proximity to neighbours, the likely frequency of burning and the likely volumes of waste involved.

PART III AIR QUALITY MANAGEMENT

Explanation

This policy introduces three restrictions on outdoor burning. These are applied to both burning in or on the open ground, and incineration of waste and other materials, on residential and non-residential properties.

Part (a) indicates the Otago Regional Council's view that people undertaking outdoor burning need to adopt practices to avoid significant adverse effects beyond the boundary of the property where burning is being undertaken.

Part (b) restricts the type of materials that may be burnt by outdoor burning without a consent to waste paper, cardboard, vegetative matter and untreated wood. This is given effect to by rules ensuring that outdoor burning which has the potential to result in the discharge of hazardous air contaminants identified in Schedule 3 be carried out only in a consented incinerator, and is carefully assessed on a case-by-case basis with appropriate controls being applied. Open burning of "specified materials" as listed in Rule 16.3.3.1 is prohibited.

Part (c) recognises that proximity to a fire largely determines whether or not a discharge will have significant adverse localised effects. It recognises that because Air Zones 1 and 2 have higher population densities, the effects of the burning of waste in these areas are more significant than elsewhere in Otago. This is supported by the high number of complaints received by the Council concerning outdoor burning in these areas. Separation distances between the source of the discharge and property boundaries are needed which are related to the likelihood and significance of impacts on neighbours.

Principal reasons for adopting

This policy is adopted to avoid or mitigate the adverse effects that outdoor burning can have on people and the environment.

Restrictions on materials that may be burnt out-of-doors are adopted to reduce the potential effects of PM_{10} and the hazardous air contaminants identified in Schedule 3. The contaminants identified in Schedule 3 have been identified by the Ministry for the Environment as either being known or suspected to cause acute human health effects or significant adverse effects on the environment. Open burning of specified materials not in a consented incinerator is likely to be inefficient and thus result in the discharge of such hazardous contaminants.

Separation distances are adopted in order to avoid adverse localised effects on neighbours. Such adverse effects result from the soiling of property, the discharge of odours and smoke, reduced visibility and reductions in the pleasantness or amenity of an area, arising from outdoor burning on the open ground or in an inefficient incinerator.

Other policy 8.1.2 Rules 16.3.2.1 to 16.3.2.6, 16.3.3.1, 16.3.3.2, Methods 17.4.1.1, 17.4.2.1 to 17.4.2.3

- 8.2.7 To promote voluntary actions to assist in avoiding adverse effects from the discharge of contaminants into air from:
 - (a) The outdoor burning of waste; and
 - (b) Domestic heating using domestic heating appliances.

Explanation

Advocacy and publicity by the Otago Regional Council can raise public awareness of the adverse effects that can arise from the incineration and open burning of waste, and the use of solid fuels for domestic heating. Such awareness can result in resource users taking voluntary actions to reduce the adverse effects of discharges. The advocacy and publicity initiatives forwarded in this policy are intended to support the regulatory methods for managing the effects of these discharges. These regulatory methods are outlined in Policies 8.2.6 (burning of waste) and 9.1.1 (domestic heating) of the Plan.

Principal reasons for adopting

This policy recognises the use of advocacy and public awareness as important methods for informing the public about the adverse effects associated with discharges of contaminants into air from the burning of waste and domestic heating.

Methods 7.6.17 and 7.6.18 of the Regional Policy Statement for Otago identify the use of education and public information campaigns as means of improving community awareness and understanding of air issues, and the adverse effects associated with discharges of contaminants into air. While non-regulatory approaches alone are not effective in avoiding or mitigating the adverse effects of discharges from the incineration and open burning of waste and domestic heating, they are an important tool in improving awareness of the associated issues and regulatory methods being used.

Other policies 8.2.1, 8.2.6, 8.2.8, 9.1.4 Methods 17.4.1.1, 17.4.2.1 to 17.4.2.3

8.2.8 To avoid discharges to air being noxious, dangerous, offensive or objectionable on the surrounding local environment.

Explanation

Irrespective of any other control on discharges, a condition will be placed on all relevant permitted activities to prevent, where necessary, any noxious, dangerous, offensive or objectionable effects at or beyond property boundaries. The identification of these effects is discussed in section 16.2.8.

Principal reasons for adopting

This policy is adopted to prevent noxious, dangerous, offensive or objectionable discharges to air.

This policy is implemented by all of the permitted activity rules in this Plan

PART III AIR QUALITY MANAGEMENT

9.1 Policies for reducing discharges of PM₁₀

9.1.1 To reduce discharges of PM_{10} to comply with the ambient air quality standard for PM_{10} by 1 September 2013, in Air Zone 1 by following curved line paths to compliance, and in specified airsheds in Air Zone 2 by following straight line paths to compliance.

Explanation

By 1 September 2013, the NESAQ requires the concentration of PM_{10} in any airshed to meet an ambient air quality standard of 50 µg/m³ (24-hour mean) with only one excedence allowed in a 12-month period.

For airsheds where the concentration of PM_{10} in ambient air exceeds the standard, regulation 17 (including 17A – 17D) of the NESAQ allows a regional council to select either a "straight line path" or a "curved line path" to achieve compliance by 1 September 2013. A straight line path means that a consistent reduction in PM_{10} discharges occurs from 1 September 2005, whereas a curved line path allows time for measures to be implemented that will achieve a reduction in PM_{10} discharges. Examples of both straight and curved line paths to compliance are shown in Figure 2.

Figure 2: Examples of Straight and Curved Line Paths to Compliance by 1 September 2013



Note: This graph is for illustrative purposes only and does not represent a path to compliance for any airshed within Otago.

A curved line path to compliance is intended to achieve the reduction of PM_{10} in airsheds in Air Zone 1, as given in Table 2. The start-point values apply from 14 April 2007 and reflect the typical level of wintertime PM_{10} observed in that airshed. As domestic heating appliances contribute significantly to PM_{10} concentrations in airsheds, Rules 16.3.1.1, 16.3.1.2, 16.3.1.5 and 16.3.1.6 are intended to ensure an ongoing reduction in PM_{10} . From 2007 until 2011, a small decrease is anticipated annually as low or no emission heating options gradually

replace existing solid fuel heating, and applications for resource consents are considered with respect to compliance with the curved line paths. More stringent requirements for domestic heating (which contributes approximately 99% of PM_{10} in Air Zone 1) are required to ensure compliance with the ambient air quality standard for PM_{10} by 2013. A timeframe of 1 January 2012 is proposed to allow time for those in Air Zone 1 to replace any domestic heating appliances that have high PM_{10} emissions, and thereby assist in meeting the ambient air quality standard for PM_{10} by 2013. The rapid decrease in PM_{10} shown by the curved line path is expected to occur between 2011 and the 1 September 2013 deadline, as stricter requirements for discharges from domestic heating appliances come into force.

Date	Maximum Level of Particulate Matter (μg/m ³)								
	2007: From 14 April	2008	2009	2010	2011	2012	2013: By 1 September		
Alexandra	131	129	127	125	123	68	50		
Arrowtown	130	128	126	124	122	68	50		
Clyde	66	66	65	65	64	54	50		
Cromwell	97	96	95	93	93	60	50		

Table 2: Data representing curved line paths towards achievement of the ambient air quality standard for PM_{10} in Air Zone 1

A straight line path to compliance is intended to achieve the reduction of PM_{10} in specified airsheds in Air Zone 2. The start-point values given in Table 3 apply from 8 December 2005 and reflect the typical level of wintertime PM_{10} observed in that airshed. Rules 16.3.1.1 and 16.3.1.3 to 16.3.1.6 will ensure a steady reduction in PM_{10} discharges as low or no emission heating options gradually replace existing solid fuel heating, and applications for resource consents are considered with respect to compliance with the straight line path.

Table 3: Start points for the straight line path to achievement of the ambient air quality standard for PM_{10} in Air Zone 2

	Maximum Level of Particulate Matter (µg/m³)							
Date	2007: From 14 April	2008	2009	2010	2011	2012	2013: By 1 September	
Balclutha	54	53	53	52	51	51	50	
Central Dunedin	57	56	55	54	52	51	50	
Green Island	76	72	67	63	59	54	50	
Milton	57	56	55	54	52	51	50	
Mosgiel	83	78	72	67	61	56	50	
North Dunedin	51	51	51	51	50	50	50	
Oamaru	86	80	74	68	62	56	50	
Palmerston	86	80	74	68	62	56	50	
Port Chalmers	51	51	51	51	50	50	50	

	Maximum Level of Particulate Matter (μg/m³)							
Date	2007: From 14 April	2008	2009	2010	2011	2012	2013: By 1 September	
South Dunedin	69	66	63	60	56	53	50	
Waikouaiti	54	53	53	52	51	51	50	

Note: Only those airsheds within Air Zone 2 where monitoring has shown breaches of the ambient air quality standard for PM_{10} have straight line paths to achieve compliance with the NESAQ.

The other Air Zone 2 airsheds and the Air Zone 3 airshed do not breach the ambient air quality standards for PM_{10} and therefore do not require a path to compliance.

Principal reasons for adopting

This policy is adopted to meet the requirement of regulation 17B(2)(a) of the NESAQ, for a plan to contain curved line paths that show how the ambient air quality standard for PM₁₀ for airsheds in Air Zone 1 will be achieved by 1 September 2013. The curved path better recognises the likely path to compliance for these airsheds.

The policy also specifies the start points of the straight line paths for the specified airsheds within Air Zone 2 that currently breach the ambient air quality standard for PM_{10} , and identifies in one regulatory document the straight line path information needed to interpret the NESAQ and rules within this Plan for considering resource consent applications under section 16.3.15. For these airsheds, the straight line paths reflect the most likely path to compliance with the ambient air quality standard for PM_{10} by 1 September 2013.

Other policies 8.1.1, 8.1.2, 8.2.6, 8.2.7, 9.1.2, 9.1.3, 9.1.4 Rules 16.3.1.1 to 16.3.1.7, 16.3.2.1 to 16.3.2.6, 16.3.4.1 to 16.3.4.3, 16.3.5.1 to 16.3.5.3, 16.3.5.9, 16.3.14.1, 16.3.15.1 to 16.3.15.5 Methods 17.2.4.1, 17.3.1.1, 17.4.1.1, 17.4.2.1 to 17.4.2.3, 17.5.1.1, 17.5.1.2, 17.5.2.1, 17.5.2.2, 17.5.3.2, 17.5.3.2, 17.5.4.1

- 9.1.2 To enable the use of offsets for the discharge of PM₁₀ when:
 - (a) All practicable mechanisms to prevent or minimise the effects of the discharge have been applied;
 - (b) The offset is applied to a source in the airshed where the direct effects of the discharge may be experienced;
 - (c) The offset takes effect within 1 year after the grant of the resource consent; and
 - (d) The offset is effective for the duration of the consent.

Explanation

Regulation 17C of the NESAQ enables the use of offsets in particular circumstances and is silent on their use in other circumstances.

The Otago Regional Council expects an applicant to demonstrate that measures have been taken to prevent or minimise the effects of any PM_{10} discharge. Where these measures are demonstrated, and a risk remains of a breach of the applicable curved or straight line path, or ambient air quality standard, then at Council's discretion, consideration will be given to the use of offsets within the airshed.

The size of the offset will be determined by:

- The characteristics of the PM₁₀ discharge, including how much PM₁₀ is to be discharged, and the frequency, rate and manner of the discharge;
- To what extent the proposed PM₁₀ discharge contributes to the airshed breaching, or being at risk of breaching, the applicable curved or straight line path or the ambient air quality standard; and
- Whether a full or partial reduction in PM_{10} is required under Rule 16.3.15.3.

In exercising its discretion to enable the use of offsets, Council will give consideration to the interaction between the offset discharge and the proposed discharge.

Any reduction in discharges of PM_{10} by offsetting must take effect within one year after the grant of the resource consent and be effective for the duration of the consent.

Principal reasons for adopting

This policy recognises that the discharge of PM_{10} from certain activities is possible, even when the best technology is applied. It enables industries to operate using appropriate technology, and allows mitigation of the adverse effects of any residual PM_{10} discharges in the vicinity by providing for the use of offsets in certain circumstances

Other policies 8.1.1, 8.1.2, 8.2.2 to 8.2.5, 9.1.1 Rules 16.3.1.7, 16.3.2.6, 16.3.3.2, 16.3.4.3, 16.3.5.9, 16.3.6.3, 16.3.7.3, 16.3.11.2, 16.3.14.1, 16.3.15.2 to 16.3.15.5

- 9.1.3 To reduce PM₁₀ emissions from domestic heating appliances:
 - (a) In Air Zone 3, by requiring all new domestic heating appliances on properties less than 2 hectares in size to meet stringent emission and thermal efficiency standards, with an exemption for cookers; and
 - (b) In Air Zones 1 and 2, by requiring all new domestic heating appliances to meet stringent emission and thermal efficiency standards, with the most stringent standards to be applied to Air Zone 1; and
 - (c) In Air Zone 1, by prohibiting domestic heating appliances that do not meet stringent emission and thermal efficiency standards; and
 - (d) By providing in some circumstances for discharges from domestic heating appliances that do not meet the permitted emission and thermal efficiency standards in recognised heritage buildings and commercial

premises, having regard to cultural, heritage, amenity and economic values; and

(e) By encouraging and promoting the use of clean heating technologies.

Explanation

The main source of PM_{10} emissions in all areas during winter is derived from the heating of buildings, especially dwellings, using solid fuel. Human health and safety needs to be assured through allowing for adequate heating of buildings without causing unacceptable levels of PM_{10} contamination in ambient air.

Requiring the installation of domestic heating appliances with low PM_{10} emissions will be the principal means for ensuring PM_{10} concentrations in ambient air follow the curved line paths depicted in Policy 9.1.1, and ensuring that ambient air quality in the rest of Otago remains high. Promotion of clean heat technologies, such as well designed domestic heating appliances and fuels that are subject to strict quality control, will further reduce PM_{10} discharges.

On properties less than 2 hectares in size, the NESAQ requires an emission standard for woodburners of less than 1.5 g/kg of dry wood burnt, and a thermal efficiency standard of not less than 65%. This standard applies unless rules in this Plan are more stringent.

Both the emission and thermal efficiency standards set by the NESAQ for woodburners are applied to all other domestic heating appliances located on properties of less than 2 hectares in size, unless they are located in Air Zone 1 or 2, as this ensures all domestic heating appliances are treated equitably. In Air Zones 1 and 2 the emission and thermal efficiency standards apply regardless of property size. In Air Zone 3 an exemption is made for cookers as ambient air quality is generally high and it is important for these smaller communities or isolated smaller properties to be able to be self-sufficient for heating and cooking as it may take longer to restore power in the event of a break to supply.

A more stringent emission standard applies in Air Zone 1 due to the high number of excedences of the PM_{10} ambient air quality standard. All newly installed domestic heating appliances must meet this standard, and those existing domestic heating appliances that do not have sufficiently low PM_{10} emissions will be prohibited from 2012.

Within Air Zones 1 and 2, and for under 2 hectare properties in Air Zone 3, discharges from any other domestic heating appliance installed from 14 April 2007, including any appliance that is untested or has been modified after being tested, must generally be prohibited if the Plan's objectives are to be met. The cumulative environmental effects in an air zone are likely to be significant if such appliances were able to be operated under authorisation such as a resource consent.

Provision in some circumstances is made for domestic heating appliances in recognised heritage buildings and in commercial premises to avoid adverse localised effects of this prohibition on cultural, heritage, amenity and economic values.

Principal reasons for adopting

This policy is adopted to assist in ensuring public health impacts of PM_{10} air pollution are minimised while allowing for the heating of buildings by means that are environmentally acceptable and sustainable. A stringent approach, including prohibition, is necessary for domestic heating which is the largest source of PM_{10} emissions in built up areas. Recognised heritage buildings must be provided for in certain circumstances, and allowing applications for resource consent for discharges from commercial premises provides an opportunity for businesses to consider a range of alternatives and make choices taking into account cultural, heritage, amenity and economic factors, while allowing the Council to consider environmental effects and offsets where required.

Other policies 8.1.1, 8.1.2, 8.2.6, 8.2.7, 9.1.1, 9.1.2, 9.1.4 Rules 16.3.1.1 to 16.3.1.7, 16.3.15.1 to 16.3.15.5 Methods 17.2.4.1, 17.3.1.1, 17.4.1.1, 17.4.2.1 to 17.4.2.3, 17.5.1.1, 17.5.1.2, 17.5.2.1, 17.5.2.2, 17.5.3.2, 17.5.4.1

9.1.4 To promote clean heating in new residential areas where discharges are likely to have an adverse impact on air quality in Air Zones 1 or 2, or degrade high quality ambient air.

Explanation

Discharges from domestic heating appliances in new residential areas may adversely affect air quality. When such developments are situated near airsheds in Air Zones 1 or 2, the discharge could adversely affect air quality in those airsheds, causing a breach of the ambient air quality standard for PM_{10} . Discharges from such developments could also adversely affect air quality in their own local area. There is a greater likelihood of this occurring in areas with similar geographical and climatic conditions to those airsheds in Air Zone 1.

While this Plan sets region-wide policies and rules for controlling the discharge of contaminants into air from domestic heating appliances, city and district councils are encouraged to set more stringent standards, using land use planning mechanisms, to further improve ambient air quality. Such mechanisms include conditions on consents and covenants specifying what types of domestic heating appliances can be installed in a building. The Otago Regional Council will encourage and support the use of such mechanisms through written submissions on plan changes and consents, commenting on urban growth strategies and liaising with developers.

Principal Reasons for Adopting

This policy is adopted to assist in ensuring public health impacts of PM_{10} from domestic heating appliances are minimised and areas of high ambient air quality are not degraded by promoting clean heating in areas of new residential development, and to meet the NESPM₁₀.

Other policies 8.2.7 Methods 17.2.1.1, 17.2.1.2, 17.2.4.1, 17.3.1.1, 17.4.1.1, 17.4.2.1 to 17.4.2.3, 17.5.1.1, 17.5.1.2, 17.5.2.2, 17.5.3.2, 17.5.4.1

10.1 Policy for dust from area sources

- 10.1.1 The Otago Regional Council will encourage:
 - (a) People undertaking land use activities to adopt management practices to avoid, remedy or mitigate any adverse effects of dust beyond the boundary of the property; and
 - (b) City and district councils to use land use planning mechanisms and other land management techniques to manage land use activities which have the potential to result in dust beyond the boundary of the property.

Explanation

Part (a) of this policy indicates the Otago Regional Council's view that people undertaking land use activities need to adopt management practices to avoid, remedy or mitigate any adverse effects beyond the property where these activities are being undertaken.

Part (b) indicates the Otago Regional Council's view that city and district councils should consider using land use planning and other mechanisms as a means of managing land use activities which have the potential to generate dust.

Principal reasons for adopting

This policy recognises that there are management practices available to people undertaking land use activities, and land use mechanisms available to city and district councils, which can assist in avoiding, remedying or mitigating the adverse effects of dust.

Methods 17.2.1.1, 17.2.1.2, 17.2.3.1, 17.4.1.1, 17.5.1.1, 17.5.1.2, 17.5.2.1, 17.5.2.2

11.1 Policy for odour

- 11.1.1 To avoid or mitigate any adverse effects on human health or amenity values resulting from the discharge of offensive or objectionable odour through the use of:
 - (a) Good management practices (including the use of codes of practice) and process technology that has an inherently low odour potential to ensure the amount of odorous contaminants generated by a process or activity is minimised;
 - (b) Appropriate control technologies to reduce the emission of odorous contaminants;
 - (c) Site planning mechanisms and other land use management techniques to reduce the potential for adverse off site effects; and
 - (d) Tools and techniques that provide an objective assessment of odour, such as olfactometry, odour dose response assessments and community surveys.

Explanation

It is not always practicable or necessary to avoid all detectable odours, some of which may have only minor adverse effects. As a consequence, this policy concentrates on avoiding or mitigating the discharge of odours which are "offensive" or "objectionable" and which have the greatest potential to result in adverse effects on the well being and health of people.

The general methods identified represent the best practicable options for managing the adverse effects of odours. Options (a) and (b) are the responsibility of producers, whereas (c) refers to site management and land use planning techniques that can be implemented by both producers and city and district councils to reduce the potential for adverse off-site effects to occur. Part (c) indicates the Otago Regional Council's view that odour producers should, wherever practicable, take measures to ensure no adverse effects of their activities occur beyond their property boundaries. Measures may include buffer zones, or controls on the location of the odour-generating components of the development. City and district councils can also manage land uses so as to avoid the potential for incompatible land uses to develop within an area. The techniques listed in (d) indicate the type of measurement tools that the Regional Council considers appropriate in order to undertake odour assessments. The preferred technique will vary on a case-by-case basis and will be established in conjunction with producers.

The odour control technologies referred to in (b) include, but are not limited to, technologies which involve gas collection and control, absorption, biofiltration, incineration and odour modification. Some examples of these technologies are set out in Appendix 3 of the Ministry for the Environment publication *Odour Management Under the Resource Management Act* (1995).

Principal reasons for adopting

This policy provides guidance on how to avoid or mitigate offensive or objectionable odours. It also recognises the subjective nature of odour problems and the need for any management approach to be flexible while at the same time focusing on those discharges which are most likely to be "offensive" or "objectionable".

Rules 16.3.1.2 to 16.3.1.7, 16.3.2.1 to 16.3.2.6, 16.3.3.2, 16.3.4.1 to 16.3.4.3, 16.3.5.1 to 16.3.5.9, 16.3.7.1 to 16.3.7.3, 16.3.8.1, 16.3.8.2, 16.3.11.1, 16.3.11.2, 16.3.13.1, 16.3.13.2, 16.3.14.1 Methods 17.2.1.1, 17.2.1.2, 17.5.1.1, 17.5.2.1, 17.5.2.2

12.1 Policy for agrichemical spray drift

12.1.1 The Otago Regional Council will:

- (a) Require the applicators of agrichemicals to undertake spraying in a manner that avoids:
 - (i) Spray drift beyond the target area or boundary of the property being sprayed; and
 - (ii) Adverse effects on human health and safety, ecosystems, sensitive areas or places, amenity values and other non-target areas or species; and
- (b) Encourage city and district councils to use land use planning mechanisms and other land management techniques to mitigate adverse effects from agrichemical spray drift.

PART III AIR QUALITY MANAGEMENT

Explanation

Part (a) of this policy indicates the Otago Regional Council's view that it is not adequate to merely remedy or mitigate adverse effects, and that people applying agrichemicals need to take proactive measures to avoid adverse effects beyond the target area or boundary of the property.

In order to address the effects of spray drift, it is important that best practice is adopted to avoid the drift itself occurring. Best practice is the subject of the *Code of Practice for the Management of Agrichemicals* (NZS 8409:1999), developed by the New Zealand Agrichemical Education Trust. The code details management practices that can be adopted to lower the risk of drift hazard and reduce the potential for adverse effects to occur. Schedule 4 of this Plan contains a summary of these practices, and the Otago Regional Council encourages those applying agrichemicals to follow them, to ensure that users are adopting best practice.

The avoidance of adverse effects shall be achieved primarily through adopting management practices which produce the lowest risk of drift hazard. Should this fail to avoid spray drift, this policy provides guidance to people applying agrichemicals about the range of values, areas or places which are sensitive to the effects of agrichemical sprays and which people applying agrichemicals should avoid affecting.

Sensitive areas or places shall include but not be limited to:

- Residential dwellings and associated private property;
- Educational facilities;
- Places of public assembly;
- Public amenity areas including parks, reserves, gardens, sports grounds, beaches, and thoroughfares;
- Public roads;
- Domestic or community water supply catchments and intakes;
- Water bodies and wetlands, and associated riparian vegetation;
- Areas of significant indigenous vegetation and significant habitats of indigenous fauna; and
- Commercially important or sensitive plants, crops or farming systems (eg, organic farms).

This list however is not exhaustive, as sensitive areas may change over time and there may be other areas, places or features that are particularly sensitive to the effects of agrichemical spray at the local level.

Part (b) indicates the Otago Regional Council view that city and district councils should consider using land use planning and other mechanisms as a means of addressing the adverse effects of agrichemical drift following discharge into air. Some options are outlined in Method 17.2.1.2.

Principal reasons for adopting

This policy recognises that where the use of agrichemicals is necessary, applying good management practices will reduce the risk of spray drift and the potential for adverse effects to occur. It also recognises that there are mechanisms available to city and district councils which can assist in mitigating the adverse effects of spray drift and achieving integrated management.

Rules 16.3.9.1 to 16.3.9.4, 16.3.14.1 Methods 17.2.1.1, 17.2.1.2, 17.3.1.1,17.5.1.1, 17.5.1.2, 17.5.2.1, 17.5.2.2

13.1 Policy for the burning of vegetative matter on production land

13.1.1 To encourage people undertaking vegetation burning to adopt good management practices, including those set out in Schedule 5 to avoid or mitigate adverse effects.

Explanation

The discharge of smoke from the burning of vegetation can cause adverse effects including nuisance, amenity and visibility effects. In order to avoid or mitigate these effects, it is important that good burning practices are adopted. These practices are outlined in Schedule 5. Where good practices do not mitigate such effects, the Council will encourage the use of alternative means to dispose of or clear unwanted vegetation.

Principal reasons for adopting

Policy 7.5.2 of the Regional Policy Statement for Otago requires adverse effects on human health, the environment, visual impacts and odour to be avoided, remedied or mitigated. This policy aims to avoid or mitigate the adverse effects of discharges into air from the burning of vegetation on production land.

Method 17.2.1.1, 17.2.1.2, 17.5.1.2

14.1 Policy for motor vehicle emissions

14.1.1 The Otago Regional Council will:

- (a) Advocate and support the development of a nationally co-ordinated programme for the management of motor vehicle emissions which:
 - (i) Develops national vehicle emission testing standards;
 - (ii) Encourages the use of transport fuels which minimise the emissions of contaminants into air;
 - (iii) Promotes the use of improved vehicle technology to reduce emissions; and
 - (iv) Promotes the use of fuel efficient and well maintained vehicles;
- (b) Encourage city and district councils to use land use planning and traffic management mechanisms to avoid the occurrence of localised air quality problems associated with emissions from motor vehicles;
- (c) Include appropriate provisions in the Regional Land Transport Strategy and Passenger Transport Plan for Otago aimed at avoiding, remedying or mitigating the adverse environmental effects of the discharge of contaminants into air from motor vehicles; and
- (d) Promote understanding of the effects of motor vehicle emissions on the region's air resource.

Explanation

Part (a) of the policy indicates the Otago Regional Council's view that while emissions from motor vehicle sources have the potential to be a major source of pollution in Otago, a national approach towards controlling vehicle emissions using technical means is essential in order to achieve an equitable and coordinated approach throughout all regions in New Zealand. In this context "technical means" refer principally to on-vehicle techniques, which aim to improve the emission performance capability of motor vehicles.

At the time of writing this Plan, several national initiatives were being developed. These include:

- A Vehicle Fleet Emissions Control Strategy being developed by the Ministry of Transport. This strategy will recommend the most cost-effective means of managing vehicle emissions;
- A Ministry for the Environment co-ordinated programme to develop a core set of national environmental indicators for transport; and
- Monitoring of the effects of transport emissions by the Ministry of Transport and Ministry for the Environment.

Part (b) recognises that local land use planning and traffic management mechanisms can reduce the concentration of emission producing activity and therefore avoid localised problems. Such mechanisms can supplement and complement the technical means referred to in part (a) of the policy.

Parts (c) and (d) recognise that the Regional Council can assist in co-ordinating local initiatives and monitoring, researching, and disseminating information about air quality.

Principal reasons for adopting

This policy is adopted to reflect the Regional Council's view that it is more appropriate for any regulations or minimum standards for discharges into air from mobile sources, such as motor vehicles, to be developed at a national level. It also recognises that local initiatives can supplement and complement national initiatives.

This approach is consistent with the Regional Policy Statement for Otago, in particular, the policies in the Air and Built Environment chapters.

Methods 17.2.1.1, 17.2.1.2, 17.2.3.1, 17.3.1.1, 17.5.3.1

15.1 Policy for global issues

15.1.1 To support and promote, as appropriate, central government initiatives to control and minimise emissions of greenhouse gases and ozone layer depleting substances.

Explanation

Central government has the primary role for developing and implementing policy for discharges to air of global significance. The Regional Council will support and implement national policies or agreements when such initiatives are appropriate in the Otago context.

Principal reasons for adopting

This policy indicates the Regional Council's commitment to support national initiatives to control and minimise emissions of greenhouse gases and ozone depleting substances.

Method 17.5.3.1

PART III AIR QUALITY MANAGEMENT

Part IV Rules



16.1 Introduction

This part of the Plan contains the following:

- A users' guide, which provides an overview of the rules and some explanatory notes for interpreting the rules;
- The Plan rules, which determine whether a particular activity is allowed without the need for consent, is regulated, or is prohibited in the Otago region;
- The principal reasons for adopting the rules; and
- A description of the type of information that will be required with any resource consent application.

16.2 Users' guide

16.2.1 Introduction

This users' guide provides:

- A discussion of the relationship between this Plan and the Regional Plan: Waste for Otago, and between this Plan and the Resource Management (Marine Pollution) Regulations;
- A brief explanation of how to determine a particular activity's status, that is, whether it is allowed, regulated, or prohibited under this Plan;
- A summary index of the rules;
- Explanations of the circumstances in which the Council may consider processing applications for block discharge permits and regional discharge permits for activities involving discharges of contaminants into air; and
- A guide to interpreting the terms "noxious", "dangerous", "offensive" and "objectionable", which are used in many of the rules.

16.2.2 Relationship to Regional Plan: Waste

The rules contained within this Plan **do not apply** to the discharge of contaminants into air associated with the following, which are controlled by the Regional Plan: Waste for Otago:

- Contaminated sites;
- Facilities for the treatment or disposal of hazardous wastes;
- New or operating landfills;
- Closed landfills;
- Offal pits on production land, intensive farms and industrial and trade premises;
- Farm landfills;
- Composting and silage production; and
- Greenwaste landfills.

16.2.3 Relationship to Resource Management (Marine Pollution) Regulations 1998

Except in the case of rules for the three activities listed in the next paragraph, the rules contained within this Plan **do not apply** to the discharge of contaminants into air associated with the normal operation of ships within Otago's coastal marine area.

Normal operation activities are specified in Schedule 4 of the Resource Management (Marine Pollution) Regulations 1998. This Plan contains rules applying to discharges from the following three activities:

- Abrasive blasting;
- Spray painting; and
- Conveying of bulk materials.

16.2.4 Relationship to NESAQ

The NESAQ was gazetted in September 2004, and its requirements override any less stringent requirements in regional plans. Plan Change 2 to this Plan resulted in changes to matters relating to the discharge of PM_{10} , making it consistent with the NESAQ requirements for PM_{10} . The NESAQ also sets standards for levels of carbon monoxide, nitrogen dioxide, sulphur dioxide and ozone, to protect ambient air quality. These matters must be taken into consideration for any application for resource consent to discharge contaminants to air.

The NESAQ prohibits the discharge of dioxins and other toxics to air, and contains requirements with regard to landfill gas. Landfill discharges are currently addressed by the Regional Plan: Waste for Otago, therefore the NESAQ requirements for landfill discharges do not affect the implementation of this Plan.

16.2.5 Key steps to determine status

The key steps to determine whether a discharge of contaminants into air is allowed without the need for consent or requires a resource consent are as follows:

- Decide whether or not the activity is to be undertaken on an industrial or trade premises. The definition of "Industrial or trade premises" is contained in the Glossary;
- Determine how the activity is affected by the rules. It is important to consider all activities that may be undertaken on a site. For example, discharges associated with both agrichemical application and intensive farming may occur on one site;
- If an activity is controlled by a particular rule but cannot meet **the conditions of that rule** it will be a **discretionary activity** under Rule 16.3.14.1 unless otherwise stated;
- If the activity is at an industrial or trade premises and is not considered by the rules in sections 16.3.3 to 16.3.13, it will be a discretionary activity under Rule 16.3.14.1; and
- If the activity is not at an industrial or trade premises, and is **not covered** by rules in sections 16.3.1 to 16.3.13, it can be undertaken without a resource consent.

16.2.6 Summary index to rules

Each of the rules in the Plan specifies whether a particular activity is **permitted**, **discretionary** or **prohibited**.

If an activity is:

- Permitted, no resource consent is required, provided all the conditions stated in the rule are met.
- Discretionary, a resource consent is required, and the Otago Regional Council has retained its discretion as to whether it will grant a consent or not. In considering any application for a discretionary activity, the Council will be guided by the policies contained in this Plan, the Regional Policy Statement for Otago, and the requirements of the Resource Management Act. Conditions may be included on any resource consent granted.
- Prohibited, the activity is expressly prohibited by this Plan. These are activities which may not occur within Otago and for which no resource consent will be granted.

The Resource Management Act definitions of these terms are contained in the Glossary.

The resource consents required by the rules in this Plan are called "discharge permits".

The activities covered by the rules in this Plan are set out in Table 5.

Please note that Table 5 is intended to provide only a summary of the rules. To determine the exact status of an activity, it is necessary to refer to the rules directly.

In some circumstances, consents or approvals may also need to be obtained from a city or district council. The relevant authority should be consulted.

The NESAQ sets standards for levels of carbon monoxide, nitrogen dioxide, PM_{10} , sulphur dioxide and ozone discharged to air, to protect ambient air quality. The NESAQ must be taken into consideration for any application for resource consent to discharge these contaminants to air.

Table 5: Index to rules

Note: Section 16.3.15 (Discharges of PM₁₀) may also apply to any discretionary activity under Rules 16.3.1 to 16.3.14.

Rule number	Page	Activity status	Description of rule
	Discha	rges from dom	estic heating appliances (section 16.3.1)
16.3.1.1	54	Prohibited	Discharges from domestic heating appliances that do not meet permitted activity Rules 16.3.1.2 to 16.3.1.6 or discretionary activity Rule 16.3.1.7.
16.3.1.2	54	Permitted	Discharges from domestic heating appliances in Air Zone 1

PART IV RULES

Rule number	Page	Activity status	Description of rule			
16.3.1.3	55	Permitted	Discharges from domestic heating appliances in Air Zone 2			
16.3.1.4	55	Permitted	Discharges from cookers in Air Zone 3			
16.3.1.5	55	Permitted	Discharges from domestic heating appliances other than cookers in Air Zone 3			
16.3.1.6	56	Permitted	Discharges from any domestic heating appliances installed in a recognised heritage building			
16.3.1.7	56	Discretionary	Discharges from any domestic heating appliances installed in a recognised heritage building or a building on commercial premises			
	•	Outdoo	r burning (section 16.3.2)			
16.3.2.1	57	Permitted	Discharges from outdoor burning on residential properties in Air Zone 1 or 2			
16.3.2.2	57	Permitted	Discharges from outdoor burning on non-residential properties, including production land, in Air Zone 1 or 2			
16.3.2.3	58	Permitted	Discharges from outdoor burning on properties which are not production land, in Air Zone 3			
16.3.2.4	58	Permitted	Discharges from outdoor burning on production land in Air Zone 3			
16.3.2.5	58	Permitted	Discharges from outdoor burning of any campfire or celebratory fire, or for the cooking of food			
16.3.2.6	58	Discretionary	Other discharges from outdoor burning			
Burning or incineration of specified materials (section 16.3.3)						
16.3.3.1	60	Prohibited	Burning or incineration of specified materials			
16.3.3.2	60	Discretionary	Discharges from the burning or incineration of materials specified in Rule 16.3.3.1			
Pr	oducts o	f combustion fr	om fuel burning equipment (section 16.3.4)			
16.3.4.1	61	Permitted	Discharges from fuel burning equipment in Air Zone 1 or 2			
16.3.4.2	61	Permitted	Discharges from fuel burning equipment in Air Zone 3			
16.3.4.3	62	Discretionary	Other discharges of the products of combustion from fuel burning equipment			
	Discha	rges from indus	strial or trade processes (section 16.3.5)			
16.3.5.1	63	Permitted	Discharges from the processing of plant or animal matter			
16.3.5.2	63	Permitted	Discharges from sorting, crushing, screening, conveying and storage of powdered or bulk products			
16.3.5.3	64	Permitted	Discharges from mineral extraction and processing			
16.3.5.4	64	Permitted	Discharges from chemical processing, manufacturing and industrial or trade processes which discharge hazardous air contaminants			
16.3.5.5	65	Permitted	Discharges from metal processing and foundries			

Rule number	Page	Activity status	Description of rule			
16.3.5.6	65	Permitted	Discharges from surface coating (including spray painting) and di-isocyanate use			
16.3.5.7	66	Permitted	Discharges from petroleum and hydrocarbon processes			
16.3.5.8	66	Permitted	Discharges from the processing of wood and wood products			
16.3.5.9	67	Discretionary	Other discharges from industrial or trade processes			
		Abrasiv	e blasting (section 16.3.6)			
16.3.6.1	67	Permitted	Discharges from wet or vacuum abrasive blasting			
16.3.6.2	68	Permitted	Discharges from dry abrasive blasting			
16.3.6.3	68	Discretionary	Other discharges from abrasive blasting			
		Waste ma	anagement (section 16.3.7)			
16.3.7.1	69	Permitted	Discharges from the storage, transfer, treatment and disposal of liquid-borne municipal, industrial or trade waste			
16.3.7.2	69	Permitted	Discharges from farm waste management			
16.3.7.3	70	Discretionary	Other discharges from waste management			
Intensive farming (section 16.3.8)						
16.3.8.1	71	Permitted	Discharges from intensive farming			
16.3.8.2	71	Discretionary	Other discharges from intensive farming			
Agrichemical application (section 16.3.9)						
16.3.9.1	72	Permitted	Discharges from agrichemical application on residential properties			
16.3.9.2	72	Permitted	Discharges from agrichemical application on production land and industrial or trade premises			
16.3.9.3	72	Permitted	Discharges from agrichemical application in public amenity areas and places of public assembly			
16.3.9.4	73	Discretionary	Other discharges from agrichemical application			
		Water vapour,	heat and energy (section 16.3.10)			
16.3.10.1	74	Permitted	Discharges of steam or water vapour			
16.3.10.2	75	Permitted	Discharges of energy, including heat			
16.3.10.3	75	Discretionary	Other discharges of water vapour, heat or energy			
	V	entilation or vap	oour displacement (section 16.3.11)			
16.3.11.1	76	Permitted	Discharges for the purpose of ventilation or vapour displacement			
16.3.11.2	76	Discretionary	Other discharges from ventilation or vapour displacement			

Rule number	Page	Activity status	Description of rule				
Nuclea	Nuclear power generation or nuclear weapon manufacturing (section 16.3.12)						
16.3.12.1	76	Prohibited	Discharges from nuclear power generation or nuclear weapon manufacturing				
	•	General perm	itted activities (section 16.3.13)				
16.3.13.1	77	Permitted	Discharges from miscellaneous activities				
16.3.13.2	77	Permitted	Discharges from burning by NZ Fire Service				
	Other discretionary activities (section 16.3.14)						
16.3.14.1	78	Discretionary	Discretionary activities (general rule); applies when a discharge from an industrial or trade premises is not provided for in the rules in sections 16.3.1 to 16.3.13				
	Discharges of PM ₁₀ (section 16.3.15)						
16.3.15.1	78	Prohibited	Discharges of PM_{10} , which are not permitted activities or authorised by a resource consent				
16.3.15.2	79	Discretionary	Discharges of PM_{10} , before 1 September 2013, where the ambient air quality standard for PM_{10} is already breached				
16.3.15.3	79	Discretionary	Discharges of PM_{10} , before 1 September 2013, if a consent cannot be granted under Rule 16.3.15.2, and the applicant proposes offsets				
16.3.15.4	80	Discretionary	Discharges of PM_{10} before 1 September 2013, where the ambient air quality standard for PM_{10} is not breached				
16.3.15.5	80	Discretionary	Discharges of PM_{10} after 31 August 2013, which are not permitted activities				

16.2.7 Block discharge permits

In some cases, such as large factories, there may be a multitude of discharge points from different types of activities and sources. In these instances the Otago Regional Council will consider processing one consent for the entire site, provided that all the actual or potential discharge points are clearly identified in the resource consent application. In such circumstances different consent conditions may be applied to different activities on the site.

16.2.8 Regional discharge permits

Some activities which result in discharges into air are mobile in nature (e.g., abrasive blasting and asphalt manufacture). For these types of activities, the Council will consider granting regional discharge permits which will allow an operator to work throughout the region, subject to appropriate conditions.

This approach is intended to give those resource users undertaking processes which are mobile and which discharge contaminants into air, the freedom to conduct their activities relatively unhindered, but within the bounds of their discharge permit.

16.2.9 Noxious, dangerous, offensive and objectionable effects

Several rules in this Plan use the terms "noxious", "dangerous", "offensive", and "objectionable". These terms are also included in Section 17 of the Resource Management Act 1991. They are not, however, defined in the Act and this means that they bear their natural and ordinary meaning as applied by common English usage.

The terms are not defined in the Glossary to this Plan because of the need to take account of case law precedent as it develops, i.e., the Plan cannot override interpretations decided by the judiciary.

The following notes are intended to provide some guidance for interpreting these terms. It should be noted however, that they are not objective measures and that what may be considered noxious, dangerous, offensive or objectionable will depend on the circumstances relevant to each case.

Noxious, dangerous - The Concise Oxford Dictionary defines "*noxious*" as "*harmful, unwholesome*". At the time of writing this Plan, the term "*noxious*" did not appear to have been defined or considered in case law relating to the Resource Management Act 1991. Noxious effects may include significant adverse effects on the environment (e.g., on plant and animal life) although the effects may not be dangerous to humans.

"Dangerous" is defined as "involving or causing danger". Dangerous discharges include those that are likely to cause adverse physical health effects, such as discharges containing toxic levels of chemicals.

Offensive, objectionable - "Offensive" is defined as "... giving or meant or likely to give offensive... disgusting, foul smelling, nauseous, repulsive...". "Objectionable" is defined as "open to objection, unpleasant, offensive". Case law has established that what may be offensive or objectionable under the Resource Management Act 1991 cannot be defined or prescribed except in the most general of terms. Each case will depend upon its own circumstances. Key considerations include:

- (i) The location of an activity and sensitivity of the receiving environment -What may be considered offensive or objectionable in an urban area, may not necessarily be considered offensive or objectionable in a rural area. The converse may also be true;
- (ii) **Reasonableness** Whether or not an activity is likely to be considered offensive or objectionable by an ordinary person who is representative of the community and neither hypersensitive nor insensitive, deciding whether the activity is disgusting, nauseous, repulsive or otherwise objectionable. Representative community surveys can also be used in some instances; and
- (iii) Existing uses It is important to consider what lawfully established activities exist in an area, that is, if a new activity requires a permit, the cumulative effects of both the existing and proposed discharges of contaminants into air should be considered.

Within Otago the effects of offensive or objectionable odours are often commented on by individuals. While each investigation of a complaint concerning offensive or objectionable odour will depend on the specific circumstances of the discharge, the general investigative approach adopted by Council officers will be to take into account the FIDOL factors.

The FIDOL factors were identified by the Ministry for the Environment in a report entitled "*Odour Management Under the Resource Management Act*" (1995) as the main factors which influence the significance of adverse odour effects. The FIDOL factors are:

- **Frequency** of the odour occurrence;
- **Intensity** of the odour;
- **Duration** of exposure to the odour;
- Offensiveness of the odour; and
- Location of the discharge.

Such assessments are likely to be based initially on observations made by Council officers. Information may also be gained from the discharger, independent consultants, other observers and people living or working in the area. Techniques such as odour diaries and community surveys, olfactometry measurements, or electronic measuring devices, may also be used.

16.3 Rules

Note: The rules in this Plan apply to any discharge from any industrial or trade premises which would not have required any licence or authorisation under the Clean Air Act 1972 and which has, under Section 418 of the Resource Management Act 1991, been exempt from Section 15(1)(c) of the Act.

16.3.1 Discharges from domestic heating appliances

Note:

- 1. Discharges from all domestic heating appliances (which include open fires) in buildings (domestic and non-domestic), are permitted providing they meet the relevant requirements of Rules 16.3.1.2 to 16.3.1.6. If the discharge is from a recognised heritage building or from a building on commercial premises, and that discharge does not meet permitted activity conditions, consent may be applied for under Rule 16.3.1.7. If these rules are unable to be met, the discharge is prohibited by Rule 16.3.1.1.
- 2. Unless a rule sets more stringent requirements, all woodburners in buildings on properties less than 2 hectares in size must also meet the requirements set by the NESAQ, which is incorporated in Rules 16.3.1.2 to 16.3.1.6.
- 3. Domestic heating appliances (excluding open fires) installed in the former Schedule 1.2 areas between 28 February 1998 and 14 April 2007 were required to meet a particulate emission rate of 4 g/kg or less, or have a resource consent.
- 4. Discharges from domestic heating appliances that are outside of buildings are addressed by the Rules under section 16.3.2 of this Plan (Outdoor Burning).
- 5. Particulate emission rates and thermal efficiency are to be determined using the testing procedures described in Method 17.5.4.1. The rates are achievable by many domestic heating appliances currently on the market.

16.3.1.1 Discharges from domestic heating appliances – prohibited activity

Except as provided for by Rules 16.3.1.2 to 16.3.1.7, the discharge of contaminants into air from any domestic heating appliance in any building is a *prohibited activity*, for which no consent will be granted.

16.3.1.2 Discharges from domestic heating appliances in Air Zone 1 – permitted activity

The discharge of contaminants into air from any domestic heating appliance in any building in Air Zone 1:

(1) If the domestic heating appliance was lawfully installed and meets a particulate emission rate of 0.7 g/kg or less of fuel burnt and has a thermal efficiency of not less than 65%;

is a *permitted activity*, providing:

- (a) Any discharge of smoke, odour, particulate matter or gas is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property; or
- (2) If the domestic heating appliance was lawfully installed before 14 April 2007 in Air Zone 1 (Alexandra, Arrowtown or Cromwell), or was lawfully installed before 1 April 2009 in Air Zone 1 (Clyde), and meets a particulate emission rate of less than 1.5 g/kg of fuel burnt;

is a *permitted activity*, providing:

- (a) Any woodburner installed after 1 September 2005 in a building on a property with an allotment size of less than 2 hectares also has a thermal efficiency of not less than 65%; or
- (b) Any discharge of smoke, odour, particulate matter or gas is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property; or
- (3) If the domestic heating appliance was lawfully installed before 14 April 2007 in Air Zone 1, and has a particulate emission rate of greater than or equal to 1.5 g/kg of fuel burnt;
- is a *permitted activity*, until 1 January 2012, providing:
 - (a) Any woodburner installed after 1 September 2005 in a building on a property with an allotment size of less than 2 hectares meets a discharge of less than 1.5 g/kg of dry wood burnt and has a thermal efficiency of not less than 65%; and
 - (b) Any discharge of smoke, odour, particulate matter or gas is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.1.3 Discharges from domestic heating appliances in Air Zone 2 – permitted activity

The discharge of contaminants into air from any domestic heating appliance in any building in Air Zone 2:

(1) If the domestic heating appliance was lawfully installed after 14 April 2007 and meets a particulate emission rate of less than 1.5 g/kg of fuel burnt and has a thermal efficiency of not less than 65%;

is a *permitted activity*, providing:

- (a) Any discharge of smoke, odour, particulate matter or gas is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property; or
- (2) If the domestic heating appliance was lawfully installed before 14 April 2007;

is a *permitted activity*, providing:

- (a) Any woodburner installed after 1 September 2005 in a building on a property with an allotment size of less than 2 hectares meets a discharge of less than 1.5 g/kg of dry wood burnt and has a thermal efficiency of not less than 65%; and
- (b) Any discharge of smoke, odour, particulate matter or gas is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.1.4 Discharges from cookers in Air Zone 3 – permitted activity

The discharge of contaminants into air from any cooker, lawfully installed in any building in Air Zone 3;

is a *permitted activity*, providing:

(a) Any discharge of smoke, odour, particulate matter or gas is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.1.5 Discharges from domestic heating appliances other than cookers in Air Zone 3 – permitted activity

Except as provided for by Rule 16.3.1.4, the discharge of contaminants into air from any domestic heating appliance in any building in Air Zone 3;

is a *permitted activity*, providing:

(a) Any woodburner installed after 1 September 2005, or any other domestic heating appliance installed after 14 April 2007, in a building on a property with an allotment size of less than 2 hectares, meets a discharge of less than 1.5 g/kg of fuel burnt and has a thermal efficiency of not less than 65%; and

(b) Any discharge of smoke, odour, particulate matter or gas is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.1.6 Discharges from any domestic heating appliance installed in a recognised heritage building – permitted activity

The discharge of contaminants into air from any domestic heating appliance lawfully installed in any recognised heritage building:

is a *permitted activity*, providing:

- (a) The domestic heating appliance contributes to the significance of the recognised heritage building; and
- (b) Any woodburner installed after 1 September 2005 in a building on a property with an allotment size of less than 2 hectares meets a particulate emission rate of less than 1.5 g/kg of dry wood burnt and has a thermal efficiency of not less than 65%; and
- (c) Any discharge of smoke, odour or particulate matter is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.1.7 Discharges from any domestic heating appliance installed in a recognised heritage building or a building on commercial premises – discretionary activity

Except as provided for by Rules 16.3.1.2 to 16.3.1.6, the discharge of contaminants into air from any domestic heating appliance lawfully installed in:

- (1) Any recognised heritage building; or
- (2) Any building on commercial premises;

is a *discretionary activity*, providing:

(a) Any woodburner installed after 1 September 2005 in a building on a property with an allotment size of less than 2 hectares meets a particulate emission rate of less than 1.5 g/kg of dry wood burnt and has a thermal efficiency of not less than 65%.

Principal reasons for adopting:

Rules 16.3.1.1 to 16.3.1.7 are adopted to allow solid fuel heating of buildings, while managing the adverse effects that discharges from domestic heating appliances can have on ambient air quality, and subsequently on human health, in particular areas. The most stringent rules apply in Air Zone 1, where the most breaches of the ambient air quality standard occur. The least stringent rules apply to properties over 2 hectares in size in Air Zone 3, where ambient air quality is generally very good. Some flexibility is provided in this domestic heating rule framework to provide for cookers in Air Zone 3, recognised heritage buildings and commercial premises.

The boundary effects condition in all of the above permitted activity rules is adopted to ensure that discharges do not have significant adverse effects on the surrounding local environment and, in particular, on neighbours.

16.3.2 Outdoor burning

Note:

- 1. City and district councils in implementing the Forests and Rural Fires Act 1977 may also have bylaws controlling outdoor burning of materials for fire safety purposes.
- 2. The Health Act 1956 also has some control on nuisance effects from the discharge of contaminants into air.
- 3. These rules do not permit the use of fire accelerants such as waste petroleum products or tyres. The burning of such materials is prohibited by Rule 16.3.3.1 and incineration of such materials may be allowed only if a consent is obtained.
- 4. Where the separation distances specified in the rules cannot be achieved or other conditions of the rules are not met, consents are required to be obtained from the Otago Regional Council.

16.3.2.1 Discharges from outdoor burning on residential properties in Air Zone 1 or 2 - permitted activity

Except as provided for by Rule 16.3.2.5, the discharge of contaminants into air from outdoor burning on any residential property in Air Zone 1 or 2;

is a *permitted activity*, providing:

- (1) Only paper, cardboard, vegetative matter or untreated wood is burnt; and
- (2) The material is from the property where the burning occurs; and
- (3) The material is dry at the time of burning; and
- (4) The burning does not occur within 50 metres of the closest part of the boundary of the property; and
- (5) Any discharge of smoke, odour or particulate matter is not offensive or objectionable at or beyond the boundary of the property.

16.3.2.2 Discharges from outdoor burning on non-residential properties, including production land, in Air Zone 1 or 2 - permitted activity

Except as provided for by Rule 16.3.2.5, the discharge of contaminants into air from outdoor burning on any non-residential property, including production land in Air Zone 1 or 2;

is a *permitted activity*, providing:

- (a) Only paper, cardboard, vegetative matter or untreated wood is burnt; and
- (b) The material is from the property where the burning occurs; and
- (c) The material is dry at the time of burning; and
- (d) The burning does not occur within 100 metres of any dwelling on any other property; and
- (e) Any discharge of smoke, odour or particulate matter is not offensive or objectionable at or beyond the boundary of the property.

16.3.2.3 Discharges from outdoor burning on properties which are not production land, in Air Zone 3 - permitted activity

Except as provided for by Rule 16.3.2.5, the discharge of contaminants into air from outdoor burning on any property which is not production land, in Air Zone 3;

is a *permitted activity*, providing:

- (a) Only paper, cardboard, vegetative matter or untreated wood is burnt; and
- (b) The material is from the property where the burning occurs; and
- (c) The material is dry at the time of burning; and
- (d) Any discharge of smoke, odour or particulate matter is not offensive or objectionable at or beyond the boundary of the property.

16.3.2.4 Discharges from outdoor burning on production land in Air Zone 3 - permitted activity

Except as provided for by Rule 16.3.2.5, the discharge of contaminants into air from outdoor burning on any property which is production land, in Air Zone 3;

is a *permitted activity*, providing:

- (a) No material specified in Rule 16.3.3.1 is burnt; and
- (b) Any discharge of smoke, odour or particulate matter from burning waste is not offensive or objectionable at or beyond the boundary of the property.

16.3.2.5 Discharges from outdoor burning of any campfire or celebratory fire, or for the cooking of food - permitted activity

The discharge of contaminants into air from outdoor burning of any campfire or celebratory fire, or outdoor burning for the cooking of food by any barbecue, hangi, umu or similar means;

is a *permitted activity*, providing:

- (a) No material specified in Rule 16.3.3.1 is burnt; and
- (b) The material is dry at the time of burning; and
- (c) Any discharge of smoke, odour or particulate matter is not offensive or objectionable at or beyond the boundary of the property.

16.3.2.6 Other discharges from outdoor burning – discretionary activity

Except as provided for by Rules 16.3.2.1 to 16.3.2.5, or prohibited by Rule 16.3.3.1, the discharge of contaminants into air from outdoor burning is a *discretionary activity*.

Principal reasons for adopting

Rules 16.3.2.1 to **16.3.2.6** set up a regime that allows outdoor burning, provided that the burning is undertaken in an appropriate location, with only clean-burning material generated on the property, and does not result in adverse localised effects at or beyond the boundary of the property. The first two rules target areas with a high density of development, because proximity to a fire is a significant determinant as to whether the discharge will have adverse effects on neighbours and the environment.

Rule 16.3.2.1 applies to outdoor burning on residential properties in Air Zone 1 or 2. In order to assist in managing the effects of such burning in these areas with relatively high population densities, the rule requires burning to be undertaken more than 50 metres from the boundary of the property.

Rule 16.3.2.2 applies to outdoor burning on non-residential properties such as, but not limited to, educational facilities, industrial or trade premises, parks, reserves and production land located in Air Zone 1 or 2. The rule recognises that outdoor burning on non-residential properties in areas with relatively high population densities is likely to have more significant adverse localised effects than such burning on residential properties, as greater volumes of material are likely to be involved. In order to assist in mitigating these effects, the rule requires burning to be undertaken more than 100 metres from the boundary of the property.

Rule 16.3.2.3 applies to outdoor burning on all properties, other than production land, such as, but not limited to, residential properties, educational facilities, industrial or trade premises, parks and reserves, Air Zone 3. It adopts a similar level of control to that applying in Air Zones 1 and 2, except that a separation distance is not required in recognition of the often lower density of development in these areas, and the lower incidence of complaints received from them. Furthermore, if the Plan did not permit this activity on industrial or trade premises, it would require a resource consent under Section 15(1) of the Resource Management Act.

Rule 16.3.2.4 applies to outdoor burning on all production land properties, in Air Zone 3. The level of control reflects the fact that such properties are normally of larger size, and few if any complaints are received from burning waste on production land. Note that Rule 16.3.3.1 does not prohibit the burning of animal carcasses on this production land.

Rule 16.3.2.5 applies to minor or infrequent outdoor burning on any Otago property, for the purposes of backyard cooking of food, such as in any barbecue, hangi or umu, or for a campfire or a celebratory fire, such as for Guy Fawkes or New Year's Eve. Conditions relating to separation distances or that the material burnt be generated on the property are not required, in recognition of the small scale or infrequent occurrence of these activities.

Where a discharge into air from outdoor burning does not comply with any of Rules 16.3.2.1 to 16.3.2.5 (whichever is applicable), and is not prohibited by Rule 16.3.3.1, it becomes a discretionary activity in terms of **Rule 16.3.2.6** and a consent is required to be obtained. This allows any adverse effect to be assessed.

16.3.3 Burning or incineration of specified materials

Note:

- 1. City and district councils in implementing the Forests and Rural Fires Act 1977 may also have bylaws controlling outdoor burning of materials for fire safety purposes.
- 2. The Health Act 1956 also has some control on nuisance effects from the discharge of contaminants into air.

16.3.3.1 Burning or incineration of specified materials - prohibited activity

Except as provided for by Rules 16.3.3.2 and 16.3.13.2, the discharge of contaminants into air from the burning or incineration of any of the following materials:

- (a) Chlorinated organic chemicals including, but not limited to, dioxins, furans, polychlorinated biphenyls (PCB);
- (b) Contaminated material from contaminated sites and buildings;
- (c) Food waste;
- (d) Materials containing heavy metals;
- (e) Material associated with the recovery of metal from coated or covered cables;
- (f) Motor vehicles and vehicle parts;
- (g) Materials containing mineral fibres including but not limited to asbestos;
- (h) Paint and other surface coatings;
- (i) Pathological materials excluding animal carcasses on production land;
- (j) Agrichemicals and agrichemical waste;
- (k) All plastic, including, but not limited to, polyvinylchloride (PVC), polystyrene, nylon, styrofoam, but not including polyethylene;
- (l) Tyres and other rubber;
- (m) Timber treated with copper, chrome and arsenic (CCA) or organochlorine preservatives;
- (n) Waste oil or other waste petroleum products; or
- (o) Sewage sludge and associated solids, or solids derived from liquid-borne municipal, industrial or trade waste; or
- (p) Asphalt surfaces (seal burning);

is a *prohibited activity*, for which no consent will be granted.

16.3.3.2 Discharges from the burning or incineration of materials specified in Rule 16.3.3.1 - discretionary activity

The discharge of contaminants into air from the burning or incineration of:

- (1) Any material specified in Rule 16.3.3.1, in an incinerator or crematorium; or
- (2) Waste oil, in a frost pot, or fuel burning equipment;

is a discretionary activity.

Principal reasons for adopting

Rules 16.3.3.1 and 16.3.3.2 recognise that the burning of the materials specified in Rule 16.3.3.1 can result in the discharge of hazardous air contaminants (identified in Schedule 3), in addition to effects such as smoke and odour which are common to the burning of all waste. Contaminants of this nature have been identified by the Ministry for the

Environment as being known, or suspected, to cause significant adverse effects on human health and the environment.

Rule 16.3.3.1 recognises that when the materials specified are burnt on the open ground or by uncontrolled means it is not possible to take any measures to manage the discharge of hazardous air contaminants. Because of this, adverse effects can be avoided only through prohibiting this type of burning.

Rule 16.3.3.2 recognises that, in some circumstances, the materials specified in Rule 16.3.3.1 are best disposed of by incineration. There is, however, considerable variation in the technical specifications of incinerators and some are more suited to burning the specified materials than others. Because of this and the fact that the burning of these materials may have significant adverse effects, the management of these effects needs to be considered on a case-by-case basis through a resource consent process. For example, it may be appropriate to incinerate waste oil in fuel burning equipment such as boilers and kilns, but the effects need to be assessed through a consent process.

16.3.4 Products of combustion from fuel burning equipment

16.3.4.1 Discharges from fuel burning equipment in Air Zone 1 or 2 - permitted activity

The discharge into air of products of combustion arising from fuel burning equipment from single activities or a combination of activities located on one site (excluding domestic heating appliances subject to Rules 16.3.1.1 to 16.3.1.6) in Air Zone 1 or 2 which:

- (1) Does not exceed a heat generation capacity of 1MW; or
- (2) Does not exceed a heat generation capacity of 5MW and burns only gas, oil or bio-oils (excluding waste oil) with a sulphur content of less than 1%;

is a *permitted activity*, providing:

- (a) In the case of equipment installed after 28 February 1998, any chimney complies with Schedule 6 ("Determination of Chimney Heights"); and
- (b) No material specified in Rule 16.3.3.1 is burnt; and
- (c) Any discharge of smoke, odour, particulate matter or gases is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.4.2 Discharges from fuel burning equipment in Air Zone 3 - permitted activity

The discharge into air of products of combustion arising from fuel burning equipment (excluding domestic heating appliances subject to Rules 16.3.1.1 to 16.3.1.6) located on a site in Air Zone 3 which does not exceed a heat generation capacity of 5MW (from single activities or a combination of activities located on one site);

is a *permitted activity*, providing:

- (a) In the case of equipment installed after 28 February 1998, any chimney complies with Schedule 6 ("Determination of Chimney Heights"); and
- (b) No material specified in Rule 16.3.3.1 is burnt; and
- (c) Any discharge of smoke, odour, particulate matter or gases is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.4.3 Other discharges of the products of combustion from fuel burning equipment – discretionary activity

Except as provided for by Rule 16.3.4.1 or 16.3.4.2, the discharge into air of products of combustion from fuel burning equipment is a *discretionary activity*.

Principal reasons for adopting

Rule 16.3.4.1 permits minor discharges into air from fuel burning equipment in Air Zone 1 or 2. It recognises that allowing larger scale discharges without consent in these areas is inappropriate given that ambient air quality in Air Zone 1 or 2 is more degraded than elsewhere in Otago (Air Zone 3).

The rule also makes a distinction between fuel types due to the difference in potential effects. Gas fuels, light oils and bio-oils are easy to burn cleanly. Combustion of other fuels such as coal, wood or high sulphur oils can result in a greater discharge of some contaminants. For example, coal boilers typically discharge between 4 and 8 kg of particulate per tonne of fuel, and some can discharge up to 30 kg per tonne depending on the type of equipment and the ash content of the coal. As a comparison, a light fuel oil-fired boiler produces less than 0.3 kg per tonne, while particulate emissions from gas combustion are negligible.

Oils with a sulphur content of greater than 1%, and some coals, have the potential to discharge significant quantities of sulphur dioxide to the atmosphere. Fuel oils with a high sulphur content also tend to have a correspondingly high potential to produce particulate discharges. This is because the sulphur content is related to ash content and factors such as viscosity which affects atomisation of the fuel in a burner, thereby making it more difficult to burn cleanly.

Given these factors, it is not appropriate to treat all fuels in the same way. Fuels other than gas, light oil or bio-oils must be treated in a precautionary manner.

Rule 16.3.4.2 recognises that in Air Zone 3, discharges from fuel burning equipment will have less effect. This is because there is a lower density of combustion discharges.

Where a discharge into air from fuel burning equipment does not comply with either Rules 16.3.4.1 or 16.3.4.2 (whichever is applicable), and is not prohibited by Rule 16.3.3.1, it becomes a discretionary activity in terms of Rule 16.3.4.3 and a consent is required to be obtained. Note that Rule 16.3.3.2 allows waste oil to be incinerated in fuel burning equipment as a discretionary activity. This allows any adverse effect to be assessed.

16.3.5 Discharges from industrial or trade processes

Note:

- 1. Industrial or trade processes may also be subject to city or district council bylaws or district plan rules.
- 2. Rules 16.3.5.2 and 16.3.5.6 apply to discharges from activities on ships.
- 3. Discharges from fuel burning equipment used in the following processes are dealt with by Rules in 16.3.4, and are not covered by the following rules.

16.3.5.1 Discharges from the processing of plant or animal matter - permitted activity

The discharge of contaminants into air from:

- (1) Deep fat frying, oil frying, roasting, drying, boiling, baking or smoking of plant or animal matter where the total raw material capacity is less than 250 kg/hr; or
- (2) Food preparation or cooking activities associated with restaurants, and other similar premises including takeaway bars, and the making of bread and cheese; or
- (3) The drying of milk or milk products where the total raw material capacity is less than 250 kg/hr; or
- (4) Wine making, brewing and other fermentation processes for production of food or beverages; or
- (5) The slaughter or skinning of animals; or
- (6) The drying of grain where the total raw material capacity on site is less than 1000 kg/hr for activities located in Air Zone 1 or 2 and less than 5000 kg/hr for activities located in Air Zone 3;

is a *permitted activity*, providing:

- (a) In the case of equipment installed after 28 February 1998, any chimney complies with Schedule 6 ("Determination of Chimney Heights"); and
- (b) Any discharge of odour, or particulate matter, including fat and oils, is not offensive or objectionable at or beyond the boundary of the property.

16.3.5.2 Discharges from the sorting, crushing, screening, conveying and storage of powdered or bulk products - permitted activity

The discharge of contaminants into air from the sorting, crushing, screening, storage and conveying (including loading and unloading) of fertilisers, grains, berries, coal, coke, wood chips, sawdust, wood shavings, bark, sand, aggregates, and other powdered and bulk products whether in dry or liquid form, where:

- (1) The total capacity of outside storage of bulk materials is less than 1,000 m³ if located on a site in Air Zone 1 or 2; and
- (2) The crushing and screening of bulk materials is at a rate less than 100 tonnes an hour;

is a *permitted activity*, providing any discharge of odour, or particulate matter is not offensive or objectionable at or beyond the boundary of the property.

16.3.5.3 Discharges from mineral extraction and processing - permitted activity

The discharge of contaminants into air from:

- (1) The extraction of minerals from the surface or from an open pit at a rate less than 20,000 cubic metres per month and 100,000 cubic metres per year; or
- (2) The crushing and screening of minerals at a rate less than 200 tonnes an hour; or
- (3) The drying or heating of minerals from single activities or a combination of activities on one site with equipment that has a heat generation capacity of less than 500 kW; or
- (4) The making of refractory, bricks or ceramic products at a rate less than 200 kg/hr of products;

is a *permitted activity*, providing:

- (a) The mineral extraction, crushing and screening activities are located in Air Zone 3; and
- (b) In the case of equipment installed after 28 February 1998, any chimney complies with Schedule 6 ("Determination of Chimney Heights"); and
- (c) Any discharge of smoke, odour or particulate matter is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.5.4 Discharges from chemical processing, manufacturing and industrial or trade processes which discharge hazardous air contaminants - permitted activity

The discharge of contaminants into air from single activities or a combination of activities located on one site involving:

- (1) Processes used for water treatment by chlorine or ozone, where less than 10 kg/hr of chlorine or chlorinated chemicals are used; or
- (2) Chemical manufacturing or processing where the discharge of sulphur dioxide or nitrogen dioxide is less than 5 kg/hr; or
- (3) Processes that discharge less than 0.01 kg/hr of Category A, B1 and B2 carcinogens as identified in Schedule 3; or
- (4) Processes that discharge less than 0.01 kg/hr of heavy metals; or
- (5) Processes that discharge less than 1 kg/hr of hazardous air contaminants identified in Schedule 3, excluding heavy metals and Category A, B1 and B2 carcinogens as identified in Schedule 3;

is a *permitted activity*, providing:

- (a) In the case of equipment installed after 28 February 1998, any chimney complies with Schedule 6 ("Determination of Chimney Heights"); and
- (b) Any discharge of odour, particulate matter or gases is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.
16.3.5.5 Discharges from metal processing and foundries - permitted activity

The discharge of contaminants into air in connection with the melting of lead, aluminium, iron, copper, brass or bronze on a site where the aggregated melting capacity is less than 250 kg/hr;

is a *permitted activity*, providing:

- (a) In the case of equipment installed after 28 February 1998, any chimney complies with Schedule 6 ("Determination of Chimney Heights"); and
- (b) Any discharge of odour, particulate matter, gases or fumes is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.5.6 Discharges from surface coating (including spray painting) and diisocyanate use - permitted activity

The discharge of contaminants into air from:

- (1) Permanent surface coating facilities including single activities or a combination of activities located on one site involving:
 - (i) The application of coating materials not containing di-isocyanates (including paint, paint solvents, varnish, lacquer, dyes, metal oxide coatings, adhesive coatings, elastomer coatings, stains and polishes) where less than 2 tonnes per month and 20 tonnes per year of coating is used, and less than 20 litres per week of organic solvents are used; or
 - (ii) The application of di-isocyanate coating materials or the manufacture of polyurethane foams, where less than 1 kg/hr of di-isocyanate is used; or
 - (iii) The stoving, curing, baking or drying of coating materials (including paint, paint solvents, varnish, lacquer, dyes, metal oxide coatings, adhesive coatings, elastomer coatings, stains and polishes) by heat, with equipment that has a heat generation capacity of less than 500 kW; or
- (2) The surface coating (including spray painting) of roads, buildings, bridges and other structures, by mobile equipment;

- (a) In the case of equipment installed after 28 February 1998, any chimney complies with Schedule 6 ("Determination of Chimney Heights"); and
- (b) Any overspray or discharge of odour or particulate matter is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property; and
- (c) Any overspray or discharge of odour or particulate matter in a public amenity area is not noxious, dangerous, offensive or objectionable at or beyond 20 metres from the discharge.

16.3.5.7 Discharges from petroleum and hydrocarbon processes - permitted activity

The discharge of contaminants into air from:

- (1) The handling and purification by distillation of dry cleaning substances at retail outlets; or
- (2) The handling and storage of hydrocarbons at service stations, bulk terminals and other facilities concerned with marketing and delivery of fuels and lubricants; or
- (3) Processes not listed above that involve the handling and processing of hydrocarbons where the discharge from single activities or a combination of activities located on one site is less than 1 kg/hr of hazardous air contaminants, excluding Category A, B1 and B2 carcinogens as identified in Schedule 3; or
- (4) Processes not listed above that involve the handling and processing of hydrocarbons where the discharge from single activities or a combination of activities located on one site is less than 0.01 kg/hr of Category A, B1 and B2 carcinogens as identified in Schedule 3;

is a *permitted activity*, providing:

- (a) In the case of equipment installed after 28 February 1998, any chimney complies with Schedule 6 ("Determination of Chimney Heights"); and
- (b) Any discharge of odour or gases is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.5.8 Discharges from the processing of wood and wood products - permitted activity

The discharge of contaminants into air from:

- (1) The drying of timber in a kiln at a rate less than 2000 cubic metres per month and 20,000 cubic metres of timber per year from single activities or a combination of activities located on one site; or
- (2) Timber sawing, milling, joining and moulding, where the timber is processed at a rate less than 2000 cubic metres per month and 20,000 cubic metres per year, from single activities or a combination of activities located on one site;

- (a) In the case of equipment installed after 28 February 1998, any chimney complies with Schedule 6 ("Determination of Chimney Heights"); and
- (b) Any discharge of smoke, odour or particulate matter is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.5.9 Other discharges from industrial or trade processes – discretionary activity

Except as provided for by Rules 16.3.5.1 to 16.3.5.8 and 16.3.6.1, 16.3.6.2, 16.3.7.1, 16.3.9.2, 16.3.10.1, 16.3.10.2, 16.3.11.1, 16.3.13.1 and 16.3.13.2, or prohibited by Rule 16.3.3.1, the discharge of contaminants into air from industrial or trade processes is a *discretionary activity*.

Principal reasons for adopting

Discharges of contaminants into air from industrial or trade premises can only occur if expressly allowed by a rule in a regional plan and proposed regional plan, a resource consent or by regulations (Section 15(1) of the Resource Management Act).

The activities permitted by **Rules 16.3.5.1**, to **16.3.5.8** involve small scale discharges and, provided the conditions of the rules are met, these activities will have no more than minor adverse effects on the environment.

Where a discharge into air from industrial or trade processes does not comply with any of Rules 16.3.5.1 to 16.3.5.8 (whichever is applicable), and is not prohibited by Rule 16.3.3.1, it becomes a discretionary activity in terms of **Rule 16.3.5.9** and a consent is required to be obtained. This allows any adverse effect to be assessed.

16.3.6 Abrasive blasting

Note:

- 1. City and district councils may have controls in relation to abrasive blasting activities, particularly in terms of noise.
- 2. Rules in this section apply to discharges from abrasive blasting on ships.

16.3.6.1 Discharges from wet or vacuum abrasive blasting - permitted activity

The discharge of dust or contaminants into air from wet or vacuum abrasive blasting, whether mobile or in permanent facilities;

- (a) Any unused abrasive media or waste material is kept covered or stored so that it cannot be blown around by wind; and
- (b) Waste material is removed from temporary sites at the completion of the project; and
- (c) Any overspray or discharge of odour or particulate matter is not noxious, dangerous, offensive or objectionable beyond the boundary of the property; and
- (d) Any overspray, or discharge of odour or particulate matter in a public amenity area, is not noxious, dangerous, offensive or objectionable beyond 20 metres from the discharge.

16.3.6.2 Discharges from dry abrasive blasting - permitted activity

The discharge of dust or contaminants into air from dry abrasive blasting, whether mobile or in permanent facilities;

is a *permitted activity*, providing:

- (a) Any unused abrasive media or waste material is kept covered or stored so that it cannot be blown around by wind; and
- (b) Waste material is removed from temporary sites at the completion of the project; and
- (c) In the case of permanent facilities all items are blasted within a building or enclosure (shroud, screen or cover) from which the only discharge into air is from an exhaust system fitted with emission controls designed to achieve an emission less than 125 mg/Nm³ of dust at 0°C and 101.3 kPa on a dry gas basis; and
- (d) In the case of mobile operations all items are blasted within a building or enclosure (shroud, screen or cover); and
- (e) Any overspray or discharge of odour or particulate matter is not noxious, dangerous, offensive or objectionable beyond the boundary of the property; and
- (f) Any overspray or discharge of odour or particulate matter in a public amenity area, is not noxious, dangerous, offensive or objectionable beyond 20 metres from the discharge.

16.3.6.3 Other discharges from abrasive blasting – discretionary activity

Except as provided for by Rule 16.3.6.1 or 16.3.6.2, the discharge of contaminants into air from abrasive blasting is a *discretionary activity*.

Principal reasons for adopting

Rules 16.3.6.1 and 16.3.6.2 have been adopted to avoid abrasive blasting giving rise to off-site nuisance effects.

The rules do not contain conditions limiting the blasting media which may be used or the material which may be blasted. While certain contaminants may give rise to adverse health effects such effects are usually limited to within the site and therefore addressed under the Health and Safety in Employment Act 1992. Furthermore, any off-site health effects that may occur are likely to become evident well after the activity has given rise to nuisance effects, and thus at a time when the operation would already be undergoing review (in order to meet the permitted activity standards) or consents were being applied for. Off-site health effects are one matter which the Council will consider when processing such an application.

The rules do not address the effects of discharges from abrasive blasting into water or onto land where material may enter water. These effects are addressed in the Regional Plan: Water for Otago.

Rule 16.3.6.1 is adopted to permit wet or vacuum abrasive blasting as these processes are unlikely to have any significant adverse environmental effects on the environment. Use of wet abrasive blasting confines the contaminants to the water stream and minimises aerial dispersion. Vacuum blasting also minimises aerial dispersion as the vacuum

method enables greater control of waste material. Notwithstanding this, the Council has adopted a level of control through the stated conditions which enables control to be exercised in circumstances where this is warranted.

Rule 16.3.6.2 is adopted to permit the discharge of contaminants into air from dry abrasive blasting. The conditions required for compliance with this rule are more restrictive than those required for wet or vacuum blasting under Rule 16.3.6.1. This is because dry abrasive blasting results in much greater concentrations of dust which tend to drift further. It also recognises the Council's preference for wet or vacuum blasting to be used as an alternative to dry abrasive blasting.

Where a discharge into air from abrasive blasting does not comply with either Rules 16.3.6.1 or 16.3.6.2 (whichever is applicable) it becomes a discretionary activity in terms of Rule 16.3.6.3 and a consent is required to be obtained. This allows any adverse effect to be assessed.

16.3.7 Waste management

16.3.7.1 Discharges from the storage, transfer, treatment and disposal of liquidborne municipal, industrial or trade waste - permitted activity

The discharge of contaminants into air from the storage, transfer, treatment or disposal (including land application of treated effluent and sludge, but excluding the burning of sludge and associated solids) of liquid-borne municipal, industrial or trade waste, where the influent liquid waste does not exceed a BOD₅ of 850 kg per day;

is a *permitted activity*, providing:

- (a) Ponds constructed after 1 January 2002 are located at least 150 metres from the closest part of the boundary of the property; and
- (b) Land application does not occur within:
 - (i) 150 metres from any residential dwelling on a neighbouring property or from a building used for employment purposes on a neighbouring property; and
 - (ii) 20 metres from a formed public road; and
 - (iii) 150 metres from any public amenity area or place of public assembly, excluding formed public roads, and
- (c) **Any discharge** of odour, particulate matter, droplets or gases is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.7.2 Discharges from farm waste management - permitted activity

The discharge of contaminants into air from farm waste management involving:

- (1) The storage, transfer and disposal of animal bedding, manure and other biological material from intensive farming; or
- (2) The treatment, storage and disposal via land application of liquid animal effluent and sludge;

- (a) Ponds constructed after 30 June 2001 and outdoor stockpiles of waste animal bedding, manure and other biological material are located at least 150 metres from the closest part of the boundary of the property; and
- (b) Land application does not occur within:
 - (i) 150 metres from any residential dwelling on a neighbouring property or from a building used for employment purposes on a neighbouring property; and
 - (ii) 20 metres from a formed public road; and
 - (iii) 150 metres from any public amenity area or any place of public assembly, excluding formed public roads; and
- (c) The volume of liquid waste involved does not exceed a BOD₅ of 700 kg per day; and
- (d) Any discharge of odour, particulate matter, droplets or gases is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.7.3 Other discharges from waste management – discretionary activity

Except as provided for by Rule 16.3.7.1 or 16.3.7.2, the discharge of contaminants into air from waste management is a *discretionary activity*.

Principal reasons for adopting

Rules 16.3.7.1 and 16.3.7.2 recognise that the Regional Plan: Water for Otago is the main mechanism by which the Council controls the operation of liquid effluent management processes, and that if the processes are operating efficiently there should be no significant adverse effects on air quality. Should odours occur, they would represent a failure of the treatment processes which would be primarily addressed via enforcement of the Regional Plan: Water rules or conditions of a discharge permit.

Rule 16.3.7.1 recognises Section 418 of the Resource Management Act which requires any industrial or trade premises used for the storage, transfer, treatment, or disposal of waste materials or other waste management purposes which commenced after the 1st day of October 1991, to have a resource consent to discharge contaminants into air unless that discharge is provided for by a rule in a regional plan. This rule is included to enable activities which have no more than minor adverse effects to be undertaken without the need for consents.

Rule 16.3.7.2 recognises that on-farm waste management processes can result in the emission of odours. This rule introduces separation distances between new ponds or land application areas and residential dwellings, public roads or places of assembly, with a view to avoiding adverse effects of odour beyond the boundary of the property where the waste management is undertaken.

Where a discharge into air from waste management does not comply with either Rules 16.3.7.1 or 16.3.7.2 (whichever is applicable) it becomes a discretionary activity in terms of Rule 16.3.7.3 and a consent is required to be obtained. This allows any adverse effect to be assessed.

16.3.8 Intensive farming

Note:

1. Property owners/managers should consult with their local city or district council and assess whether the applicable district plan contains any controls on location, noise levels or any other activities which may affect the operation. Land use consents may be required from city and district councils.

16.3.8.1 Discharges from intensive farming - permitted activity

The discharge of contaminants into air from ventilation associated with intensive farming;

is a *permitted activity*, providing:

- (a) No more than 2000 pigs or 100,000 poultry are kept at any one time; and
- (b) Any discharge of odour or particulate matter is not offensive or objectionable at or beyond the boundary of the property.

16.3.8.2 Other discharges from intensive farming – discretionary activity

Except as provided for by Rule 16.3.8.1, the discharge of contaminants into air from intensive farming is a *discretionary activity*.

Principal reason for adopting

Rule 16.3.8.1 has been adopted to allow small scale discharges without consent, provided there is no noxious, dangerous, offensive or objectionable odour or dust at or beyond the boundary of the property. The Council has however retained control over large scale pig and poultry farming operations because they can have more than minor adverse effects on the environment, particularly in terms of particulate matter and odour.

Where a discharge into air from intensive farming does not comply with Rule 16.3.8.1, it becomes a discretionary activity in terms of **Rule 16.3.8.2** and a consent is required to be obtained. This allows any adverse effect to be assessed.

16.3.9 Agrichemical application

Note:

- 1. District plans may have land use rules relating to agrichemical application and should be referred to in connection with these permitted activity rules.
- 2. In carrying out any agrichemical application in terms of the following rules, the practices recommended in Section 5 of the *Code of Practice for the Management of Agrichemicals* (NZS 8409:1999; New Zealand Agrichemical Education Trust, August 1999), or in Schedule 4 of this Plan which is based on that code, should be used, noting that to do so does not negate the requirement to meet rule conditions.
- 3. Signage on public roads (Rule 16.3.9.3) needs to meet the requirements of the appropriate road controlling authority (city or district councils in the case of local roads and Transit New Zealand in the case of state highways).

16.3.9.1 Discharges from agrichemical application on residential properties - permitted activity

The discharge of any agrichemical into air arising from the application of any agrichemical using hand held appliances:

- (1) For domestic purposes within a residential property; or
- (2) On the residential portion of any other property; or
- (3) On road frontages of residential properties where the spray is applied by the residents of the adjoining residential property;

is a *permitted activity*, providing:

- (a) The agrichemical and any associated additive are authorised for use in New Zealand and are used in accordance with the authorisation; and
- (b) The discharge is carried out in accordance with the manufacturer's directions; and
- (c) The discharge does not exceed the quantity, concentration or rate required for the intended purpose; and
- (d) The application does not result in any ambient concentrations of contaminants at or beyond the boundary of the property that have noxious or dangerous effects.

16.3.9.2 Discharges from agrichemical application on production land and industrial or trade premises - permitted activity

The discharge of any agrichemical into air using aerial or ground based application methods:

- (1) On production land; or
- (2) On roadsides adjoining production land when applied by the adjacent landowner or his/her employee; or
- (3) On industrial or trade premises;

is a *permitted activity*, providing:

- (a) The agrichemical and any associated additive are authorised for use in New Zealand and are used in accordance with the authorisation; and
- (b) The discharge is carried out in accordance with the manufacturer's directions; and
- (c) The discharge does not exceed the quantity, concentration or rate required for the intended purpose; and
- (d) The application does not result in any ambient concentrations of contaminants at or beyond the boundary of the property that have noxious or dangerous effects.

16.3.9.3 Discharges from agrichemical application in public amenity areas and places of public assembly - permitted activity

Except as provided for in Rules 16.3.9.1 and 16.3.9.2, the discharge of agrichemicals in public amenity areas or in places of public assembly using aerial or ground based application methods;

- (a) The agrichemical and any associated additive are authorised for use in New Zealand and are used in accordance with the authorisation; and
- (b) The discharge is carried out in accordance with the manufacturer's directions; and
- (c) The discharge does not exceed the quantity, concentration or rate required for the intended purpose; and
- (d) The application does not result in any ambient concentrations of contaminants at or beyond the boundary of the property that have noxious or dangerous effects; and
- (e) Every person applying agrichemicals after 1 March 1999 using:
 - (i) Hand-held appliances with a maximum capacity of up to 15 litres shall hold a current Standard Growsafe Certificate; and
 - (ii) Ground-based application methods including hand-held appliances with a maximum capacity in excess of 15 litres shall hold a current Growsafe Registered Chemical Applicators Certificate; and
- (f) The application is undertaken in accordance with Section 5 of the Code of Practice for the Management of Agrichemicals (NZS 8409:1999; New Zealand Agrichemical Education Trust, August 1999); and
- (g) Signs advising that spraying is in progress are placed at all points where the public commonly have entry when application occurs in parks, reserves, gardens and sports grounds, but not alongside public roads or railways; and
- (h) Where signs are required under Condition (g), adequate signage is maintained in place to ensure the public do not enter the affected land until the re-entry period for that particular chemical has expired; and
- (i) Where application occurs alongside public roads and railways, vehicles associated with the spraying shall display prominent signs advising that spraying is in progress.

16.3.9.4 Other discharges from agrichemical application – discretionary activity

Except as provided for by Rules 16.3.9.1 to 16.3.9.3, the discharge of contaminants into air from agrichemical application is a *discretionary activity*.

Principal reasons for adopting

Rules 16.3.9.1, 16.3.9.2 and 16.3.9.3 have been adopted to enable discharges from agrichemical spraying to occur without the need for consent, provided that any off-site and non-target effects are no more than minor. This will avoid or mitigate the adverse effects of agrichemicals on human health and safety, sensitive areas or places, and non-target areas or species. The three rules distinguish between the application of agrichemicals on the basis of land use.

The rules do not address the effects of agrichemical application into water or onto land where the agrichemicals may enter water. These effects are addressed in the Regional Plan: Water for Otago. Agrichemical application in the coastal marine area is controlled by the Regional Plan: Coast for Otago.

Pilots are required by the Civil Aviation Act to hold an agrichemical rating certificate if they are applying agrichemicals from planes or helicopters. Pilots are also subject to enforcement action from Civil Aviation. Thus, while there is a greater potential for spray drift to occur from aerial application if good management practices are not followed, no additional controls are contained in this Plan because aerial applicators are already required to undergo additional training. There is no need to duplicate the provisions available in other institutional arrangements by including the same requirement in rules in this Plan.

Rule 16.3.9.1 restricts the application of agrichemicals on residential properties unless it is undertaken in a manner specified by the manufacturer of the agrichemicals. This rule recognises that the application of agrichemicals on residential properties typically involves only small quantities of agrichemicals using hand held application methods which are generally operated at ground level and at low pressures. Given these circumstances the adverse effects of agrichemical application are generally insignificant, thus the Council has adopted a level of control which enables it to apply performance standards to avoid the misuse of agrichemicals and to avoid the occurrence of adverse effects such as damage to the vegetation on adjoining properties.

Rule 16.3.9.2 permits the application of agrichemicals on production land, on roadsides adjoining production land when applied by the adjacent landowner or his/her employee and on industrial or trade premises. This is because there are greater quantities of agrichemicals applied on production land and the application methods commonly used are more likely to result in spray drift. The rule also allows landowners to control plant pests on road frontages adjoining their properties.

Rule 16.3.9.3 permits the application of agrichemicals in public amenity areas and places of public assembly subject to compliance with the stated conditions. The conditions are more restrictive than those for Rules 16.3.9.1 and 16.3.9.2 recognising that public interest in these areas is higher than in those areas where access is either restricted or not provided.

Where a discharge into air from agrichemical application does not comply with any of Rules 16.3.9.1 to 16.3.9.3 (whichever is applicable), it becomes a discretionary activity in terms of **Rule 16.3.9.4** and a consent is required to be obtained. This allows any adverse effect to be assessed.

16.3.10 Water vapour, heat and energy

16.3.10.1 Discharges of steam or water vapour - permitted activity

Except as provided for by the Rules in section 16.3.1 to 16.3.9, the discharge of condensed water vapour, including steam, into air;

- (a) Any plume does not impair visibility on any road or in any aircraft flight path; and
- (b) There is no drift of a noxious, dangerous, offensive or objectionable plume beyond the boundary of the property; and
- (c) There is no venting of steam or water directly above footpaths or onto or over other properties.

16.3.10.2 Discharges of energy, including heat - permitted activity

Except as provided for by the Rules in section 16.3.1 to 16.3.9, the discharge into air of:

- (1) Air heated above ambient temperature, including but not limited to heated air from heat exchangers, and air used for the purpose of cooling plant and equipment; or
- (2) Energy from sources of electromagnetic radiation, including radio and television transmitters and other telecommunications facilities, cell phones, and generators;
- (3) Heat generated by transmission lines and ancillary equipment;

is a *permitted activity*.

Note:

- 1. The siting of radio and telephone transmitters and other telecommunication facilities is controlled by city and district councils.
- 2. The control of ionising radiation (including x-rays and gamma rays) is administered by the National Radiation Laboratory under the Radiation Protection Act. Nonionising radiation is mainly associated with the broadcasting and communication industries and there is a New Zealand Standard available for the control of such radiation.

16.3.10.3 Other discharges of water vapour, heat or energy – discretionary activity

Except as provided for by Rule 16.3.10.1 or 16.3.10.2, the discharge of contaminants into air of water vapour, heat or energy is a *discretionary activity*.

Principal reasons for adopting

Rule 16.3.10.1 provides performance standards to ensure that the discharge of steam or water vapour does not cause a traffic hazard, does not adversely affect amenity values and does not deposit on surfaces where it may freeze, thereby creating a potential risk to public safety. Discharges of steam do not produce any other adverse effects on the environment and it is unnecessary to require any regulation of discharge quality.

Rule 16.3.10.2 provides certainty to users and contributes to administrative efficiency for managing activities that either have no adverse effects on air quality or where any adverse effects will be minor.

Part (1) provides for discharges of heat where no combustion has occurred. The effects of such discharges on a local scale are minimal since heat disperses rapidly in the air.

Part (2) provides for the release of energy from radio and television and the like. These sources generate relatively low radiation, which have only minor effects on air quality. Furthermore as such sources are found throughout the country, the discharge of energy from such appliances and equipment is considered to be more appropriately addressed at the national level. Permitting these activities avoids the duplication of other requirements and controls relating to electromagnetic radiation. Part (3) recognises that some electrical 'process loss' (in the form of heat) can occur, but its effects are no more than minor.

Where a discharge into air of water vapour, heat or energy does not comply with either Rules 16.3.10.1 or 16.3.10.2 (whichever is applicable), it becomes a discretionary activity in terms of **Rule 16.3.10.3** and a consent is required to be obtained. This allows any adverse effect to be assessed.

16.3.11 Ventilation or vapour displacement

16.3.11.1 Discharges for the purpose of ventilation or vapour displacement - permitted activity

The discharge of contaminants into air from:

- (1) Ventilation from indoor working spaces, unless otherwise covered in this Plan; or
- (2) Fume cupboards; or
- (3) Tanks used for the storage of liquids, excluding the storage of hydrocarbons as permitted by Rule 16.3.5.7 on any industrial or trade premises; or
- (4) The venting of gas pipelines, pumps, compressors, tanks or associated equipment when released for the purposes of refilling, servicing or repair, on any industrial or trade premises;

is a *permitted activity*, providing any discharge of odour or particulate matter is not noxious, dangerous, offensive or objectionable beyond the boundary of the property.

16.3.11.2 Other discharges from ventilation or vapour displacement – discretionary activity

Except as provided for by Rule 16.3.11.1, the discharge of contaminants into air from ventilation or vapour displacement is a *discretionary activity*.

Principal reason for adopting

Rule 16.3.11.1 provides for the discharge of contaminants into air associated with ventilation. Most ventilation generates only negligible or minor adverse effects on the environment provided certain conditions are met. The conditions of this rule are concerned with addressing the potential effects of the discharge beyond the boundary. On-site effects are generally managed under other legislation, such as the Health and Safety in Employment Act 1992.

Where a discharge into air from ventilation or vapour displacement does not comply with Rule 16.3.11.1, it becomes a discretionary activity in terms of **Rule 16.3.11.2** and a consent is required to be obtained. This allows any adverse effect to be assessed.

16.3.12 Nuclear power generation or nuclear weapon manufacturing

16.3.12.1 Discharges from nuclear power generation or nuclear weapon manufacturing - prohibited activity

The discharge of contaminants into air from nuclear power generation or nuclear weapon manufacturing is a *prohibited activity*, for which no consent will be granted.

Principal reason for adopting

Rule 16.3.12.1 is adopted to provide for Policy 12.5.1 of the Regional Policy Statement for Otago. The rule prohibits all discharges from nuclear power generation plants or nuclear weapon manufacturing, some of which may have otherwise been permitted by this Plan.

16.3.13 General permitted activities

16.3.13.1 Discharges from miscellaneous activities - permitted activity

The discharge of contaminants into air from:

- (1) Vehicle engine maintenance and servicing; or
- (2) Building and construction activities, including road construction and maintenance, but excluding the remediation of asphalt surfaces (seal burning); or
- (3) Any outdoor general engineering activity; or
- (4) Using a smoke tracer to test underground drains or sewers; or
- (5) Fertiliser application; or
- (6) The engine exhaust of any motor vehicle, train or aircraft on an industrial or trade premises; or
- (7) Any smokeless heater frost-fighting device not using waste oil;

is a *permitted activity*, providing any discharge of smoke, odour, particulate matter or gas is not noxious, dangerous, offensive or objectionable at or beyond the boundary of the property.

16.3.13.2 Discharges from burning by New Zealand Fire Service - permitted activity

The discharge of contaminants into air from the burning of any material for training activities or fire safety research or educational purposes, carried out by the New Zealand Fire Service, even though material specified in Rule 16.3.3.1 may be present in structures or vehicles burnt;

is a *permitted activity*, providing:

- (a) The Fire Service carries out no more than a total of three fires burning a structure or vehicle, per calendar year, within each territorial authority district; and
- (b) The New Zealand Fire Service ensures every occupier of land within 500 metres of any structure or vehicle to be burnt, and the road-controlling authority for any road within 500 metres of that structure or vehicle, is notified at least two days prior to the burning; and
- (c) No material specified in Rule 16.3.3.1 is added to any structure or vehicle to be burnt; and
- (d) There is no asbestos present in any material to be burnt.

Principal reasons for adopting

Rule 16.3.13.1 and 16.3.13.2 recognise that some activities involve only small scale or infrequent discharges and, provided the conditions of the rules are met, those activities will have no more than minor adverse effects on the environment.

16.3.14 Other discretionary activities

16.3.14.1 Discretionary activities (general rule)

The discharge of contaminants into air from any process or activity on an industrial or trade premises:

- (1) Excluding any discharge associated with the following activities regulated by the Regional Plan: Waste:
 - (i) A contaminated site;
 - (ii) A facility for the treatment or disposal of hazardous wastes;
 - (iii) A new or operating landfill;
 - (iv) A closed landfill;
 - (v) An offal pit on production land, intensive farm, or industrial or trade premises;
 - (vi) A farm landfill;
 - (vii) Composting or silage production; or
 - (viii) A greenwaste landfill; and
- (2) Which is not expressly provided for by the rules of this Plan; and
- (3) Which is not a prohibited activity under Rule 16.3.1.1, 16.3.3.1 or 16.3.12.1;

is a *discretionary activity*.

Principal reasons for adopting

Some discharges into air are provided for as permitted activities in this Plan, if the effects on the environment are no more than minor. The rules for permitted activities contain conditions to ensure there will be no significant adverse effects. If, however, the activity does not comply with the conditions in the rule for permitted activities, a resource consent will be required because of the potential for significant adverse effects on the environment.

16.3.15 Discharges of PM₁₀

This section may also apply to any discretionary activity under Rules 16.3.1 to 16.3.14, as follows:

(1) Before 1 September 2013

- Rules 16.3.15.1 to 16.3.15.3 may apply if the relevant airshed already breaches the ambient air quality standard for PM₁₀, and the discharge is likely to increase significantly the concentration of PM₁₀ in the airshed;
- Rules 16.3.15.1 and 16.3.15.4 may apply if the relevant airshed does not breach the ambient air quality standard for PM_{10} .

(2) After 31 August 2013

• Rules 16.3.15.1 and 16.3.15 5 may apply.

16.3.15.1 Discharges of PM₁₀ – prohibited activity

Except as provided for by the permitted activity rules in this Plan, the discretionary activity rules under sections 16.3.1 to 16.3.14 where applicable, and Rules 16.3.15.2 to 16.3.15.5 where applicable, the discharge of PM_{10} to air is a *prohibited activity*, for which no consent will be granted.

16.3.15.2 Discharges of PM_{10} in an airshed before 1 September 2013, where the ambient air quality standard for PM_{10} is already breached – discretionary activity

Except as provided for by the permitted activity rules in this Plan or prohibited by Rules 16.3.1.1, 16.3.3.1, 16.3.12.1 and 16.3.15.1, the discharge of PM_{10} to air in an airshed before 1 September 2013, where:

- (1) The concentration of PM_{10} in the airshed already breaches its ambient air quality standard; and
- (2) The discharge to be permitted by the resource consent is likely to increase significantly the concentration of PM_{10} in an airshed;

is a *discretionary activity*, providing the discharge to be permitted by the resource consent is not likely to cause, at any time, the concentration of PM_{10} in an airshed to be above the straight line path or the curved line path for that airshed.

16.3.15.3 Discharges of PM_{10} in an airshed before 1 September 2013, if a resource consent application cannot be granted under Rule 16.3.15.2, and the applicant proposes offsets – discretionary activity

Except as provided for by the permitted activity rules in this Plan or prohibited by Rules 16.3.1.1, 16.3.3.1, 16.3.12.1 and 16.3.15.1, the discharge of PM_{10} to air in an airshed before 1 September 2013, if a resource consent application cannot be granted under Rule 16.3.15.2, and:

(1) If the concentration of PM_{10} in the airshed, at the time the application is decided, is on or below the straight line path or the curved line path;

is a *discretionary activity*, providing:

- (a) The applicant reduces the amount of PM_{10} discharged from another source into the same airshed where the direct effects of the discharge may be experienced; and
- (b) The reduction in discharges will be equal to or greater than the concentration of PM_{10} in the airshed above the straight line path or the curved line path caused by the discharge permitted by the resource consent; and
- (c) The reduction in discharges of PM_{10} will:
 - (i) Take effect within 1 year after the grant of the resource consent; and
 - (ii) Be effective for the duration of the resource consent; or
- (2) If the application has been made in circumstances to which Section 124 of the Resource Management Act 1991 applies, and the concentration of PM_{10} in the airshed, at the time the application is decided, is above the straight line path or the curved line path;

is a *discretionary activity*, providing:

(a) The applicant reduces the amount of PM_{10} discharged from another source into the same airshed where the direct effects of the discharge may be experienced; and

- (b) The reduction in discharges will be equal to or greater than the amount of the discharge permitted by the resource consent; and
- (c) The reduction in discharges of PM_{10} will:
 - (i) Take effect within 1 year after the grant of the resource consent; and
 - (ii) Be effective for the duration of the resource consent.

16.3.15.4 Discharges of PM_{10} in an airshed before 1 September 2013, where the ambient air quality standard for PM_{10} is not breached – discretionary activity

Except as provided for by the permitted activity rules in this Plan or prohibited by Rules 16.3.1.1, 16.3.3.1, 16.3.12.1 and 16.3.15.1, the discharge of PM_{10} to air in an airshed before 1 September 2013, where the concentration of PM_{10} in the airshed does not breach its ambient air quality standard;

is a *discretionary activity*, providing the discharge to be permitted by the resource consent is not likely, at any time, to cause an airshed to exceed the ambient air quality standard for PM_{10} .

16.3.15.5 Discharges of PM₁₀ after 31 August 2013 – discretionary activity

Except as provided for by the permitted activity rules in this Plan or prohibited by Rules 16.3.1.1, 16.3.3.1, 16.3.12.1 and 16.3.15.1, the discharge of PM_{10} to air in an airshed after 31 August 2013;

is a discretionary activity, providing:

- (a) The concentration of PM_{10} in the airshed does not breach its ambient air quality standard; or
- (b) The granting of the resource consent is not likely, at any time, to cause the concentration of PM_{10} in an airshed to breach its ambient air quality standard.

Principal reasons for adopting

Rules 16.3.15.1 to 16.3.15.5 are adopted to meet the requirements of regulations 17 to 19 of the NESAQ; to assist in ensuring public health impacts of PM_{10} air pollution are minimised; and to assist in achieving the curved or straight line paths to compliances in Air Zone 1 and specified airsheds in Air Zone 2, or the ambient air quality standard for PM_{10} , and ultimately the Otago Goal Level for PM_{10} .

The rule framework meets the requirements of the NESAQ for curved line paths, enables the use of offsets to mitigate the effects of a PM_{10} discharge in some circumstances, and prohibits the grant of resource consent in specific situations as required by the NESAQ. In exercising its discretion to enable the use of offsets at any time, the Council will give consideration to the interaction between the offset discharge and the proposed discharge.

16.4 Information requirements

16.4.1 Introduction

The Resource Management Act requires applications for resource consents to be made in accordance with Section 88. The Act further requires that, where an assessment of the effects of the proposed activity are required, this assessment be prepared in accordance with the Fourth Schedule of Act.

In general, applications for resource consent for activities affecting Otago's air resource will be required to demonstrate that:

- (a) The effects of the proposed activity comply with the relevant objectives, policies and rules of this Plan;
- (b) Information has been included, in accordance with the Fourth Schedule of the Resource Management Act, to enable the consent authority to make an assessment of the effects of the proposed activity; and
- (c) Where practicable, consultation has occurred with parties likely to be affected by the proposed activity.

Without limiting the requirements of Section 88 of the Resource Management Act, or of the Fourth Schedule to the Act, any application for any activity which this Plan specifies as being a discretionary activity will be required to supply information as specified in this part of the Plan:

Pursuant to Section 88(2) of the Act, no application shall be made for an activity that this Plan specifies as a prohibited activity once the time for making or lodging submissions or appeals against the proposed rule has expired and;

- (a) No such appeals or submissions have been lodged; or
- (b) All such submissions or appeals have been withdrawn or dismissed.

Applications for resource consents shall be made on the prescribed forms available from the Otago Regional Council.

16.4.2 General information requirements

The following information must be supplied with all resource consent applications:

- 1. The name of the applicant, and the name of the owner or occupier if different from that of the applicant.
- 2. The address of the applicant, owner or occupier.
- 3. A description of the proposed activity and its purpose.
- 4. The location of the proposed activity together with a site plan, legal description, construction plan (if applicable), and relevant map references.
- 5. A description of the consultation undertaken in relation to the application, and the outcomes of that consultation.
- 6. An assessment of the:

- Mass;
- Composition and concentrations of contaminants; and
- Frequency, rate and manner of the discharge, including the maximum ground level concentrations of significant contaminants in the discharge.
- 7. An assessment of any actual or potential effects that the activity may have on the environment, and the ways in which any adverse effects may be mitigated. This assessment shall be in such detail as corresponds with the scale and significance of the actual or potential effects that the activity may have on the environment, and shall be prepared in accordance with the Fourth Schedule of the Act.

In particular, the assessment of environmental effects shall focus on:

- (a) Any adverse effects on:
 - Human health and safety;
 - Amenity values;
 - Resources or values of significance to Kai Tahu;
 - Surface water or groundwater;
 - Soil, plants, animals and ecosystems; and
- (b) Any cumulative effects that may arise over time or in combination with other discharges; and
- (c) Any effects of low probability or high potential impact; and
- (d) When an application is made to continue a discharge of contaminants at a site, the assessment of effects will also include information on the standard of the past or present discharge, and its effects; and
- (e) Whether there is PM_{10} within the proposed discharge; and
- (f) Whether any offset proposed will provide adequate mitigation for the effect of the PM_{10} to be discharged; and
- (g) Any other adverse effects on ambient air quality in any airshed.
- 8. An assessment of:
 - Local meteorology;
 - Geographical features; and
 - Surrounding environmental conditions that may affect the frequency, duration, intensity and degree of environmental effects.
- 9. Any alternative methods of treating the discharge, or alternative locations which have been considered.
- 10. A description of the measures to be undertaken to help prevent or reduce any actual or potential effects.
- 11. Information on how equipment controlling the discharge will be operated and maintained and what contingency plans are in place in the event of equipment failure.
- 12. An assessment of whether any other resource consent is required from any other consent authority to undertake the proposed activity and whether any such consent has been applied for, or obtained.

- 13. Any proposed monitoring provisions.
- 14. Any additional information that may be required in relation to the application as specified in 16.4.3.
- 15. How the discharge to air will meet the requirements of the NESAQ.

16.4.3 Provision of further information

Pursuant to Section 92 of the Resource Management Act, a consent authority may, at any reasonable time before the hearing of a resource consent application, by written notice to an applicant for a resource consent, require the applicant to provide further information relating to the application. Due to the variable nature and site specific aspects of activities that discharge into air within Otago, it may be difficult to define all of the required information until closer investigation of a specific application has been conducted. The applicant may be required to supply further information, pursuant to Section 92, in cases where additional information is necessary to enable the Otago Regional Council to better understand the nature of the activity in respect of the application for which a resource consent is made, the effect it will have on the environment, or the ways in which any adverse effects may be mitigated.

16.4.4 Further information - modelling requirements

Applicants should consult the Otago Regional Council in the early stages of preparing an application to determine whether dispersion modelling is required. Modelling may be required for significant discharges.

Where modelling is required, the applicant will be required to model the ground level concentrations of any contaminant present, or likely to be present, in significant quantities in the discharge in accordance with any relevant standards or guidelines. Also, the following information will be submitted with a resource consent application:

- (a) A discussion of the model and how particular algorithms or settings were used;
- (b) The source emission data used in the modelling procedure;
- (c) A description of the contaminants in the discharge;
- (d) A description of the meteorological data input to the model;
- (e) A site description including building and local terrain;
- (f) Tables and graphical presentations of the predicted maximum ground level concentrations for each contaminant at regular and appropriate intervals from the discharge points;
- (g) A comparison of the predicted maximum ground level concentrations with the appropriate guideline or design levels;
- (h) A discussion of the results and conclusions with respect to both minor and significant adverse effects on the environment; and
- (i) A justification for any deviations from best practice modelling procedures.

PART IV RULES

Part V Methods Other Than Rules



17.1 Introduction

This part of the Plan outlines methods other than rules which will be used by the Otago Regional Council to aid in implementing the objectives and policies of this Plan.

17.2 Liaison with city and district councils

17.2.1 Land use planning

- 17.2.1.1 The Otago Regional Council will seek the inclusion of appropriate land use policy, rules and methods within district plans as necessary to further the objectives and policies contained in this Plan.
- 17.2.1.2 The Otago Regional Council will encourage Otago's city and district councils to control the adverse effects on air quality from land use activities and in particular those involving dust, agrichemical application or potentially odorous discharges through district plans, land use consents or education and information by:
 - (1) Achieving physical separation of incompatible land uses through buffer zones or shelter belts;
 - (2) Recognising existing use rights and reverse sensitivity; and
 - (3) Encouraging people undertaking land use activities to manage the effects of their activities through following codes of practice or environmental management systems where appropriate.

Principal reasons for adopting

Method 17.2.1.1 is adopted to promote integrated management of the effects of land use activities on air quality between the Otago Regional Council and Otago's city and district councils.

Method 17.2.1.2 recognises the importance of land use planning provisions in the management of adverse effects on air quality arising from land use activities involving agrichemical application, dust emissions or potentially odorous discharges. It indicates the Otago Regional Council's intention to support and promote the role that city and district councils can play, within their functions, to control any actual or potential effects on air quality arising from the use, development or protection of land. Furthermore the method recognises that such functions are not restricted to regulatory mechanisms and that there are a variety of non-regulatory approaches which can be used to raise public awareness of these issues.

The term "reverse sensitivity" generally refers to the development of a sensitive activity in an area where it may be adversely affected by activities that are lawfully pre-existing. The new, sensitive development then raises an expectation that those existing activities should be constrained for the benefit of the new one. Case law has established that reverse sensitivity can be recognised by land use planning mechanisms within district plans that regulate or control certain land uses because of their sensitivity to discharges of contaminants from other land uses.

17.2.2 Joint hearings

17.2.2.1 To provide for joint hearings where consents are required from both the Otago Regional Council and Otago's city and district councils.

Principal reasons for adopting

Method 17.2.2.1 is adopted because joint hearings over discharge permit applications to the Otago Regional Council and land use consents to territorial authorities provide for integrated management. Continued liaison and consultation will be needed to ensure the most efficient and effective management of the air resource of Otago.

17.2.3 Transportation planning

17.2.3.1 The Otago Regional Council will encourage city and district councils to adopt traffic management practices to manage the effects of motor vehicle emissions and the operation of the roading network on air quality.

Principal reasons for adopting

Method 17.2.3.1 recognises that city and district councils have responsibilities outside of land use planning, which are effective in managing the adverse effects of motor vehicle emissions and the operation of the roading network. The Otago Regional Council will encourage city and district councils to adopt management practices which stem from these responsibilities where such measures would help to achieve the objectives and policies of this Plan.

17.2.4 Domestic heating appliances

17.2.4.1 Otago Regional Council will liaise with Otago's city and district councils to develop mechanisms to enable Rules 16.3.1.2 to 16.3.1.6 concerning the installation of domestic heating appliances to be integrated with the city and district council's own building and land use consent processes.

Principal reasons for adopting

Method 17.2.4.1 is adopted to integrate the Resource Management Act and Building Act requirements for installing domestic heating appliances. Integration ensures that regional council and territorial authority requirements are considered at the same time. This process will reduce the time required for applicants to obtain approval and will ensure that the requirements of the Otago Regional Council and Otago's city and district councils are fully considered before new heating appliances are installed.

17.3 Liaison with other organisations

17.3.1 Liaison with government organisations

17.3.1.1 The Otago Regional Council will liaise with government organisations that control activities which discharge contaminants into air.

Principal reasons for adopting

Method 17.3.1.1 recognises that government organisations have responsibilities for managing adverse effects of discharges into air. For example, the Ministry of Health has responsibilities for adverse effects on the health of people, and the Ministry of Agriculture has responsibilities for agrichemical spray drift. In order to achieve integrated management it is important that there is liaison between the Otago Regional Council and other agencies with responsibilities for managing the effects of discharges into air.

17.4 Environmental education and promotion

17.4.1 Community understanding of air quality

17.4.1.1 The Otago Regional Council will use environmental education to help the community and industry understand the types of effects that can occur as a result of discharges of contaminants into air and the overall effects of such discharges on ambient air quality.

Principal reasons for adopting

Method 17.4.1.1 is adopted to ensure that the adverse effects associated with discharges of contaminants into air are well understood by both the community and industry. It recognises that awareness about effects can lead to people adopting practices which can bring about changes in the quality of the air resource, and that environmental education can be an effective alternative to enforcement as a means of changing people's behaviour.

17.4.2 Domestic and outdoor burning discharges

- 17.4.2.1 To encourage and support the use of practices which assist in reducing the adverse effects associated with discharges from domestic heating appliances. Such practices include, but are not limited to:
 - (a) Following manufacturers' instructions and good burning practices for the type of fuel and the appliance being used;
 - (b) Avoiding the burning of waste or materials that are likely to result in hazardous air contaminants in domestic heating appliances;
 - (c) Burning wood only when it is dry;
 - (d) Burning other fuels such as good quality coal, wood pellets, gas or oil;
 - (e) Replacing existing domestic heating appliances that do not comply with the particulate emission rates set in the permitted activity rules for discharges from domestic heating appliances;
 - (f) Finding an alternative to the use of any domestic heating appliance during conditions that are likely to result in high PM_{10}

concentrations, unless electricity or gas or oil supply has failed; and

(g) Installing domestic heating appliances that are efficient, minimise operator control and are designed to produce low particulate emissions both in laboratory testing and real life situations.

Education and promotion programmes will be among the methods the Council will employ to ensure good practices are followed.

- 17.4.2.2 To use education and promotion to encourage practices which assist in reducing the adverse effects associated with outdoor burning. Such practices include, but are not limited to:
 - (a) Ensuring that the burning method being used is appropriate to the type of material being burnt;
 - (b) Adopting good burning practices to reduce the volume of smoke discharged;
 - (c) Utilising alternative methods of disposal wherever practicable; and
 - (d) Promoting voluntary codes of practice for the supply, sale and use of solid fuel.
- 17.4.2.3 To support and promote the management of new areas of housing to avoid adverse effects of discharges to air from solid fuel heating and outdoor burning.

Principal reasons for adopting

Method 17.4.2.1 is adopted to raise community and industry awareness of practices that can be implemented to reduce the effects that discharges from domestic heating can have on both ambient air quality and the surrounding local environment.

Method 17.4.2.2 is adopted to raise community awareness of practices that can reduce the effects of discharges from outdoor burning on the surrounding local environment.

Method 17.4.2.3 is adopted to assist in avoiding adverse effects on local air quality or on any airshed. In some new subdivision developments in Otago, developers have set additional limits on discharges to air from solid fuel heating in order to ensure good quality air and high amenity values. Such voluntarily measures are supported as they contribute towards achieving higher quality ambient air.

17.5 Advocacy and information transfer

17.5.1 Methods to avoid, remedy or mitigate adverse effects

- 17.5.1.1 To provide information and advice on appropriate methods of avoiding, remedying or mitigating any adverse effects of discharges of contaminants into air.
- 17.5.1.2 To provide information and advice, as appropriate, to roading and earthworks contractors, roading authorities, pastoral farmers, horticulturists and other persons whose activities may generate dust or smoke, on ways of avoiding or minimising the discharge of dust or smoke to air.

Principal reasons for adopting

Methods 17.5.1.1 and 17.5.1.2 recognise that the provision of information and advice provides a further means of avoiding or mitigating adverse effects of discharges of contaminants into air, particularly when used in conjunction with regional rules.

Many of the problems associated with dust from area sources occur as a consequence of land management practices that leave soils exposed to wind. Similarly, problems associated with vegetation burning often relate to when and how burning is carried out. The Council considers that the adoption of sustainable land management practices is a key method in avoiding or mitigating adverse effects from dust and smoke and this method is directed at providing information and advice to assist in this process.

17.5.2 Codes of practice and self-regulation

- 17.5.2.1 Encourage and assist the development and use of industry codes of practice and environmental management systems that seek to achieve or implement the objectives or policies in this Plan.
- 17.5.2.2 Liaise with industry and other relevant organisations to ensure they have sufficient information and understanding of this Plan to develop appropriate means of self-regulation.

Principal reasons for adopting

Methods 17.5.2.1 and 17.5.2.2 cover a range of actions the Otago Regional Council will undertake to enable and support industrial groups and individuals to develop and implement voluntary actions that demonstrate the best practice in relation to that activity or effect.

Method 17.5.2.1 recognises that adherence to codes of practice, environmental management systems and other voluntary actions can be particularly effective in controlling the effects of small scale activities that produce minor adverse effects. For example, Schedule 4 of this Plan, which is based on the *Code of Practice for the Management of Agrichemicals* (NZS 8409:1999), recommends practices which may assist in achieving agrichemical discharge rule conditions. Where

adverse effects do occur, the Otago Regional Council has a number of regulatory actions it can follow.

Method 17.5.2.2 recognises that the Otago Regional Council can assist in providing information to industries so that they can develop voluntary actions which are consistent with the objectives and policies contained within this Plan.

17.5.3 Central government initiatives

- 17.5.3.1 Advocate the need for more comprehensive nation-wide initiatives to reduce the:
 - (a) Discharge of contaminants into air from motor vehicles; and
 - (b) Release of greenhouse gases and ozone layer depleting substances into the atmosphere.
- 17.5.3.2 Support national programmes that provide funds for the implementation of clean heating within areas of highest need, especially Otago's Air Zone 1.

Principal reasons for adopting

Methods 17.5.3.1 and 17.5.3.2 recognise that there is a role for the Otago Regional Council to continue advocating for continuing attention to these issues at a national level.

17.5.4 Testing procedures for domestic heating appliances

- 17.5.4.1 In determining whether a given model of a domestic heating appliance meets the particulate emission rates and thermal efficiency standards set in the permitted activity rules for discharges from domestic heating appliances, the Council will require evidence that an appliance has been tested according to the relevant testing procedures specified below:
 - (i) AS/NZS 4013:1999, Domestic solid fuel burning appliances Method for determination of flue gas emissions;
 - (ii) AS/NZS 4012:1999, Domestic solid fuel burning appliances Method for determination of power output and efficiency;
 - (iii) AS/NZS 5078:2007, Domestic solid fuel burning appliances Pellet heaters – Method for determination of power output and efficiency;
 - (iv) AS/NZS 4886:2007, Domestic solid fuel burning appliances Pellet heaters –Determination of flue gas emissions;

A domestic heating appliance will be considered to meet the emission standard for a particular fuel or mix of fuels, only when the particulate emissions meet the rates set in the permitted activity rules for discharges from domestic heating appliances when tested with each of those fuels, or a specified mixture of fuels. If the emission standard is exceeded when using a particular fuel or mix, the appliance is not approved under the permitted activity rules for discharges from domestic heating appliances for use with that fuel or mix of fuels. The Council is supportive in principle of innovation and new technologies for achieving cleaner discharges from solid fuel heating, and will initiate a change to this Plan when any new testing procedure for domestic heating appliances, suitable for use in the Otago region, is recognised by the Ministry for the Environment and is provided for by the NESAQ.

The Otago Regional Council will hold a publicly-available list of those domestic heating appliance models that meet the emission standards set in the permitted activity rules for discharges from domestic heating appliances.

Principal reasons for adopting

Method 17.5.4.1 is adopted to set the procedures used to determine compliance with permitted activity rules for discharges from domestic heating appliances.

Part VI Anticipated Environmental Results



18.1 Introduction

This part of the Plan outlines the environmental results anticipated from the implementation of the policies and methods (including rules) in this Plan.

18.2 Anticipated environmental results

- **18.2.1** There are no significant increases in the ambient levels of contaminants identified in Schedule 1.
- **18.2.2** Levels of contaminants in Otago's ambient air do not exceed the standards in the NESAQ, and where possible the Otago Goal Levels identified in Schedule 1.
- **18.2.3** Ambient air quality improves.
- **18.2.4** Measurable improvements occur in the levels of PM_{10} measured in Air Zones 1 and 2, particularly during the months of May to August.
- **18.2.5** Less waste is disposed of by combustion (in particular that waste which discharges hazardous air contaminants when burnt).
- **18.2.6** There is increased adoption of emission control technology and production processes that produce lower mass emissions and discharge low levels of hazardous air contaminants.
- **18.2.7** There is increased public awareness of the effects of contaminant discharges on localised and ambient air quality.

Part VII Cross-Boundary Issues



19.1 Introduction

An activity which discharges into Otago's air resource can create adverse effects beyond the immediate vicinity of the activity. Due to the mobile nature of the air resource the effects of a discharge into air may occur across district and regional boundaries.

It is important that the cross-boundary issues are identified, agreed to and are dealt with in an efficient and effective manner in order that any adverse effects of discharges are avoided, remedied or mitigated.

Where the adverse effects of an activity occur in an area under the management of another agency, administrative processes are required to ensure that the cross-boundary nature of the effect is considered, and where necessary taken into account by the agency responsible for the management of that resource. Establishing processes between local authorities in order to deal with those cross-boundary issues is required.

19.2 Methods

In order to deal with cross-boundary issues as they arise, the Otago Regional Council will use the following methods:

- 19.2.1 To liaise with adjacent regional councils over issues of concern related to the management of the air resource.
- 19.2.2 To promote and encourage the development of protocols with Otago's city and district councils and adjacent regional councils for resolving cross-boundary issues.
- 19.2.3 To consult with all agencies having responsibilities for the sustainable management of aspects of Otago's environment.
- 19.2.4 To promote and encourage joint working groups, joint council committees and other joint approaches between appropriate city and district councils and the Regional Council to consider cross-boundary issues.
- 19.2.5 To combine with appropriate city and district councils and regional councils in jointly processing resource consent applications that cross administrative boundaries.
- 19.2.6 To consider transferring functions to city and district councils under Section 33 of the Resource Management Act where it is more efficient, effective and appropriate to manage the air resource.

Explanation and principal reasons for adopting

Processes to resolve cross-boundary issues will be based on consultation and communication between Otago's city and district councils, adjacent regional councils and central government. Various approaches employing joint groups, committees or other means can be used to facilitate the consideration and decision-making between different authorities over issues that cross their boundaries.

Part VIII Monitoring and Review



20.1 Introduction

This part of the Plan outlines the procedures to be used to:

- Monitor the effectiveness of the Plan as a means of achieving its objectives and policies; and
- Review the matters contained within this Plan.

In terms of this Plan, Section 35(2) of the Act places a duty on the Otago Regional Council to monitor:

- The state of the regional environment to the extent that is necessary to carry out the Council's air management functions (baseline monitoring or environmental monitoring);
- The suitability and effectiveness of this Plan and the exercise of any functions, powers or duties delegated or transferred by the Otago Regional Council (process monitoring); and
- The exercise of resource consents for discharges to air (compliance monitoring).

NESAQ Regulation 15 requires the Otago Regional Council to monitor ambient air quality in airsheds where it is likely that the standard for any contaminant will be breached, and Schedule 2 to the NESAQ outlines the method for such monitoring.

Section 67(1)(i) of the Act requires the Otago Regional Council to include within this Plan, the procedures to be used to review the matters contained within it, and to monitor the effectiveness of the Plan as a means of achieving its objectives and policies.

20.2 Elements to be monitored

Monitoring of the suitability and effectiveness of the objectives and policies within this Plan will be carried out in conjunction with monitoring of the Regional Policy Statement for Otago, other regional plans and the NESAQ. It will also be subject to the provisions of the Council's own Annual Plan and Long Term Council Community Plan (LTCCP).

In considering the elements requiring monitoring, the Otago Regional Council will have particular regard to the anticipated environmental results as stated within this Plan and will monitor the following:

- 1. The number and nature of discharges within Otago.
- 2. The type and mass of contaminants discharged into air and the effects of those contaminants on the receiving environment.
- 3. The effects of discharges on ambient air quality, and in particular the levels and effects of the following principal indicators:
 - PM₁₀;
 - Sulphur dioxide;
 - Carbon monoxide; and
 - Nitrogen dioxide.
- 4. Existing ambient air quality in areas most likely to be affected by air pollution activities, to assess whether the air quality remains within the Ambient Air Quality Guidelines contained in Schedule 1, and meets the standards set by the NESAQ, including paths to compliance with the ambient air quality standard for PM₁₀.
- 5. The quality of significant discharges to air from industrial or trade premises and other significant discharges will be monitored, as will the air quality near to the discharges.

20.3 Monitoring techniques

In monitoring the elements detailed in Section 20.2, the following techniques may be used:

- 1. Analysis of feed back and compliments and complaints received.
- 2. Maintenance of a regional ambient air quality monitoring programme, and expansion as necessary.
- 3. Recording and analysis of unauthorised discharges to air.
- 4. Development of an emissions inventory to identify the source, scale and distribution of discharges of contaminants into air.
- 5. Requiring self monitoring of consents, where necessary, and the provision of the collected information to the Otago Regional Council.
- 6. Compliance audit monitoring to ensure the conditions on resource consents are being adhered to.
- 7. Maintaining a database of air discharge permits issued.
- 8. Records of the number of consent applications made for each activity regulated by the Plan.
- 9. Commission research, as necessary, to provide additional information on the region's air resource.
- 10. Where appropriate, develop and implement joint initiatives with local authorities, government departments and other agencies to monitor key aspects of Otago's air resource.
- 11. Make available data held by the Otago Regional Council and seek the transfer of information between agencies and local authorities about Otago's air resource.
- 12. Consultation with key stakeholders and affected parties.

20.4 Plan review

The Resource Management Act requires that this Plan be reviewed no later than 10 years from the date upon which it becomes operative.

In the interim, the Council may determine to review part of, or the entire Plan, in response to:

- 1. Information from monitoring which indicates the need for a review or change;
- 2. The identification of any significant new air quality issues in the region;
- 3. Changes in national policy including new or amended laws, regulations, national policy statements or any other actions taken which require a national response;
- 4. Changes made to the Regional Policy Statement for Otago and to other regional plans by the Otago Regional Council; or
- 5. Requests for a plan change or review made by any person in accordance with Part II of the First Schedule of the Resource Management Act.

PART VIII MONITORING AND REVIEW
Schedules



Indicator	Otago Goal Levels*	MfE Levels [†]	Averaging Times**	Preferred Techniques for Measurement***
Particulates (PM ₁₀)	35 µg/m ³	$50 \ \mu g/m^3$	24 hours	US 40 CFR Part 50,
		$20 \ \mu g/m^3$	annual	Appendix J
Sulphur dioxide	$330 \ \mu g/m^3$	$500 \ \mu g/m^3$	10 minutes	AS 3580.4.1 - 1990
	$230 \ \mu g/m^3$	350 µg/m ³ ^{††}	1 hour	
	$380 \ \mu g/m^3$	570 µg/m ³ ^{††}	1 hour	
	80 μg/m ³	$125 \ \mu g/m^{3}$	24 hours	
		$50 \ \mu g/m^3$	annual	
Carbon monoxide	20 mg/m^3	30 mg/m^3	1 hour	AS3580.7.1 – 1992
	6 mg/m^3	10 mg/m^3	8 hours	
Ozone		$150 \ \mu g/m^{3}$	1 hour	AS 3580.6.1 - 1990
	-	$100 \ \mu g/m^3$	8 hours	
Nitrogen dioxide	$130 \ \mu g/m^3$	$200 \ \mu g/m^3$	1 hour	AS 3580.5.1 - 1993
	$60\mu\text{g/m}^3$	$100 \mu g/m^3$	24 hours	
Lead		$0.2 \ \mu g/m^3$	3 months	AS 2800 - 1985
Fluoride				
- Special land use		$1.8 \mu g/m^3$	12 hours	AS 3580.13.1-1993
		$1.5 \mu g/m^3$	24 hours	AS 3580.13.2-1991
		$0.8 \ \mu g/m^3$	7 days	
		$0.4 \ \mu g/m^3$	30 days	
		$0.25 \ \mu g/m^3$	90 days	
		$3.7 \ \mu g/m^3$	12 hours	AS 3580.13.1-1993
- General land use		$2.9 \ \mu g/m^3$	24 hours	AS 3580.13.2-1991
		$1.7 \mu g/m^3$	7 days	
		$0.84 \ \mu g/m^3$	30 days	
		$0.5 \ \mu g/m^3$	90 days	
		0.5 µg/m	<i>50 days</i>	
Concernation area		0.1 µg/m ³	90 days	
- Conservation area Hydrogen sulphide		7 μg/m ³	1-hour	AS3580.4.1 – 1990,
riyurogen suipinde		/ μg/m	1-110UI	A35380.4.1 - 1990, coupled with a
				hydrogen sulphide to
				sulphur dioxide
				converter

Schedule 1 Regional Ambient Air Quality Guidelines

 $\mu g/m^3$ = micrograms per cubic metre

 mg/m^3 = milligrams per cubic metre

AS = Australian Standard

* Levels do not always equate to 66% of alert levels as a consequence of rounding.

** Averaging times are the times over which the average level of indicator should not exceed the levels given in the guidelines.

- *** Other international standard methods may be used as appropriate.
- [†] MfE (2002) updated values, except for Fluoride, where the 1994 values are still valid.
- ^{††} Refer to permissible excess in NESAQ ambient air quality standards (below).

The **Otago Goal Levels** equate to 66% of the level set by the Ministry for the Environment in its publication "*Ambient Air Quality Guidelines*" (1994 and 2002).

The 66% level has been adopted to reflect the "alert" levels being advocated in the Ministry for the Environment's Environmental Performance Indicators Programme (*Environmental Performance Indicators: Proposals for Air, Fresh Water and Land, 1997*). Concentrations above these levels can be of concern and can lead to the AAQG, and therefore the NESAQ, being exceeded if trends are not curbed. The levels were chosen because it is generally considered that air quality in Otago is high, except for PM_{10} in some areas, and that it could not be maintained or enhanced (Objective 6.1.1) if the guideline levels were adopted on their own.

The **Ministry for the Environment Levels** are those specified in the Ministry for the Environment's AAQG and are defined as the levels adequate to protect the health of the general population, although the levels set for fluoride are set to protect plants and animals which have lower tolerances than humans. The level set for hydrogen sulphide is set to avoid the occurrence of odour problems. Contaminant concentrations above the Ministry for the Environment levels are considered in the "action" category as defined by the Ministry for the Environment publication *Environmental Performance Indicators: Proposals for Air, Fresh Water and Land (1997)*. These levels are considered unacceptable by national and international standards and public health or other effects are likely. Concentrations which exceed these levels must be urgently reduced.

These Schedule 1 guidelines are designed to protect the ambient air quality in any area of Otago. Any monitoring to test compliance with the guidelines should be undertaken so the measurements are representative of the area of concern. In this respect it is important that siting of the monitoring station follows the requirements of the specific methods listed in the table. The guidelines are not to be used as design concentrations for dispersion modelling of individual point sources.

The NESAQ (NESAQ Schedule 1) ambient air quality standards are statutory standards and are mandatory nationally. They are listed below:

Contaminant	Threshold Concentration	Permissible Excess	Monitoring Method
Carbon monoxide	10 mg/m ³ expressed as a running 8-hour mean	One 8-hour period in a 12-month period	AS 3580.7.1:1992
Nitrogen dioxide	200 μ g/m ³ expressed as a 1-hour mean	9-hours in a 12- month period	AS 3580.5.1:1993
Ozone	150 μ g/m ³ expressed as a 24-hour mean	Not to be exceeded at any time	AS 3580.6.1:1990
PM ₁₀	50 μ g/m ³ expressed as a 24-hour mean	One 24-hour period in a 12-month period	AS/NZS 3580.9.6:2003 US Code of Federal Regulations, Title 40 – Protection of Environment, Vol 2 Pt 50, Appendix J
Sulphur dioxide	$350 \ \mu g/m^3$ expressed as a 1-hour mean	9-hours in a 12- month period	AS 3580.4.1:1990
	570 μ g/m ³ expressed as a 1-hour mean	Not to be exceeded at any time	

NESAQ ambient air quality standards

In the above table,

1-hour mean -

- (a) means a mean calculated every hour on the hour for the preceding hour; and
- (b) in relation to a contaminant at a particular location for a particular hour, means the mean of not more than 10-minute means, collected not less than once every 10 seconds, for the contaminant at that location during that hour

24-hour mean -

- (a) means a mean calculated every 24 hours at midnight for the preceding 24 hours; and
- (b) in relation to a contaminant at a particular location for a particular 24-hour period, means--
 - (i) the mean level at which the contaminant is recorded in the air, by continuous sampling of the air at that location, throughout that 24-hour period; or
 - (ii) the mean of the 1-hour means for that contaminant at that location for the preceding 24 hours

running 8-hour mean -

- (a) means a mean calculated every hour on the hour for that hour and the preceding 7 hours to give 1 running 8-hour mean per hour; and
- (b) in relation to a contaminant at a particular location for a particular hour, means the mean of the 1-hour means for that contaminant at that location for that hour and the preceding 7 hours.

Schedule 2 Maps Showing Air Zone Boundaries



Overview of Air Zones





Air Zone 1 (Cromwell)



Air Zone 1 (Clyde)



Air Zone 1 (Alexandra)





Air Zone 2 (Wanaka)









Air Zone 2 (Naseby)



Air Zone 2 (Ranfurly)



Air Zone 2 (Roxburgh)



$S\,\text{C}\,\text{H}\,\text{E}\,\text{D}\,\text{U}\,\text{L}\,\text{E}\,2$



Regional Plan: Air for Otago (Updated to 1 January 2009) 119





Air Zone 2 (Palmerston)



Air Zone 2 (Waikouaiti)





Air Zone 2 (Port Chalmers)

Map 2 of 2

















Air Zone 2 (Green Island)

Map 1 of 2



Air Zone 2 (Green Island)

Map 2 of 2



Air Zone 2 (Mosgiel)

Map 1 of 2



Air Zone 2 (Mosgiel)

Map 2 of 2



Air Zone 2 (Milton)


SCHEDULE 2

SCHEDULE 2

Schedule 3 Hazardous Air Contaminants

The Otago Regional Council is particularly concerned about the effects of hazardous air contaminants on human health and the environment. As such, particular regard is to be given to avoiding the adverse effects of these contaminants when preparing or making decisions on resource consent applications in terms of Policy 8.2.2 of this Plan.

The table below lists the hazardous air contaminants described by the Ministry for the Environment in their publication *Ambient Air Quality Guidelines* (1994) as either being known or suspected to cause:

- Acute human health effects, cancer or teratogenic effects, or serious or irreversible effects, reproductive dysfunction, neurological disorders, heritable genetic mutations, or other chronic health effects; or
- Significant adverse effects on the environment due to their toxicity, persistence in the environment, tendency to bioaccumulate, or any combination of these.

The table also indicates those contaminants which the United States Environmental Protection Authority (1992) identify in their *Integrated Risk Information System* (IRIS) database either being a confirmed or suspected carcinogen. The classification system is described in the notes which follow the table.

Number	Contaminant	Carcinogenic Category
75-07-0	Acetaldehyde	B2
60-35-5	Acetamide	
75-05-8	Acetonitrile	
98-86-2	Acetophenone	D
53-96-3	2-Acetylaminofluorene	
107-02-8	Acrolein	С
79-06-1	Acrylamide	B2
79-10-7	Acrylic acid	
107-13-1	Acrylonitrile	B1
107-05-1	Allyl chloride	
92-67-1	4-Aminobiphenyl	
62-53-3	Aniline	B2
90-04-0	o-Anisidine	
1332-21-4	Asbestos	А
71-43-2	Benzene	А
92-87-5	Benzidine	А
98-07-7	Benzotrichloride	B2
100-44-7	Benzyl chloride	B2
92-52-4	Biphenyl	D
117-81-7	Bis(2-ethylhexyl)phthalate (DEHP)	B2
542-88-1	Bis(chloromethyl) ether	А
75-25-2	Bromoform	B2
106-99-0	1,3-Butadiene	B2
156-62-7	Calcium cyanamide	
105-60-2	Caprolactam	

Number	Contaminant	Carcinogenic Category
133-06-2	Captan	
63-25-2	Carbaryl	
75-15-0	Carbon disulfide	
56-23-5	Carbon tetrachloride	B2
463-58-1	Carbonyl sulfide	
120-80-9	Catechol	
133-90-4	Chloramben	
57-74-9	Chlordane	B2
7782-50-5	Chlorine	
79-11-8	Chloroacetic acid	
532-27-4	2-Chloroacetophenone	
108-90-7	Chlorobenzene	D
510-15-6	Chlorobenzilate	
67-66-3	Chloroform	B2
107-30-2	Chloromethyl methyl ether	А
126-99-8	Chloroprene	
1319-77-3	Cresol / cresylic acid (mixed isomers)	С
95-48-7	o-Cresol	С
108-39-4	m-Cresol	С
106-44-5	p-Cresol	С
98-82-8	Cumene	
	2,4-D (2,4-Dichlorophenoxyacetic acid) (including salts and esters)	
72-55-9	DDE (1,1-dichloro-2,2-bis (p-	B2

$S\,{\tt C\,H\,E\,D\,U\,L\,E}\quad 3$

Number	Contaminant	Carcinogenic Category
	chlorophenyl) ethylene)	
334-88-3	Diazomethane	
132-64-9	Dibenzofuran	D
96-12-8	1,2-Dibromo-3- chloropropane	
84-74-2	Dibutyl phthalate	D
106-46-7	1,4-Dichlorobenzene	
91-94-1	3,3 ¹ -Dichlorobenzidine	B2
111-44-4	Dichlorethyl ether (bis[2- chloroethyl]ether)	B2
542-75-6	1,3-Dichloropropene	
62-73-7	Dichlorvos	B2
111-42-2	Diethanolamine	
64-67-5	Diethyl sulfate	
119-90-5	3,3 ¹ -Dimethoxybenzidine	
60-11-7	4-Dimethylaminoazobenzene	
121-69-7	N,N-Dimethylaniline	
119-93-7	3,3 ¹ -Dimethylbenzidine	
79-44-7	Dimethylcarbomoyl chloride	
68-12-2	N,N-Dimethylformamide	
57-14-7	1,1-Dimethylhydrazine	
131-11-3	Dimethyl phthalate	D
77-78-1	Dimethyl sulfate 4,6-Dinitro-o-cresol (including salts)	B2
51-28-5	2,4-Dinitrophenol	
121-14-2	2,4-Dinitrotoluene	B2
123-91-1	1,4-Dioxane (1,4- Diethyleneoxide)	B2
122-66-7	1,2-Diphenylhydrazine	B2
106-89-8	Epichlorohydrin (1-Chloro- 2,3-epoxypropane)	B2
106-88-7	1,2-Epoxybutane	
140-88-5	Ethyl acrylate	
100-41-4	Ethylbenzene	D
51-79-6	Ethyl carbamate (Urethane)	
75-00-3	Ethyl chloride (Chloroethane)	
106-93-4	Ethylene dibromide (Dibromoethane)	B2
107-06-2	Ethylene dichloride (1,2- Dichloroethane)	B2
107-21-1	Ethylene glycol	
151-56-4	Ethyleneimine (Aziridine)	
75-21-8	Ethylene oxide	
96-45-7	Ethylene thiourea	
75-34-3	Ethylidene dichloride (1,1	С

Number	Contaminant	Carcinogenic Category
	Dichloroethane)	
50-00-0	Formaldehyde	B1
76-44-8	Heptachlor	B2
118-74-1	Hexachlorobenzene	B2
87-68-3	Hexachlorobutadiene 1,2,3,4,5,6- Hexachlorocyclohexane (all stereo isomers, including lindane)	С
77-47-4	Hexachlorocyclopentadiene	D
67-72-1	Hexachloroethane	
822-06-0	Hexamethylene di- isocyanate	
680-31-9	Hexamethylphosphoramide	
110-54-3	Hexane	
302-01-2	Hydrazine	B2
7647-01-0	Hydrochloric acid (Hydrogen chloride [gas only])	
7664-39-3	Hydrogen fluoride (Hydrofluoric acid)	
123-31-9	Hydroquinone	
78-59-1	Isophorone	С
108-31-6	Maleic anhydride	
67-56-1	Methanol	
72-43-5	Methoxychlor D	
74-83-9	Methyl bromide (Bromomethane)	D
74-87-3	Methyl chloride (Chloromethane)	
71-55-6	Methyl chloroform (1,1,1- Trichloroethane)	D
78-93-3	Methyl ethyl ketone (2- Butanone)	D
60-34-4	Methylhydrazine	
74-88-4	Methyl iodide (Iodomethane)	
108-10-1	Methyl isobutyl ketone (Hexone)	
624-83-9	Methyl isocyanate	
80-62-6	Methyl methacrylate	
1634-04-4	Methyl tert-butyl ether	
101-14-4	4,4 ¹ -Methylenebis (2- chloroaniline)	
75-09-2	Methylene chloride (Dichloromethane)	
101-68-8	4,4 ¹ -Methylenediphenyl diisocyanate (MDI)	
101-77-9	4,4 ¹ -Methylenedianiline	
91-20-3	Naphthalene	D
98-95-3	Nitrobenzene	D

Number	Contaminant	Carcinogenic Category
92-93-5	4-Nitrobiphenyl	
100-02-7	4-Nitrophenol	
79-46-9	2-Nitropropane	
684-93-5	N-Nitroso-N-methylurea	
62-75-9	N-Nitrosodimethylamine	B2
59-89-2	N-Nitrosomorpholine	
56-38-2	Parathion	С
82-68-8	Pentachloronitrobenzene (Quintobenzene)	
87-86-5	Pentachlorophenol	B2
108-95-2	Phenol	D
106-50-3	p-Phenylenediamine	
75-44-5	Phosgene	
7803-51-2	Phosphine	D
7723-14-0	Phosphorus	D
85-44-9	Phthalic anhydride	
1336-36-3	Polychlorinated biphenyls (Aroclors)	B2
1120-71-4	1,3-Propane sultone	
57-57-8	beta-Propiolactone	
123-38-6	Propionaldehyde	
114-26-1	Propoxur (Baygon)	
78-87-5	Propylene dichloride (1,2- Dichloropropane)	
75-56-9	Propylene oxide	B2
75-55-8	1,2-Propylenimine (2- Methylaziridine)	
91-22-5	Quinoline	
106-51-4	Quinone (p-Benzoquinone)	
100-42-5	Styrene	
96-09-3	Styrene oxide	

Number	Contaminant	Carcinogenic Category
1746-01-6	2,3,7,8-Tetrachlorodibenzo- p-dioxin	
79-34-5	1,1,2,2-Tetrachloroethane	
127-18-4	Tetrachloroethylene (Perchloroethylene)	
7550-45-0	Titanium tetrachloride	
108-88-3	Toluene	D
95-80-7	Toluene-2,4-diamine	
584-84-9	2,4-Toluene di-isocyanate	
95-53-4	o-Toluidine	
8001-35-2	Toxaphene (chlorinated camphene)	B2
120-82-1	1,2,4-Trichlorobenzene	D
79-00-5	1,1,2-Trichloroethane	С
79-01-6	Trichloroethylene	
95-95-4	2,4,5-Trichlorophenol	
88-06-2	2,4,6-Triclorophenol	B2
121-44-8	Triethylamine	
1582-09-8	Trifluralin	С
540-84-1	2,2,4-Trimethylpentane	
108-05-4	Vinyl acetate	
593-60-2	Vinyl bromide	
75-01-4	Vinyl chloride	
75-35-4	Vinylidene chloride (1,1- Dichloroethylene)	С
1330-20-7	Xylene (mixed isomers)	D
95-47-6	o-Xylene	
108-38-3	m-Xylene	
106-42-3	p-Xylene	

Number	Contaminant	Carcinogenic Category
	Antimony Compounds	
	Arsenic Compounds (inorganic including arsine)	А
	Beryllium Compounds	B2
	Cadmium Compounds	B1
	Chromium Compounds	А
	Cobalt Compounds	
	Coke Oven Emissions	А
	Cyanide Compounds ¹	
	Glycol ethers ²	
	Lead Compounds	B2

Number	Contaminant	Carcinogenic Category
	Manganese Compounds	D
	Mercury Compounds	D
	Fine mineral fibres ³	
	Nickel Compounds	А
	Polycyclic Organic Matter ⁴	
	Radionuclides (including radon) ⁵	А
	Selenium Compounds	B2

GENERAL NOTES:

- 1. The numbers referred to are the Chemical Abstracts Service Number. This service provides a source of additional information on the substance.
- 2. The classification system used by the United States Environmental Protection Agency in terms of confirmed or suspected carcinogens is as follows:

Carcinogenic Category	Explanation
A	Human carcinogen. Sufficient evidence to support causal association
	between contaminant and cancer.
В	Probable human carcinogen.
B1	Limited evidence for carcinogenity in humans.
B2	Sufficient evidence in animals but lacking adequate data on humans.
С	Possible human carcinogen. Limited evidence of carcinogenity in animals
	and an absence of data on humans.
D	Not classified. Evidence for carcinogenity in animals is inadequate.

TECHNICAL NOTES:

For all listings above which contain the word "Compounds" and for glycol ethers, the following applies: unless otherwise specified, these listings are defined as including any unique chemical substance that contains the named chemical (ie, antimony, arsenic, etc) as part of that chemical's infrastructure.

¹ X'CN where X' = H' or any other group where a formal dissociation may occur. For example, KCN or $Ca(CN)_{2}$

 2 R-(OCH₂CH₂)_n-OR' where:

n = 1, 2 or 3

R = alkyl C7 or less, or phenyl, or alkyl-substituted phenyl

R' = H, or alkyl C7 or less, or ester, sulphate, phosphate, nitrate or sulphonate.

³ Includes mineral fibre emissions from facilities manufacturing or processing glass, rock, or slag fibres (or other mineral derived fibres) of an average diameter of 1 micrometer or less.

⁴ Includes substituted and/or unsubstituted polycyclic aromatic hydrocarbons and aromatic heterocyclic compounds, with two or more fused rings, at least one of which is benzenoid (ie, containing six carbon atoms and is aromatic) in structure. Polycyclic Organic Matter is a mixture of organic compounds containing one or more of these polycyclic aromatic chemicals. Polycyclic Organic Matter is generally formed or emitted during thermal processes including:

- (1) Incomplete combustion;
- (2) Pyrolysis;
- (3) The volatilisation, distillation or processing of fossil fuels or bitumens; or
- (4) The distillation or thermal processing of non-fossil fuels.

⁵ A type of atom which spontaneously undergoes radioactive decay.

Schedule 4 Good Management Practices for Agrichemical Application

This schedule is based on the information contained in the New Zealand Standard 8409:1999: *Code of Practice for the Management of Agrichemicals*, August 1999, developed by the New Zealand Agrichemical Education Trust. It has been included in the Plan in a simple and convenient form for general public information and education purposes. The information contained in this Schedule also provides general guidance on the best practice for avoiding or minimising adverse effects on the environment from agrichemical application. This information may be useful for those carrying out discharges of agrichemicals into air under rules in 16.3.9, but it does not negate the conditions in any such rule.

This information caters for a wide range of purposes for spraying. Some, not all, will be applicable depending on the purpose and scale of spraying.

For further, more detailed information reference should be made to the Code of Practice itself. This document can be obtained from Standards New Zealand (Private Bag 2439, Wellington, Ph (04) 498 5990 or Fax (04) 498 5990), e-mail: snz@standards.co.nz, website: www.standards.co.nz, or can be viewed at the Otago Regional Council's Dunedin Office.

To achieve the Plan's objectives, any person discharging agrichemical sprays is requested to observe the following, in conjunction with the accompanying chart. All such persons should:

- (a) Not spray upwind of the sensitive areas (listed in the explanation to Policy 12.1.1), unless adequate buffer distances are observed, or additional techniques for avoiding spray drift are used.
- (b) Not spray when wind direction is unpredictable or when there are high winds, or very low or no wind conditions. Some wind may assist in correct targeting of spray.
- (c) Not spray during inversion conditions.
- (d) Make use of appropriate and effective buffer zones and/or shelter belts to minimise the risk of spray drifting to non-target areas.
- (e) Have particular regard to the selection of nozzle size and pressure from the spray unit, in order to minimise the risk of spray drift. (*Use equipment generating a droplet size of at least 50 microns in diameter and preferably greater than 250 microns.*)
- (f) In the case of ground application methods, apply spray at a height preferably less than 0.5 metres above, but no greater than 1.5 metres above the target.
- (g) In the case of aerial application methods, be a person who holds a Growsafe Pilots' Agrichemical Rating Certificate of Qualification.
- (h) Comply with the manufacturer's instructions, as stated on the container label or in information sheets.
- (i) Preferably use spray formulations of low volatility and toxicity.
- (j) Dispose of surplus spray solution and spray containers according to the *Code of Practice for the Management of Agrichemicals*, 1999 and the recommendations of the manufacturer or supplier, as stated in the directions on the product container label.
- (k) Keep specific records of the type of each spray and any additive applied, the pest species targeted, the volume of spray, the volume of product, concentrations used, the time, date and locality, identification of any sensitive area (see (a) above), the meeting of any notification requirements, and equipment calibration details, as well as a specific inventory of the types and volumes of any chemicals in storage.

$S\,{\tt C}\,{\tt H}\,{\tt E}\,{\tt D}\,{\tt U}\,{\tt L}\,{\tt E}\,-4$

- (1) Use only those agrichemicals currently authorised for use in New Zealand.
- (m) Also take into account the information provided on the following chart to minimise the risk of drift hazard:

		Conditions that cause pot drift t	
	Factor	High	Low
	Sensitive area	Close (less than 100 m away)	More than 1 km distant
	Buffer Zone	None	Yes (>100 m)
	Shelter Belts	No shelter	Live shelter >3 m high and 1 m thick
Site factors	Wind Direction	Unpredictable	Predictable, and away from sensitive areas
Sit	Wind Speed	Zero / very low (less than 1 m/s) or greater than 6 m/s	Steady (1 - 3 m/s)
	Humidity	Low (delta $T > 8^{\circ}C$)	High (delta $T < 4^{\circ}C$)
	Atmospheric stability	Inversion layer present	No inversion layer present
nique ors	Particle (droplet size)	Less than 50 microns diameter	Greater than 250 microns diameter
Technique factors	Maximum height of release	Greater than 1.5 m above the target	Less than 0.5 m above the target
tors	Volatility	High (vapour pressure > 10 mPa)	Low (vapour pressure < 0.1 mPa)
Material factors	Toxicity	Substance is classed as a 'Poison' in terms of Section 2 of the Toxic Substances Act 1979	No substance used is classed as a 'Poison' in terms of Section 2 of the Toxic Substances Act 1979

Schedule 5 Good Management Practices to Prevent or Minimise the Discharge of Smoke from Burning Vegetation

General burning practices

- Except in the case of tree stumps or standing dead vegetation or crop stubble, vegetation that is to be burned (such as trimmings, pruning, or fellings cut from active growth) should, as a general guide, be allowed to dry for at least four weeks in summer or six weeks in winter, prior to burning.
- Where practicable, the place of burning should be located as far away from roads and state highways as possible and burning should take place when wind will dissipate smoke away from roads and state highways.
- When starting to burn, the direction and strength of wind should be such that smoke is carried away from the areas most likely to be adversely affected.
- In cases of vegetation previously treated by spray with any agrichemical, any manufacturer's instructions, as on the label of any container in respect of the burning of treated vegetation, must be observed.
- Prior to burning, assess whether the vegetation is sufficiently dry to burn, without unnecessarily exacerbating any risk of an uncontrolled fire.
- Vegetation should be stacked loosely rather than compacted to aid drying and the circulation of air for efficient burning.
- For piles of vegetation, a small fire, started with the driest material first, with further material continually fed on to it once it is blazing, is preferable to burning all of the material at the same time in one large pile.
- Once started, a fire should not be left unattended.

These general practices do not apply to high country vegetation burning.

High country vegetation burning

- No fire should be lit if weather conditions indicate that the presence or discharge of smoke or ash is likely to be prolonged.
- Seek advice from a meteorological service prior to burning, on the expected weather conditions, including wind speed and direction.
- Burning should be carried out only when winds will carry smoke and ash away from any operational ski field or other areas of high public use.
- The Cardrona, Treble Cone and Waiorau ski fields should be notified at least 24 hours prior to burning in the Wanaka Basin.

Notes:

- 1. The Transitional Regional Plan may require application for a land use consent to burn vegetation in the high country.
- 2. The burning of vegetation may be subject to the provisions of the Forest and Rural Fires Act and a fire permit may be required from:
 - The Department of Conservation (when burning occurs within the one kilometre fire safety margin of most land administered by the Department);
 - A rural fire authority eg, a city or district council or forestry company.
- 3. Vegetation burning on Crown land requires consent under the Crown Pastoral Lands Act 1998 from the Commissioner of Crown lands.
- 4. A resource consent may be required from the relevant city or district council depending on district plan provisions.

Schedule 6 Setting Chimney Heights for Industrial or Trade Processes

For Rules 16.3.4.1, 16.3.4.2, 16.3.5.1, 16.3.5.3, 16.3.5.4, 16.3.5.5, 16.3.5.6, 16.3.5.7 and 16.3.5.8, the height of the chimney shall be either:

- (i) For any discharge of contaminants other than sulphur dioxide or oxides of nitrogen (and where the activity complies with Rules 16.3.5.1, 16.3.5.3, 16.3.5.4, 16.3.5.5, 16.3.5.6, 16.3.5.7 and 16.3.5.8), the minimum chimney height shall be the higher of either 8.5 metres above the surrounding ground level or 3 metres above the highest substantial part of any building within 50 metres of the chimney; or
- (ii) For any discharge where the release of either sulphur dioxide or nitrogen oxides (expressed as nitrogen dioxide) is less than 0.5 kg/hr, the minimum chimney height shall be the higher of either 8.5 metres above surrounding ground level or three metres above the highest substantial part of any building within 50 metres of the chimney; or
- (iii) For any discharge where the release of either sulphur dioxide or nitrogen oxides (expressed as nitrogen dioxide) equals or exceeds 0.5 kg/hr but is less than or equal to 10 kg/hr the height of the chimney shall be determined in accordance with the table below, but be a minimum of three metres above the highest substantial part of the building to which the chimney may be attached or any adjacent building.

SO ₂ or NO _x (expressed as NO ₂) Emission	Chimney Height (metres)
(kg/hr)	
0.5	8.5
1.0	9.0
1.5	9.5
2.0	10.0
2.5	11.5
3.0	12.0
4.0	13.0
5.0	15.5
6.0	16.5
7.0	17.5
8.0	18.0
9.0	19.0
10.0	20.0

Schedule 7 Standard Dispersion Modelling Procedure

Acceptable Models

Where dispersion modelling is required under this Plan as noted in the information requirements in Part IV, it should be undertaken by an experienced organisation or individual. The preferred models are those developed and used for regulatory purposes by reputable authorities, such as the United States Environmental Protection Authority (USEPA) or State of Victoria's Environmental Protection Agency.

AUSPLUME is the most commonly used model but it has limitations and others may be more suitable for certain situations, such as with complex terrain or the release of gases that are heavier than air.

Examples of other suitable models are:

CTDMPLUS/CTSCREEN	- principally for complex terrain
AUSPUFF (when available)	- large range of applications
SCREEN	- simple first approximations
ISC	- similar to AUSPLUME
DISPMOD	- for complex coastal situations
AUSTOX	- for accidental releases and heavy gases

These models are used to simulate the ground level concentrations of any contaminant present in significant quantities at receptors located at regular and appropriate distances from the discharge point. All significant emissions from the site should be accounted for in the modelling procedures.

If USEPA regulatory models are used (eg, SCREEN, ISC), then it is recommended that the USEPA protocols for using these models be followed.

Input Data

The source and input data should include information on all the particular model's input requirements and default settings. There should also be a description of site related data, including:

- The location and dimensions of buildings likely to affect dispersion (usually those located within 10xL from the discharge points, where L is the lesser of the height or width of the building);
- Other nearby sources of similar contaminants or background concentrations;
- Details of topographic features which might affect the movement or dispersion of the discharges, including coastal features; and
- The location of any particularly sensitive receptors.

Meteorological Data

The minimum meteorological data requirements for most models include:

- Wind velocity;
- Wind direction;
- Turbulence conditions (stability);
- Mixing height; and
- Temperature.

Unfortunately, it is often difficult to obtain real meteorological data from a local site, particularly

information on turbulence and mixing height. A standard "theoretical" meteorological data set is therefore acceptable for the initial assessment of effects. This should cover the full range of conditions that could conceivably occur at the site so the model produces a conservative (high) estimate of the worst case downwind concentration.

In most situations it will be sufficient to use the theoretical or 'screening' meteorology, particularly if it can be clearly demonstrated that the screening approach is conservative (giving high predictions). However, if predicted concentrations exceed the design concentrations then it will be necessary to use real meteorological data for a more precise assessment to determine if this is still the case. Similarly, real meteorological data should be considered if predictions are close to the maximum acceptable concentrations (ie, within 20 percent) and if it cannot be clearly demonstrated that the screening approach was conservative.

Ideally this data should be obtained from a local meteorological station and include at least one year of information. The local data should include reliable information on turbulence and mixing height, along with wind and temperature information. Turbulence or atmospheric stability can be measured in a variety of ways and it is important that a suitably qualified organisation or individual is consulted to provide this. One of the simple ways of measuring stability is to use the "Turner" method where wind data and detailed cloud cover observations are used. The mixing height is often more difficult to obtain but it can also be estimated from basic measurements. Care should be taken in situations where the mixing height has a significant effect of mixing height is important, and it is a major source, then consideration should be given to obtaining more accurate mixing height information from temperature gradient measurements, radiation data, surface temperature data or using a Doppler acoustic sounder.

Interpretation

When using a dispersion model it is important that the correct interpretation is applied to the results. Predictions should be compared to appropriate design targets. To this end, it is not acceptable to use the regional ambient air quality guidelines (Schedule 1) as design concentrations. Such guidelines are designed for protection of air quality in areas where there is often a large number of similar sources. They cannot be used for setting emission limits or chimney heights of individual sources without some kind of modification.

In many situations it will be necessary to determine an appropriate design concentration for a particular situation in consultation with Council staff. However, the following provides a guide to what will be considered as acceptable model design concentrations for new sources in Otago:

- For a source located in a Schedule 2 area:
 - (i) 75% of the Otago Goal Levels listed in Schedule 1; or
 - (ii) The Workplace Exposure Standard (WES)* divided by 50, where appropriate, and used as a 1-hour average.
- For a source located in non-urban areas:
 - (i) The Otago Goal Levels listed in Schedule 1; or
 - (ii) The WES divided by 30, where appropriate, and used as a 1-hour average.
- For a source located in a pristine area, a suitable design concentration should be determined in consultation with Council staff.

^{*} Workplace Exposure Standards (1994), Department of Labour, Wellington

- For carcinogens or those contaminants that primarily give rise to health effects from long term exposure, it may be necessary to follow a risk assessment approach where model results are used to estimate lifetime dose via inhalation, and to compare this with other dosage pathways and acceptable risk levels.
- For those contaminants that primarily produce odour effects, it is preferable to use an odour design concentration based on olfactometry and odour dose-response assessments, determined in consultation with Council staff.
- For dusts and other pollutants that primarily produce amenity affects, a suitable design concentration should be determined in consultation with Council staff.

In all the above cases the acceptable design concentration or guideline should be compared with the maximum prediction, the highest 99.9 percentile when real meteorological data is used for 1-hour average predictions (8th highest prediction out of a whole year of hourly predictions) or the highest 99.5 percentile in the case of odour.

The correct averaging times should also be used when comparing predictions with design concentrations. Particular care should be taken when using predictions from screening meteorology, for those contaminants that have design concentrations with averaging times greater than one hour.

It should be remembered that the above are only guidelines. They do *not* represent strict standards. Deviation from these guidelines may be acceptable, but the onus is on the applicant to ensure that any deviations are based on sound science. Consequently applicants are advised to obtain appropriate specialist advice.

These guidelines also primarily apply to the assessment of new sources. Different criteria may be applied to the interpretation of modelling results for some existing sources.

Glossary



Terms marked with an asterisk * are terms defined by Section 2 of the Resource Management Act 1991.

AAQG	Ambient Air Quality Guidelines.
Abrasive blasting	The cleaning, smoothing, roughening, cutting or removing of part of the surface or any article by the use of a jet of sand, metal shot, grit or any other abrasive material propelled by a blast of compressed air or mechanically via a rotary wheel, impeller or other means.
Act	The Resource Management Act 1991 and its amendments.
Adverse effect	An unwanted or detrimental effect.
Aerial application	The discharge of any agrichemical from any aircraft.
Agrichemical spray drift	The airborne movement of aerosol or droplets containing agrichemicals onto non target areas.
Agrichemical	Any substance that is used to eradicate, modify or control flora or fauna, including animal remedies, but excluding fertilisers.
Airshed	 Means - (a) The region of a regional council excluding any area specified in a notice under paragraph (b): (b) A part of the region of a regional council specified by the Minister by notice in the <i>Gazette</i> to a separate airshed. Note: Boundaries of the Otago airsheds gazetted as per (b) above, are shown in Schedule 2.
Air Zone	 For simplicity of management, Otago's airsheds have been categorised into three Air Zones, as follows: Air Zone 1: including Alexandra, Arrowtown, Clyde and Cromwell airsheds. Air Zone 2: including Balclutha, Dunedin, Green Island, Hawea, Kingston, Milton, Mosgiel, Naseby, Oamaru, Palmerston, Port Chalmers, Queenstown, Ranfurly, Roxburgh, Waikouaiti and Wanaka airsheds. Air Zone 3: the whole of Otago, excluding the areas that are zoned 1 or 2.
	The Clyde airshed is identified as a sub-zone within Air Zone 1 until 1 April 2009, pending further monitoring and an analysis of how discharges to air within the Clyde airshed affect the Alexandra airshed.
Ambient air	The air outside buildings and structures. It does not refer to indoor air, to air in the workplace, or to contaminated air as it is discharged from a source.

Ambient air quality standard for PM ₁₀	 Means - (a) A threshold concentration of 50 µg/m³ expressed as a 24-hour mean; and (b) A permissible excess of one 24-hour period in a 12-month period.
Amenity values*	Means those natural or physical qualities and characteristics of an area that contribute to people's appreciation of its pleasantness, aesthetic coherence, and cultural and recreational attributes.
Area source discharges	Involve diffuse (or non point source) discharges into air which may not be attributable to an individual activity.
Best practicable option*	 In relation to a discharge of a contaminant or an emission of noise, means the best method for preventing or minimising the adverse effects on the environment having regard, among other things, to- (a) The nature of the discharge or emission and the sensitivity of the receiving environment to adverse effects; and (b) The financial implications, and the effects on the environment, of that option when compared with other options; and (c) The current state of technical knowledge and the likelihood that the option can be successfully applied.
Bio-oil	Oil derived from animal or plant matter
BOD	Biochemical oxygen demand (BOD) is a measure of the oxygen consumed by the degradation of organic matter by organisms, and therefore a measure of organic pollution.
Boundary of the property	Means the legal perimeter of the property upon which any discharge into air is occurring.
Buffer zone	In relation to the use of agrichemicals and potentially odorous discharges, means the distance between the downwind edge of an area where the activity is undertaken, and sensitive land uses.
Coastal marine area*	 Means the foreshore, seabed and coastal water, and the air space above the water - (a) Of which the seaward boundary is the outer limits of the territorial sea: (b) Of which the landward boundary is the line of mean high water springs, except that where that line crosses a river, the landward boundary at that point shall be whichever is the lesser of - (i) One kilometre upstream from the mouth of the river; or (ii) The point upstream that is calculated by multiplying the width of the river mouth by 5.

Commercial premises	Any premises used for tourist, hospitality or accommodation activities that displays, offers, provides, sells, or hires goods, equipment or services.
Conditions*	In relation to plans and resource consents, includes terms, standards, restrictions, and prohibitions.
Consent authority*	Means the Minister of Conservation, a regional council, a territorial authority, or a local authority that is both a regional council and a territorial authority, whose permission is required to carry out an activity for which a resource consent is required under the Resource Management Act 1991.
Consultation	The communication of a genuine invitation to give advice and a genuine consideration of that advice.
Contaminant*	 Includes any substance (including gases, liquids, solids and micro-organisms) or energy (excluding noise) or heat, that either by itself or in combination with the same, similar, or other substances, energy or heat - (a) When discharged into water, changes or is likely to change the physical, chemical or biological condition of water; or (b) When discharged onto or into land or into air, changes or is likely to change the physical, chemical, or biological condition of the land or air onto or into which it is discharged.
Contravene*	Includes fail to comply with.
Cooker	Any domestic heating appliance which has an inbuilt oven and is used for cooking.
Curved line path	 Means a curved line that – (a) Starts on the y-axis of a graph at a point representing, as at 1 September 2005 or the date that the plan is publicly notified (whichever is later), the concentration of PM₁₀ in the airshed; and (b) Ends on the x-axis of the graph at a point representing as at 1 September 2013, the ambient air quality standard for PM₁₀ in the airshed.
Discharge permit	Has the meaning set out in Section 87(e) (of the Resource Management Act 1991).
Discharge*	Includes emit, deposit and allow to escape.
Discretionary activity*	 If an activity is described in the Resource Management Act 1991, regulations, or a plan or proposed plan as a discretionary activity, - (a) A resource consent is required for the activity; and (b) The consent authority may grant the resource consent with or without conditions or decline the resource

	consent; and(c) The activity must comply with the standards, terms or conditions, if any, specified in the plan or proposed plan.
District plan*	Means an operative plan approved by a territorial authority under the First Schedule to the Resource Management Act; and includes all operative changes to such a plan (whether arising from a review or otherwise).
Domestic heating appliance	A combustion appliance, with a heat generation capacity of up to 50 kW in which solid fuel is burnt for heating or cooking, and is primarily used in residential dwellings. It includes, but is not limited to, any open fire, woodburner, multifuel, pellet or coal burning heater, or cooker including coal range.
Dry	With respect to vegetative matter, means neither damp nor so freshly cut that it does not burn cleanly.
Dry abrasive blasting	Where an abrasive is incorporated into an air stream and directed at high velocity from a nozzle onto the work piece, or where the abrasive is mechanically thrown via a rotating wheel, impeller or other means.
Dust	Means all solid particulate matter that is suspended in the air, or has settled after being airborne. By way of example "dust" may be derived from sand, cement, fertiliser, coal, soil, paint, ash, animal products or wood.
Ecosystem	A dynamic complex of plant, animal and micro-organism communities and their non-living environment interacting as a functional unit.
Effect	 Section 3 of the Resource Management Act defines the term effect to include: (a) Any positive or adverse effect; and (b) Any temporary or permanent effect; and (c) Any past, present, or future effect; and (d) Any cumulative effect which arises over time or in combination with other effects - (e) re regardless of the scale, intensity, duration or frequency of the effect, and also includes - (f) Any potential effect of high probability; and (g) Any potential effect of low probability which has a high potential impact.
Enforcement order*	Means an order made under section 319 for any of the purposes set out in section 314; and includes an interim enforcement order made under section 320 of the Resource Management Act 1991.
Environment*	Includes -(a) Ecosystems and their constituent parts, including people and communities; and

$G \mathrel{\hbox{\rm L}} o \mathrel{\hbox{\rm S}} s \mathrel{\hbox{\rm A}} r \mathrel{\hbox{\rm Y}}$

	 (b) All natural and physical resources; and (c) Amenity values; and (d) The social, economic, aesthetic, and cultural conditions which affect the matters stated in paragraphs (a) to (c) of this definition or which are affected by those matters.
Fauna	All the animal life of a given place.
Fertiliser	Any substance other than animal waste which is in a state suitable for application to land or plants for the purpose of increasing the growth or productivity of beneficial plants.
Flora	All the plant life of a given place.
Frost pot	An appliance in which oil or diesel is commonly burnt, which is used to prevent frost damage in orchards.
Fuel burning equipment	Any boiler, furnace, gas turbine, internal or external combustion engine (excluding motor vehicles, ships, trains and aircraft) that includes a chimney or exhaust and is used primarily for the production of energy, or other like processes.
General engineering activity	Welding, soldering, brazing, arc air gouging, grinding, cutting, or activities of a similar nature.
Ground based application methods	Any method of application where that part of the equipment from where the agrichemical is emitted is either on the ground or attached to equipment that is on the ground.
Hand held application methods	Any method of application where the applicator holds that part of the equipment from which the agrichemical is emitted.
Hazardous air contaminant	Means a contaminant identified in Schedule 3.
Heat generation capacity	Refers to gross heat release from the fuel not steam production capacity or heat transfer capacity.
Incineration	A process which completely reduces materials to ash through combustion in a purpose-built appliance.
Incinerator	An appliance used primarily for the destruction of materials which reduces them to ash through combustion. Note, for the purposes of this Plan, this does not include a modified drum or garden incinerator.

Industrial or trade premises*	 (a) Any premises used for any industrial or trade purposes; or (b) Any premises used for the storage, transfer, treatment or disposal of waste materials or for other waste-management purposes, or used for composting organic materials; or (c) Any other premises from which a contaminant is discharged in connection with any industrial or trade process - but does not include any production land.
Industrial or trade process*	Includes every part of a process from the receipt of raw material to the dispatch or use in another process or disposal of any product or waste material, and any intervening storage of the raw material, partly processed matter, or product.
Industrial or trade wastes	Wastes from an industrial or trade process or premises.
Intensive Farming	Means any production of livestock which is carried out predominantly within buildings or fenced outdoor areas where the stocking density precludes the maintenance of pasture or ground cover.
Issue	A matter of concern to the region's community regarding activities affecting some aspect of natural and physical resources and the environment of the region.
Iwi	Tribe.
Kai Tahu	Descendants of Tahu, the tribe.
Kaitiaki	Guardians.
Kaitiakitanga*	The exercise of guardianship by the tangata whenua of an area in accordance with tikanga Maori in relation to natural and physical resources; and includes the ethic of stewardship.
Land use consent*	Has the meaning set out in section 87(a) (of the Resource Management Act 1991).
Local authority	A term that collectively describes regional councils, city councils, and district councils.
Mahika kai	Place where food is produced or procured.
Mana	Authority, prestige, influence.
Marae	Courtyard, meeting place for tangata whenua.
Mass emission rate	The quantity of contaminants discharged on a mass per unit time basis.
Mauri	Life force.

Mean high water springs	The average line of spring high tide.
Method	The practical action by which a policy is implemented.
Micron	One-millionth of a metre.
Mineral	Means a naturally occurring inorganic substance beneath or at the surface of the earth, whether or not under water; and includes all metallic minerals, non-metallic minerals, fuel minerals, precious stones, industrial rocks and building stones, and a prescribed substance within the meaning of the Atomic Energy Act 1945.
Mitigate	To make or become less severe or harsh. To moderate.
Mobile operations	In relation to abrasive blasting and spraypainting, mobile operations refer to operations where the activity is undertaken for five days or less on one site.
Multifuel heater	Means a domestic heating appliance designed to burn more than one type of solid fuel.
Municipal waste	Waste generated by the local community and disposed of at a central location.
Natural and physical	Includes land, water, air, soil, minerals, and energy, all forms of plants and animals (whether native to New Zealand or
resources*	introduced), and all structures.
resources* NESAQ	-
	introduced), and all structures. Resource Management (National Environmental Standards Relating to Certain Air Pollutants, Dioxins and Other Toxics)
NESAQ	introduced), and all structures. Resource Management (National Environmental Standards Relating to Certain Air Pollutants, Dioxins and Other Toxics) Regulations 2004. The desired result, end state, situation or condition that is aimed
NESAQ Objective	 introduced), and all structures. Resource Management (National Environmental Standards Relating to Certain Air Pollutants, Dioxins and Other Toxics) Regulations 2004. The desired result, end state, situation or condition that is aimed for. (a) Means petroleum in any form (other than gas); and (b) Includes crude oil, fuel oil sludge, oil refuse, and refined
NESAQ Objective Oil	 introduced), and all structures. Resource Management (National Environmental Standards Relating to Certain Air Pollutants, Dioxins and Other Toxics) Regulations 2004. The desired result, end state, situation or condition that is aimed for. (a) Means petroleum in any form (other than gas); and (b) Includes crude oil, fuel oil sludge, oil refuse, and refined oil products (e.g. diesel fuel, kerosene and motor gasoline). A fireplace which is designed to operate as an open fireplace in

Outside storage Means material stored outside without wind protection above and on at least three sides (or ³/₄ of the perimeter) of the storage area.

- **Permitted activity*** If an activity is described in the Resource Management Act 1991, regulations, or a plan or proposed plan as a permitted activity, a resource consent is not required for the activity if it complies with the standards, terms, or conditions, if any, specified in the plan or proposed plan.
- **Person*** Includes the Crown, a corporation sole, and also a body of persons, whether corporate or unincorporate.
- **Place of public assembly** Land or buildings which are used in whole or in part for the assembly or gathering of people for such purposes as meetings, conferences, worship, entertainment, recreation, celebration, education or similar purposes and includes such buildings associated with public or private hotels, travellers' accommodation and marae.

PM₁₀ Means particulate matter that is -

- (a) Less than 10 microns in aerodynamic diameter; and
- (b) Measured in accordance with the United States Code of Federal Regulations, Title 40 – Protection of Environment, Volume 2, Part 50, Appendix J – Reference method for the determination of particulate matter as PM₁₀ in the atmosphere.

Note: There is a link to (b) on the Ministry for the Environment website (www.mfe.govt.nz).

- Point sourceA discharge of water or contaminant that enters the air from adischargedefinable point, often through a stack or chimney.
- **Policy** The course of action to achieve the objective.
- **Production land*** Means any land and auxiliary buildings used for the production (but not processing) of primary products (including agricultural, pastoral, horticultural, and forestry products): Does not include land or auxiliary buildings used or associated with prospecting, exploration, or mining for minerals; and "production" has a corresponding meaning.
- **Prohibited activity** If an activity is described in the Resource Management Act 1991, regulations, or a plan as a prohibited activity, no application may be made for that activity and a resource consent must not be granted for it.
- **Proposed plan*** Means a proposed plan, or variation to a proposed plan, or change to a plan that has been notified under clause 5 of the First Schedule but has not become operative in terms of clause

Public amenity area	 20 of the First Schedule, but does not include a proposed plan or change originally requested by a person other than a local authority or a Minister of the Crown, unless the proposed plan or change is adopted and notified by the local authority under clause 25(2)(a) of the First Schedule. Means any area to which the public has right of access under any statute, regulation, law or by-law, including: (a) Crown and council properties, reserves, gardens, parks and airfields; (b) Grasslands, sports grounds and recreational turf; (c) Forest and bush areas; (d) Roads railways and their verges and embankments, pedestrian walkways, malls and precincts; (e) Beaches, beach reserves and adjacent foreshore areas.
Rangatiratanga	Chieftainship, decision-making rights.
Regional coastal plan*	Means an operative plan approved by the Minister of Conservation under the First Schedule of the Resource Management Act, and includes all operative changes to such a plan (whether arising from a review or otherwise).
Regional plan*	Means an operative plan (including a regional coastal plan) approved by a regional council or the Minister of Conservation under the First Schedule; and includes all operative changes to such a plan (whether arising from a review or otherwise).
Regional Plan: Air	Means the Regional Plan: Air for Otago.
Regional policy statement*	Means an operative regional policy statement approved by a regional council under the First Schedule; and includes all operative changes to such a policy statement (whether arising from a review or otherwise).
Recognised heritage building	Any Category I or Category II building as listed on The Register of Historic Buildings, Historic Areas, Wahi Tapu and Wahi Tapu Areas, or any other building listed as a heritage building in a district plan.
Residential dwelling	Means any building, that is occupied, whether permanently or temporarily, in whole or in part, as a residence; and includes any structure or outdoor living area that is accessory to, for the purposes of, the residence; but does not include the land upon which the residence is sited.
Residential portion	The portion of a property in the vicinity of a residential dwelling which is used primarily for domestic purposes.

- **Residential property** A property that contains at least one permanent residential dwelling and is used primarily for domestic purposes; but does not include a hostel or boarding house or other specialised accommodation.
- **Resource consent** A consent for an activity as set out in Section 87 (of the Resource Management Act); and includes all conditions to which the consent is subject.
- Ship*Has the same meaning as in Section 2 of the Maritime
Transport Act 1994.
- **Solid fuel** Means a solid substance that releases useable energy when burnt (for example, wood and coal).
- Taoka Treasure.
- **Territorial authority** A term that collectively describes city councils and district councils, but not regional councils.
- The ActIn this Plan, reference to "the Act" means the Resource
Management Act 1991.
- The Plan Means the Regional Plan: Air for Otago.
- **Thermal efficiency** Ratio of useable heat energy output to energy input.
- Treaty of WaitangiThe same meaning as the word "Treaty" as defined in Section 2(Te Tiriti oof the Treaty of Waitangi Act 1975.Waitangi)Output
- Untreated wood Any wood material or product, including sawdust, which is not treated with copper, chrome or arsenic ("CCA" treated or "Tanalised"), or with any organochlorine preservative.
- **Vacuum blasting** Dry blasting where the abrasive is applied via a nozzle surrounded by a shroud or mask equipped with a vacuum, and where debris and abrasive is effectively collected in control equipment.
- Waahi taoka Treasured resource.
- Waahi tapu Sacred places.
- Wairua Life principle, spirit.
- Waste Means substances or objects that are disposed of or intended to be disposed of.
- Wet abrasiveAbrasive blasting when water or a mixture of abrasive and
water is added to the air flow carrying the abrasive material
prior to the blasting nozzle exit, or when the blasting medium

is predominantly a pressurised slurry.

Woodburner

(a) Means a domestic heating appliance that burns wood; but

- (b) Does not include
 - (i) An open fire; or
 - (ii) A multifuel heater, a pellet heater, or a coal burning heater; or
 - (iii) A stove that is -
 - (A) Designed and used for cooking; and
 - (B) Heated by burning wood.